SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

Form 10-Q

X QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d)
OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarter ended June 30, 1994

0R

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from _____ to ____

Commission file number 1-10235

IDEX Corporation

(Exact name of registrant as specified in its charter)

Delaware State or other jurisdiction of incorporation or organization 36-3555336 (I.R.S. Employer Identification No.)

630 Dundee Road, Suite 400
Northbrook, Illinois 60062
(Address of principal executive offices) (Zip Code)

Registrant's telephone number, including area code (708) 498-7070

Former name, former address and former fiscal year, if changed since last

report.

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes X No

Number of shares of common stock of IDEX Corporation ("IDEX" or the "Company") outstanding as of August 9, 1994: 12,710,161 shares.

Documents Incorporated by Reference: None.

Item 1. Financial Statements

IDEX CORPORATION AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS (in thousands)

	June 30, 1994	December 31, 1993
	(unaudited)	
ASSETS		
Current Assets		
Cash and cash equivalents	\$ 3,427	\$ 3,513
Receivables - net	56,579	43,318
Inventories	79,119	60,973
Deferred taxes	7,986	6,602
Other current assets	2,426	1,060
Total Current Assets	149,537	115,466
Property, Plant and Equipment - net	66,361	53, 525
Intangible Assets - net	149, 258	84, 772
Other Noncurrent Assets	5,006	5,204
	<u>-</u>	
Total Assets	\$370,162	\$258,967
LIADTITITES AND SHADEHOLDERS! FOLITY	======	======
LIABILITIES AND SHAREHOLDERS' EQUITY		
Current Liabilities		
Trade accounts payable	\$ 30,090	\$ 21,405
Accrued expenses	30,338	21,235
Total Current Liabilities	60,428	42,640
Long-Term Debt	191, 930	117,464
Other Noncurrent Liabilities	18, 284	15,177
Total Liabilities	270,642	175,281
TOTAL ELABILITIES		
Shareholders' Equity		
Common stock, par value \$.01 per share;		
Shares authorized: 50,000,000		
Shares issued and outstanding:		
1994: 12,708,744		
1993: 12,701,588	127	127
Additional paid-in capital	84,793	84,713
Retained earnings	18,076	2,551
Accumulated translation adjustment	(3,476)	(3,705)
Total Charaholders! Equity	00 520	92 696
Total Shareholders' Equity	99,520	83,686
Total Liabilities and Shareholders'		
Equity	\$370,162	\$258,967
•	======	======

IDEX CORPORATION AND SUBSIDIARIES STATEMENTS OF CONSOLIDATED OPERATIONS (In thousands, except per share amounts)

For the Second Quarter Ended June 30,	1994	1993
		(unaudited)
Net sales Cost of sales	\$93,559 57,402	\$78,052 47,893
Gross profit Selling, general and administrative	36,157	30,159
expenses	19,799	17,221
Income from operations	16,358	12,938
Other expense - net	558	242
Interest expense	3,113	2,809
Income before income taxes	12,687	9,887
Provision for income taxes	4,509	3,427
TOVESED TO THOOME CAXOSTITITITITITI		
Net income	\$ 8,178	\$ 6,460
	======	======
Earnings per common share	\$.63	\$.50
	======	======
Weighted average common shares		
outstanding	13,042	12,947
•	======	======

IDEX CORPORATION AND SUBSIDIARIES STATEMENTS OF CONSOLIDATED OPERATIONS (In thousands, except per share amounts)

For the Six Months Ended June 30,	1994	1993
	(una	audited)
Net sales Cost of sales	\$179,433 109,886	\$151,603 93,723
Gross profitSelling, general and administrative	69,547	57,880
expenses	38,781	34,441
Income from operations	30,766	23,439
Other expense - net	1,029	377
Interest expense	5,746	5,668
Income before income taxes	23,991	17,394
Provision for income taxes	8,466	6,055
Net income	\$ 15,525	\$ 11,339
	=======	=======
Earnings per common share	\$ 1.19	\$.88
.	======	======
Weighted average common shares		
outstanding	13,035	12,943
·	=======	=======

IDEX CORPORATION AND SUBSIDIARIES STATEMENT OF CONSOLIDATED SHAREHOLDERS' EQUITY (In thousands)

Shareholders' Equity

	Common Stock	Additional Paid-In Capital	Retained Earnings	Accumulated Translation Adjustment
Balance: December 31, 1993	\$127	\$84,713	\$ 2,551	\$(3,705)
Stock options exercised		80		
Unrealized trans- lation adjustment				229
Net income			15,525	
Balance: June 30, 1994 (unaudited)	\$127 ====	\$84,793 =====	\$ 18,076 ======	\$(3,476) ======

IDEX CORPORATION AND SUBSIDIARIES STATEMENTS OF CONSOLIDATED CASH FLOWS (In thousands)

For the Six Months Ended June 30,	1994	1993
	(unaudit	
Cash Flows From Operating Activities: Net income	\$15,525	\$11,339
provided by operating activities: Depreciation and amortization	4,619 1,652 318 (4,762) 479 3,676 (528) (736) 952	4,463 1,400 322 (2,743) 1,343 (118) 492 1,809 1,916
Net cash flows from operating activities	21,195	20,223
Cash Flows From Investing Activities: Additions to property, plant and equipment Acquisition of businesses (net of cash	(4,194)	(3,482)
acquired)	(91,553)	
Net cash flows from investing activities	(95,747)	(3,482)
Cash Flows From Financing Activities: Net borrowings (repayments) of long-term debt Increase in accrued interest Payment of deferred financing costs	74,000 466	(16,000) 295 (638)
Net cash flows from financing activities	74, 466	(16,343)
Net increase (decrease) in cash Cash and cash equivalents at beginning of period	(86) 3,513	398 2,370
Cash and cash equivalents at end of period	\$ 3,427 ======	\$ 2,768 =====
Supplemental Disclosure of Cash Flow		
Cash paid during the period for: Interest Taxes (including foreign)	\$ 4,835 7,016	\$ 4,943 1,095

IDEX CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

1. Organization and Acquisition

Pursuant to the requirements of the Securities and Exchange Commission, the January 22, 1988 Acquisition of the initial six businesses comprising IDEX Corporation ("IDEX" or the "Company") was not accounted for as a purchase transaction. Consequently, the accounting for the acquisition does not reflect any adjustment of the carrying value of the assets and liabilities to their fair values at the time of the acquisition. Accordingly, the shareholders' equity of IDEX at June 30, 1994 and December 31, 1993 includes a charge of \$96.5 million which represents the excess of the purchase price over the book value of the subsidiaries purchased at the date of the acquisition.

Acquisition of Hale Products, Inc.

On May 26, 1994, IDEX purchased all of the outstanding shares of common stock of Hale Products, Inc. ("Hale"), a leading manufacturer of fire-fighting pumps and rescue tools. The purchase price for the acquisition including stock purchase, debt assumption and transaction fees was approximately \$95 million in cash. The purchase was financed through borrowings under the Company's bank revolving credit facility which was amended to provide an additional \$50 million of availability (\$150 million of total availability) and improvement to the interest rate structure. The excess of the purchase price over the fair value of the net assets acquired of \$61.5 million will be amortized over 40 years. The acquisition has been accounted for by the purchase method of accounting. The liabilities assumed in connection with the acquisition of Hale were as follows:

Fair value of assets acquired Cost in excess of net assets acquired Cash paid for common stock of Hale and	\$ 49,578 61,459
related transaction expenses	(56,068) (38,750)
Liabilities assumed	\$ 16,219 ======

Hale's financial performance for the period January 1 to May 25, 1994, prior to acquisition by IDEX, was adversely affected by several factors. Customarily, Hale's shipments are stronger in the second half of a calendar year than the first half due to the purchasing practices of customers in industries that it serves. In 1994, shipments were further reduced by production curtailments at the Conshohocken facilities because of severe winter weather and unexpected facility repairs at its foundry. In addition, Hale was in the process of moving production of certain products between its Conshohocken, Pennsylvania and St. Joseph, Tennessee facilities during this period which created certain temporary inefficiencies and loss of overhead absorption. Higher than normal selling, general and administrative expenses were incurred during this period due to Hale's participation in a major international fire and rescue trade show which is held every six years in Germany. During the period January 1 through May 25, 1994 order activity remained strong as sales backlogs increased by \$5.5 million.

The unaudited pro forma consolidated results of operations of IDEX for the six months ended June 30, 1994 and 1993 reflecting the allocation of the purchase price and related financing of the transactions, would have been as follows (in thousands except per share amounts), assuming that the Hale acquisition had occurred at the beginning of each of the respective periods.

	1994	1993
Net sales	\$202,019	\$183,665
Income before extraordinary item and cumulative effect of changes in		
accounting for income taxes	13,210	11,476
Net income	13,210	10,881
Earnings per common share:		
Income before extraordinary item and		
cummulative effect of change in		
accounting for income tax	1.01	.89
Net income	1.01	.84

3.(a) Significant Accounting Policies

In the opinion of management, the unaudited information presented as of June 30, 1994 and for the six-month periods ended June 30, 1994 and 1993 reflects all adjustments necessary, which consist only of normal recurring adjustments, for a fair presentation of the interim periods.

(b) Earnings Per Share

Earnings per share is computed by dividing net income by the weighted average number of shares of common stock and common stock equivalents outstanding during the period. Common stock equivalents, in the form of stock options, have been included in the calculation of weighted average shares outstanding under the treasury stock method.

4. Inventory

The components of inventories as of June 30, 1994 and December 31, 1993 were (000's omitted):

	June 30, 1994 	December 31, 1993
Inventories		
Raw materials and supplies	\$ 9,407	\$ 8,498
Work in process	10,776	7,018
Finished goods	58,936	45,457
Totals	\$79,119	\$60,973

The inventories on a LIFO basis amounted to \$45,873 and \$25,874 at June 30, 1994 and December 31, 1993, respectively. The excess of current cost over LIFO inventory value and the impact on earnings of using the LIFO method are not material.

5. Preferred Stock

The Company had five million shares of preferred stock authorized but unissued at June 30, 1994 and December 31, 1993.

For the Second Quarter Ended June 30,	1994(1)	1993
	(unaudit	ed)
Fluid Handling Group (2)		
Net sales	\$63,267	\$53,542
Income from operations	13,739	10,891
Operating margin	21.7%	20.3%
Depreciation and amortization (4)	\$ 2,507	\$ 2,275
Capital expenditures	2,096	1,292
Industrial Products Group (2)		
Net sales	\$30,393	\$24,582
Income from operations	4,644	3,646
Operating margin	, 15.3%	14.8%
Depreciation and amortization (4)	\$ 754	\$ 648
Capital expenditures	490	504
Company (3)		
Net sales	\$93,559	\$78,052
Income from operations	16,358	12,938
Operating margin	17.5%	16.6%
Depreciation and amortization (4)	\$ 3,273	\$ 2,935
Capital expenditures	2,636	1,822
·	,	, -

- (1) Includes the operations of Signfix from January 1, 1994, which are not material to the Company, and Hale from May 26, 1994.
- (2) Income from operations excludes unallocated corporate operating expenses.
- (3) Includes the operations of the two business groups in addition to corporate operating expenses and inter-group eliminations.
- (4) Excludes amortization of debt issuance expenses.

For the Six Months Ended June 30,	1994(1)	1993
	 (unaudited	
	(4.14442200	• /
Fluid Handling Group (2)		
Net sales	\$119,818	\$104,713
Income from operations	25,635	20,219
Operating margin	21.4%	19.3%
Depreciation and amortization (4)	\$ 4,753	\$ 4,553
Capital expenditures	3,264	2,672
Industrial Products Group (2)		
Net sales	\$ 59,785	\$ 47,028
Income from operations	8,892	6,410
Operating margin	14.9%	13.6%
Depreciation and amortization (4)	\$ 1,492	\$ 1,287
Capital expenditures	880	781
Company (3)		
Net sales	\$179,433	\$151,603
Income from operations	30,766	23,439
Operating margin	17.1%	15.5%
Depreciation and amortization (4)	\$ 6,271	\$ 5,863
Capital expenditures	4,194	3,482

- (1) Includes the operations of Signfix from January 1, 1994, which are not material to the Company, and Hale from May 26, 1994.
- (2) Income from operations excludes unallocated corporate operating expenses.
- (3) Includes the operations of the two business groups in addition to corporate operating expenses and inter-group eliminations.
- (4) Excludes amortization of debt issuance expenses.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

Historical Overview and Outlook

IDEX sells a broad range of fluid handling and industrial products to a diverse customer base in the United States and internationally. Accordingly, IDEX's businesses are generally affected by levels of industrial activity and economic conditions in the United States and in those foreign countries where its products are sold, and to some extent, by the relationship of the dollar to other currencies. Among the factors that affect the demand for IDEX's products are interest rates, levels of capital spending by industry and overall industrial growth.

The incoming order rate for IDEX for the second quarter of 1994 increased 24% over the second quarter of 1993 and represented a new record for orders written in a quarter. Base businesses contributed 14% of the increase while Signfix, acquired late in 1993, and Hale, acquired in May of 1994, contributed 10% of the increase. As a result, backlogs are somewhat stronger; however, the Company continues to operate with very low order backlogs, typically about 1-1/2 months' sales, in order to provide superior customer service, and any decline in orders would have an immediate effect on sales and profits.

The second quarter of 1994 compared favorably to the second quarter of 1993 as IDEX achieved record quarterly sales, net income and earnings per common share. Sales increased 20%, net income was up 27% and earnings per share rose 26% in the second quarter of 1994 compared to the second quarter of 1993. Changes in IDEX's quarterly results from the comparable prior year period resulted from improved economic conditions as all of IDEX's base business units reported stronger operating results which were further enhanced by inclusion of activity from recent acquisitions - Signfix in November 1993 and Hale in May 1994. Signfix was an addition to the Company's Industrial Products group and Hale was added to the Fluid Handling Group.

The Company is well positioned in its markets, serves diverse industries, sells throughout the world and enjoys high margins. Given current business conditions and a continuation of current industrial activity, management expects that IDEX will attain new records in sales, net income, and earnings per share for the year.

For purposes of this discussion and analysis section, reference is made to the tables set forth on pages 8 and 9 and the Company's Statements of Consolidated Operations included in the Financial Statement section. IDEX consists of two business segments: Fluid Handling and Industrial Products.

Performance in the Second Quarter Ended June 30, 1994 Compared to 1993

Sales, net income and earnings per common share were at record levels in the three months ended June 30, 1994 versus the same period in 1993. Incoming orders, also at record levels, rose 14% in the Company's base businesses and the acquisition of Signfix and Hale added another 10% to the prior year's second quarter order rate.

Second quarter 1994 consolidated net sales of \$93.6 million increased \$15.5 million or 20% from the comparable period in 1993. Fluid Handling Group sales of \$63.3 million increased \$9.7 million or 18% due to stronger volume resulting from improved economic conditions and inclusion of Hale activity from its May 1994 acquisition. Sales in the Industrial Products Group of \$30.4 million increased \$5.8 million or 24% due to the inclusion of Signfix activity from its November 1993 acquisition and improved sales demand for the other units in this Group

Gross profit of \$36.2 million in the second quarter of 1994 increased \$6.0 million or 20% from the corresponding 1993 period. Gross profit as a percentage of sales in the 1994 period of 38.7% was approximately equivalent to that of 1993. Selling, general and administrative expenses increased to \$19.8 million in the second quarter of 1994 from \$17.2 million in the second quarter of 1993 principally as a result of acquisitions, but decreased as a percentage of sales to 21.1% in 1994 from 22.1% in 1993.

Income from operations increased 26% to \$16.4 million in the second quarter of 1994 from \$12.9 million in the second quarter of 1993 as operating margin as a percent of sales increased to 17.5% from 16.6%. In the Fluid Handling Group, income from operations of \$13.7 million and operating margin of 21.7% for the second quarter of 1994 were higher than income from operations of \$10.9 million and operating margin of 20.3% in the second quarter of 1993 due to volume-related gains with improving business conditions. Income from operations in the Industrial Products Group of \$4.6 million and operating margin of 15.3% in the second quarter of 1994 were higher than income from operations of \$3.6 million and operating margin of 14.8% in the second quarter of 1993 as all units in the Industrial Products Group experienced volume-related improvements.

Interest expense increased to \$3.1 million in the second quarter of 1994 from \$2.8 million in the comparable 1993 period due to additional borrowings from the November 1993 acquisition of Signfix and the May 1994 acquisition of Hale and higher interest rates.

The provision for income taxes increased to \$4.5 million in the second quarter of 1994 from \$3.4 million in the second quarter of 1993. The effective tax rate of 35.5% in the second quarter of 1994 increased from the 1993 rate of 35.0% as the amortization of goodwill expense arising from the Hale acquisition is not deductible for tax purposes. Net income was \$8.2 million in the second quarter of 1994 and was 27% higher than the \$6.5 million recorded in the 1993 period. Earnings per common share amounted to \$.63 in the second quarter of 1994 which was 26% higher than the \$.50 recorded in the second quarter of 1993.

Sales, net income and earnings per common share were also at record levels in the six months ended June 30, 1994 versus 1993. Incoming orders rose 12% in the Company's base businesses and inclusion of Hale and Signfix activity added another 7% to the prior year's first six months rate.

Six-month 1994 net sales of \$179.4 million increased \$27.8 million or 18% from 1993. Fluid Handling Group sales of \$119.8 million increased \$15.1 million or 14% due to stronger volume from improved market conditions and Hale activity from its May 1994 acquisition. Sales in the Industrial Products Group of \$59.8 million increased \$12.8 million or 27% due to the inclusion of Signfix activity in the first six months of 1994 and improved sales demand for the other units in this group.

Gross profit of \$69.5 million in the first six months of 1994 increased \$11.7 million or 20% from the corresponding 1993 period, and gross profit as a percentage of sales of 38.8% in 1994 increased from 38.2% in 1993. Selling, general and administrative expenses increased to \$38.8 million in the first six months of 1994 from \$34.4 million in the first six months of 1993, but as a percentage of sales, decreased to 21.6% in 1994 from 22.7% in 1993.

Income from operations for the six months ended June 30, 1994 increased to \$30.8 million or 31% from \$23.4 million in the same period of 1993. In the Fluid Handling Group, income from operations of \$25.6 million and operating margin of 21.4% for the six months ended June 30, 1994 were higher than income from operations of \$20.2 million and operating margin of 19.3% in the same period of 1993 due to volume-related gains with improving business conditions. Income from operations in the Industrial Products Group of \$8.9 million and operating margin of 14.9% in the first six months of 1994 were higher than income from operations of \$6.4 million and operating margin of 13.6% in the comparable 1993 period as all units in the Industrial Products Group experienced volume-related improvements.

Interest expense of \$5.7 million in the first six months of 1994 was approximately equivalent to the comparable 1993 period as interest rates and average borrowings outstanding under the Credit Agreement during the first six months of 1994 were approximately equivalent to the 1993 period.

The provision for income taxes increased to \$8.5 million in the first half of 1994 from \$6.1 million in the first half of 1993. The 1994 effective tax rate of 35.3% increased slightly from the 35.0% 1993 rate as the amortization of goodwill expense arising from the Hale acquisition is not deductible for tax purposes. Net income was \$15.5 million in the first six months of 1994, up 37% from \$11.3 million in the 1993 period. Earnings per common share amounted to \$1.19 in the first six months of 1994 which was 35% higher than the \$.88 recorded in the first six months of 1993.

Liquidity and Capital Resources

On June 30, 1994 IDEX's working capital was \$89.1 million and its current ratio was 2.5 to 1. Internally generated funds were adequate to fund capital expenditures of \$4.2 million and \$3.5 million for the six months ended June 30, 1994 and 1993, respectively. These expenditures were generally for machinery and equipment which improved productivity, although a portion was for repair and replacement of equipment and facilities. Management believes that IDEX has ample capacity in its plant and equipment to meet expected needs for future growth in the intermediate term. During the six months ended June 30, 1994 and 1993, depreciation and amortization expense, excluding amortization of debt issuance expenses, was \$6.3 million, and \$5.9 million, respectively.

In connection with the acquisition of Hale, the Credit Agreement was amended on May 23, 1994 to provide for an additional \$50 million of availability and improved interest rate structure. IDEX borrowed approximately \$95 million under the Credit Agreement to finance the acquisition of Hale. At June 30, 1994, the maximum amount available under the Credit Agreement was \$150 million, of which \$114 million was being used. The availability under the Credit Agreement declines in stages commencing December 31, 1995 to \$100 million on December 31, 1997. Any amount outstanding at June 30, 1999 becomes due at that date. Interest is payable quarterly on the outstanding balance at the Bank Agent's reference rate, or at rates applicable to certain dollar deposits in the interbank Eurodollar market plus 1-1/4%.

IDEX believes it will generate sufficient cash flow from operations to meet its operating requirements, scheduled amortization payments under the Credit Agreement, interest and principal payments on the Senior Subordinated Notes and approximately \$11 million of planned capital expenditures in 1994. From commencement of operations in January, 1988 until June 30, 1994, IDEX borrowed \$210 million to complete seven acquisitions and during this same period generated, principally from operation, cash flow of \$186 million to reduce its indebtedness. IDEX intends to consider additional acquisitions in the future. In the event that suitable businesses or assets are available for acquisition by IDEX upon terms acceptable to the Board of Directors, IDEX may obtain all or a portion of the financing for the acquisitions through the incurrence of additional long-term indebtedness.

Part II. Other Information

- Item 1. Legal Proceedings. None.
- Item 2. Changes in Securities. Not Applicable.
- Item 3. Defaults upon Senior Securities. None.
- Item 4. Submission of Matters to a Vote of Security Holders. None.
- Item 5. Other Information.

Set forth below is the information required by 7(a), Financial Statements of Acquired Businesses, and 7(b), Pro Forma Financial Statements of Form 8-K with respect to the Hale Products, Inc. acquisition filed with the SEC on June 6, 1994.

Unaudited Financial Statements

Consolidated Balance Sheet as of May 25, 1994 and December 31, 1993	F-1
Consolidated Statements of Operations for the period January 1 to May 25, 1994 and the six months ended June 30, 1993	F-2
Consolidated Statements of Stockholders' Equity (Deficit) for the period January 1 to May 25, 1994	F-3
Consolidated Statements of Cash Flows for the period January 1 to May 25, 1994 and the six months ended June 30, 1993	F-4
Notes to the Consolidated Financial Statements	F-5
Audited Financial Statements	
Report of Independent Auditors	F-6
Consolidated Balance Sheets as of December 31, 1993 and 1992	F-7
Consolidated Statements of Operations for the Years Ended December 31, 1993, 1992, and 1991	F-9
Consolidated Statements of Stockholders' Equity (Deficit) for the Years Ended December 31, 1993, 1992, and 1991	F-10
Consolidated Statements of Cash Flows for the Years Ended December 31, 1993, 1992, and 1991	F-11
Notes to the Consolidated Financial Statements	F-12
Pro Forma Financial Statements	F-21
Unaudited Pro Forma Statement of Operations for the twelve months ended December 31, 1993	F-22
Unaudited Pro Forma Statement of Operations for the six months ended June 30, 1994	F-23
Unaudited Pro Forma Notes to Statements of Operations	F-24

Hale Products, Inc. Consolidated Balance Sheets May 25, 1994 and December 31, 1993

	(IN]	THOUSANDS)
	MAY 25,	DECEMBER 31,
	1994	1993
ASSETS		AUDITED)
Current accets:		
Current assets:	¢ 2 26E	¢4 E02
Cash and cash equivalents Accounts receivable	\$ 3,265 8,730	\$4,583 12,452
Inventories	15,216	13,545
Prepaid expenses	637	355
Income taxes receivable	637	000
Deferred income taxes	420	449
Total current assets	28,905	31,384
		•
Property, plant and equipment, at cost	22,889	22,174
Less accumulated depreciation	10,372	9,695
Net property, plant and equipment	12,517	12,479
Other	1 651	1,566
other	1,651	1,500
	\$43,073	\$45,429
	======	======
LIABILITIES AND EQUITY (DEFICIT)		
Current liabilities:		
Accounts payable	\$4,480	\$4,466
Accrued interest	566	892
Income taxes payable	106	447
Current portion of long-term debt Other accrued liabilities	106 3,507	844 3,537
Other accided flabilities	3,307	3,331
Total current liabilities	8,659	10,186
10042 04110110 11401110100	0,000	10, 100
Deferred income taxes	1,235	1,006
Long-term debt	36,304	36,356
Stockholders' equity (deficit):		
Common stock \$.01 par value; 1,000,000 shares		
authorized, 491,935 and 486,375 shares	_	_
issued and outstanding in 1994 and 1993	5	5
Capital in excess of par value Retained earnings (deficit)	3,506 (6,663)	3,413 (5,400)
Common stock held in treasury-at cost; 12,764	(0,003)	(3,400)
shares in 1994 and 1993	(149)	(149)
Cumulative effect of foreign currency	(=.0)	(= .0)
translation	176	12
Total stockholders' equity (deficit)	(3,125)	(2,119)
	\$43,073	\$45,429
	======	======

Hale Products, Inc. Consolidated Statements of Operations for the Period January 1 to May 25, 1994 and the Six Months Ended June 30, 1993

		ousands)
	1994 199 (unaudited)	
Revenues	\$22,586	\$32,062
Cost of goods sold	15,687	20,131
Gross margin	6,899	11,931
Operating costs: Selling, general and administrative Depreciation Amortization Other	6,475 715 159	7,681 833 566 107
Operating income (loss)	(450)	2,744
Interest expense	1,399	1,844
Minority interest in net loss of subsidiary		(37)
<pre>Income (loss) before income taxes, extraordinary item, and cumulative effect of change in accounting for income taxes</pre>	(1,849)	937
Income tax (benefit) provision	(586)	471
Loss before extraordinary item and cumulative effect of change in accounting for income taxes	(1,263)	466
Extraordinary item, net of income tax benefit Cumulative effect of change in accounting		709
for income taxes		(114)
Net loss	\$(1,263) ======	\$(129) =====

Hale Products, Inc. Consolidated Statement of Stockholders' Equity (Deficit) (in thousands, except share data)

	СОММО	N STOCK	CAPITAL IN EXCESS	RETAINED EARNINGS	FOREIGN CURRENCY	TREASURY	TOTAL STOCKHOLDERS' EQUITY
	SHARES	AMOUNT	OF PAR VALUE	(DEFICIT)	TRANSLATION	STOCK	(DEFICIT)
Balance, January 1, 1994	486,375	\$5	\$3,413	\$(5,400)	\$12	\$(149)	\$(2,119)
Exercise of stock options Foreign currency translation	5,560		93	(1, 202)	164		93 164
Net loss				(1,263)			(1,263)
Balance, May 25, 1994							
(unaudited)	491,935	\$5	\$3,506	\$(6,663)	\$176	\$(149)	(\$3,125)
	======	===	======	======	====	======	======

Hale Products, Inc. Consolidated Statements of Cash Flows for the Period January 1 to May 25, 1994 and the Six Months Ended June 30, 1993

	(in thousands)	
	1994 (unau	1993 dited)
	•	
Operating activities Net loss Adjustments to reconcile net loss to net cash (used in) provided by operating activities:	\$(1,263)	\$(129)
Depreciation and amortization Noncash charges Extraordinary items Minority interest in net loss of subsidiary Cumulative effect of change in accounting for income taxes Change in operating assets and liabilities, net of extraordinary items:	874	1,399 122 709 (37) (114)
Accounts receivable Inventory Prepaid expenses Accounts payable Accrued interest Income taxes payable Other liabilities	3,840 (1,466) (250) (66) (326) (1,133) (10)	835 (1,391) (588) (2,151) 901 (391) (127)
Total adjustments	1,463	(833)
Net cash (used in) provided by operating activities	200	(962)
Investing activities Capital expenditures Acquisition of minority interest Other Net cash used in investing activities	(687) (5) (692)	(331) (102) (215) (648)
Ç .	(032)	(040)
Financing activities Proceeds from sale of senior notes Net borrowings on line of credit agreements Principal payments on long-term debt Proceeds from the exercise of stock options Purchase of treasury shares Debt fees	(806) 93	36,000 278 (33,062) 10 (3) (1,850)
Not cash (used in) provided by financing		
Net cash (used in) provided by financing activitities Effect of exchange rate changes on cash	(713) (113)	1,373 151
Net decrease in cash Cash and cash equivalents, beginning of period	(1,318) 4,583	(86) 2,491
Cash and cash equivalents, end of period	\$3,265 =====	\$2,405 =====

Hale Products, Inc. Notes to Consolidated Financial Statements May 25, 1994

1. SIGNIFICANT ACCOUNTING POLICIES

In the opinion of Hale's management, the unaudited information presented as of May 25, 1994 and for the period January 1 to May 25, 1994 and for the six months ended June 30 1993 reflects all adjustments necessary, which consists only of normal recurring adjustments, for a fair presentation of the interim period.

2. INVENTORIES

Inventories are valued at lower of cost (first-in first-out) or market and at May 25, 1994 and December 31, 1993 of the following (in thousands):

	MAY 25, 1994 (unaudited)	DECEMBER 1993
Raw Materials and finished goods Work in progress	\$12,327 2,889	\$11,364 2,181
Total inventory	\$15,216	\$13,545

3. SUPPLEMENTAL CASH FLOW INFORMATION

Cash paid during the period January 1 to May $\,$ 25, 1994 and for the six months ended June 30, 1993 was as follows (in thousands):

	MAY 25, 1994	JUNE 30 1993
		udited)
Interest	\$1,748	\$896
Income taxes	557	829

4. SUBSEQUENT EVENT

On May 26, 1994, IDEX Corporation purchased all of the company's outstanding shares of common stock for \$54 million and assumed \$36 million of senior notes.

REPORT OF INDEPENDENT AUDITORS

The Board of Directors and Stockholders Hale Products, Inc.

We have audited the accompanying consolidated balance sheets of Hale Products, Inc. as of December 31, 1993 and 1992 and the related consolidated statements of operations, stockholders' equity, and cash flows for each of the three years in the period ended December 31, 1993. These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the consolidated financial position of Hale Products, Inc. at December 31, 1993 and 1992, and the consolidated results of its operations and cash flows for each of the three years in the period ended December 31, 1993, in conformity with generally accepted accounting principles.

As discussed in Note 1 to the consolidated financial statements, the Company changed its method of accounting for income taxes in 1993.

/s/ ERNST & YOUNG ERNST & YOUNG

March 24, 1994

Consolidated Balance Sheets

	DECEMBER 31 1993 1993	
	(In Tho	usands)
ASSETS Current assets: Cash and cash equivalents Accounts receivable, net of allowance for doubtful accounts of \$171 and \$343 in 1993 and 1992, respectively Inventories	12,452	\$ 2,491 12,049 12,622 513
Prepaid expenses Deferred taxes		17
Total current assets	31,384	27,692
Property, plant and equipment, at cost: Land Building Machinery and equipment Furniture and fixtures	5,679 13,012	1,726 5,615 11,995 1,481
Less accumulated depreciation		20,817 8,006
Net property, plant and equipment	12,479	12,811
Excess of cost over net assets acquired, net of accumulated amortization of \$52 and \$2,422 in 1993 and 1992, respectively Other	103 1,463	12,481 910
	\$45,429	\$53,894
	=======	=======

Consolidated Balance Sheet (Continued)

	DECEMBER 31 1993 1992	
	(In Thou	
LIABILITIES AND STOCKHOLDERS' EQUITY (DEFICIT) Current liabilities: Accounts payable Accrued compensation Accrued pension and profit-sharing Accrued interest Income taxes payable	1,383	1,504 1,033
Current portion of long-term debt Other liabilities	892 447 844 1,098	2,680
Total current liabilities		11,902
Deferred taxes	1,006	917
Long-term debt, less current portion	36,356	33,406
Minority interest in subsidiary	-	520
Stockholders' equity (deficit): Common Stock \$.01 par value; 1,000,000 shares authorized, 486,375 and 437,702 shares issued and outstanding in 1993 and 1992 Capital in excess of par value Retained earnings (deficit) Common Stock held in treasury - at cost; 12,764 and 12,514 shares in	5 3,413 (5,400)	4 2,732 4,595
1993 and 1992 Cumulative effect of foreign currency		(145)
translation		(37)
Total stockholders' equity (deficit)		
	\$45,429 =======	•

Consolidated Statements of Operations

	YEAR ENDED DECEMBER 31 1993 1992 1991			
		Thousands		
Revenues	\$68,868	\$70,339	\$65,592	
Cost of goods sold	43,178	45,254	43,773	
Gross margin	25,690	25,085	21,819	
Operating costs: Selling, general and administrative Depreciation Amortization Goodwill write-off and other charges	1,530 1,330	14,271 1,756 1,328	11,934 1,472 1,620	
Operating (loss) income	(3,704)	7,730	6,793	
Interest expense	3,648	4,550	5,555	
Minority interest in net loss (income) of subsidiary	37	(3)	74	
(Loss) income before income taxes extraordinary items, and cumulative effect of change in accounting for income taxes		3,177	1,312	
Income tax provision	1,941	1,235	863	
(Loss) income before extraordinal items and cumulative effect of change in accounting for income taxes	ė	1,942	449	
Extraordinary items, net of income tax benefit of \$945	853	-	-	
Cumulative effect of change in accounting for income taxes	(114)	-	-	
Net (loss) income	\$(9,995) =======	\$ 1,942		===

Consolidated Statements of Stockholders' Equity (Deficit) (In thousands, except share data)

	COMMON STOCK		CAPITAL IN EXCESS	RETAINED EARNINGS	
	SHARES	AMOUNT	OF PAR VALUE		
Balance, December 31, 1990	428,235	\$4	\$2,672	\$2,204	
Purchase of treasury stock Exercise of stock options Foreign currency translation Net income	3,700	-	19	449	
Balance, December 31, 1991	431,935	4	2,691	2,653	
Purchase of treasury stock Exercise of stock options Foreign currency translation Net income	5,767	-	41	1,942	
Balance, December 31, 1992	437,702	4	2,732	4,595	
Purchase of treasury stock Exercise of stock options Stock issued in exchange for subsidiary minority interest	1,000 47,673	1	10 671		
Foreign currency translation Net loss				(9,995)	
Balance, December 31, 1993	486,375 =======		\$3,413 ===========	\$(5,400)	

	FOREIGN CURRENCY TRANSLATION	TREASURY STOCK	TOTAL STOCKHOLDERS' EQUITY (DEFICIT)
Balance, December 31, 1990	\$ 287	\$ -	\$5,167
Purchase of treasury stock Exercise of stock options Foreign currency translation Net income	(70)	(84)	(84) 19 (70) 449
Balance, December 31, 1991	217	(84)	5,481
Purchase of treasury stock Exercise of stock options Foreign currency translation Net income	(254)	(61)	(61) 41 (254) 1,942
Balance, December 31, 1992	(37)	(145)	7,149
Purchase of treasury stock Exercise of stock options Stock issued in exchange for		(4)	(4) 10
subsidiary minority interest Foreign currency translation Net loss	49		672 49 (9,995)
Balance, December 31, 1993	\$ 12 	\$ (149)	\$(2,119)

Hale Products, Inc.

Consolidated Statements of Cash Flows

	YEAR ENDED DECEMBER 31		
		(In Thousands)	
OPERATING ACTIVITIES Net (loss) income Adjustments to reconcile net (loss) income to net cash	\$(9,995)	\$1,942	\$ 449
provided by operating activities: Depreciation and amortization Goodwill write-off and other noncash charges Extraordinary items Pension fund settlement	2,860 11,838 853 (1,290)	3,084 56 - -	3,092 49 -
Minority interest in net (loss) income of subsidiary Deferred income tax provision (benefit) Cumulative effect of change in accounting for income taxes	(37) 145 (114)	3 (45)	(74) 477 -
Change in operating assets and liabilities, net of extraordinary items:	, ,		
Accounts receivable Inventory Prepaid expenses Accounts payable	(533) (1,045) 102 (261)	(3,209) 376 659 (1,216)	3,204 (475) (724) 167
Accrued interest Income taxes payable Accrued compensation Accrued pension and profit sharing	870 120 (479) (139)	(19) 412 553 138 1,367	(30) (89) (253) (35)
Other liabilities			
Total adjustments	12,309	2,159	4,198
Net cash provided by operating activities	2,314	4,101	4,647
INVESTING ACTIVITIES Capital expenditures Acquisition of GPL minority interest Other	(1,183) (102) (107)	(767) - (112)	(1,507) - 17
Net cash used in investing activities		(879)	
FINANCING ACTIVITIES Proceeds from sale of senior notes Net borrowings (payments) on line of credit agreements Principal payments on long-term debt Proceeds from the exercise of stock options Purchase of treasury shares Debt fees	12	438 (2,211) 41 (61) (152)	(37) (3,359) 19 (84)
Net cash provided by (used in) financing activities Effect of exchange rate changes on cash		(1,945) (228)	(3,461) (25)
Net increase (decrease) in cash		1,049	
Cash and cash equivalents, beginning of year	2,491	1,442	1,771
Cash and cash equivalents, end of year	\$4,583 ========	\$2,491 =========	\$1,442 ======
SUPPLEMENTAL DISCLOSURE OF CASH FLOW INFORMATION Cash paid during year for: Interest	\$2,766	\$4,459	\$6,243
Income taxes	,	\$ 628	•
	. ,	=========	

Notes to Consolidated Financial Statements

December 31, 1993

1. SIGNIFICANT ACCOUNTING POLICIES

ORGANIZATION

Hale Products, Inc. (the Company) is a worldwide manufacturer and distributor of products and services for the fire and rescue industry. The Company's wholly owned subsidiaries are Hale Fire Pump Company (HFP), American Godiva, Inc. (AGI), and Hale Products Finance, Inc. (HPF), all of which are located in the United States, and Godiva Products Limited (GPL), located in the United Kingdom.

The financial statements of those subsidiaries have been included in the Company's consolidated financial statements. The United States operations represent approximately 71% of the Company's revenues in 1993 and 1992 and 67% in 1991, and the United Kingdom represents the remaining 29% in 1993 and 1992 and 33% in 1991. All intercompany transactions have been eliminated in consolidation.

CASH AND CASH EQUIVALENTS

The Company considers all highly liquid investments with a maturity of three months or less at the time of purchase to be cash equivalents.

CAPITALIZED LEASES

Assets recorded under capital leases are amortized over the initial or remaining term of the related leases by the straight-line method. Amortization of assets recorded under capital leases is included within depreciation expense.

DEPRECIATION

For financial statement purposes, depreciation is determined on the straight-line method. The estimated useful lives for depreciation are as follows:

Building 31-50 years
Machinery and equipment 2-13 years
Furniture and fixtures 5-13 years

The Company uses accelerated depreciation methods for income tax purposes.

EXCESS OF COST OVER NET ASSETS ACQUIRED

Excess of cost over net assets acquired arising from the acquisition of HFP is being amortized by the straight-line method over a period of 20 years. In 1993, the excess of cost over net assets acquired related to the acquisition of GPL was written off (see note 3).

1. SIGNIFICANT ACCOUNTING POLICIES (CONTINUED)

DEFERRED INCOME TAXES

The Company provides deferred income taxes on the differences between financial accounting and taxable income. The differences principally relate to the timing of depreciation, pension, and warranty expense.

In January 1993, the Company adopted Financial Accounting Standards Board Statement No. 109, "Accounting for Income Taxes." The adoption of this Statement caused a change in the Company's method of accounting for deferred income taxes as the new standard requires the liability method of accounting for income taxes.

FOREIGN CURRENCY TRANSLATION

Assets and liabilities of foreign entities are translated using exchange rates in effect at the balance sheet date and operations are translated using average exchange rates for the year. Translation gains and losses are recorded in stockholders' equity, and transaction gains and losses (not significant in amount) are included in operating results for the year.

RECLASSIFICATION OF PRIOR YEAR AMOUNTS

Certain balances have been reclassified, where appropriate, to conform with the 1993 presentation.

2. INVENTORIES

Inventories are valued at lower of cost (first-in, first-out) or market and at December 31 consisted of the following (in thousands):

	1993	1992
Raw materials and finished goods Work in progress	. ,	\$10,392 2,230
Total inventory	\$13,545 	\$12,622

3. GOODWILL WRITE-OFF

As a result of an analysis of the continuing value of its remaining excess of cost over net assets acquired, the Company recorded a charge of \$11,716,000 for the year ended December 31, 1993 related to the excess of cost over net assets acquired of GPL. The primary factor in evaluating the continuing value was an assessment that cash flows from GPL's operations would not be sufficient to recover such excess cost. In that regard, since its acquisition in 1989, GPL has not achieved the sales, earnings or cash flow projections $\frac{1}{2}$ prepared at the time of the acquisition. In addition, future earnings and cash flow projections continue to deviate significantly from the original projections. A second factor relates to the management team which was in place at the time of the acquisition. The management team was assumed to contribute to the continued development of the acquired business, however, a number of the original members have had to be replaced in 1993. The third factor relates to industry standards that were in place at the time of the acquisition. The Company anticipated significant savings from the acquisition of an established product line which would require minimal additional development costs. In late 1993, the European regulatory body responsible for setting fire pump standards modified those standards which negated the anticipated savings. Accordingly, the Company's management determined that the excess of cost over net assets acquired related to the GPL acquisition was not recoverable.

4. RETTREMENT PLANS

The Company sponsors a defined benefit pension plan for the bargaining unit employees in the United States and a contributory defined benefit pension plan for all eligible employees in the United Kingdom. In the pension plan for United States bargaining unit employees, benefits are based on years of service and the employee's compensation during the last five years of employment. Benefits for eligible United Kingdom employees are based on compensation during the employee's final year of service. Employee contributions are based on 2% of current salary. Assets of the United States plan are comprised principally of equity securities and fixed income investments. Assets of the United Kingdom plan are comprised of money market funds.

The Company's policy is to annually contribute an amount between the minimum required by law and the maximum which can be deducted for income tax purposes. Contributions are intended to provide not only for benefits attributed to service to date but also for those expected to be earned in the future.

The provisions of Financial Accounting Standards Board Statement No. 87, "Employers Accounting for Pensions," require recognition of an additional minimum liability and related intangible asset for pension plans with accumulated benefits in excess of plan assets. At December 31, 1993 an additional liability of \$192,000 and an intangible asset of equal value are reflected in the consolidated balance sheet.

4. RETIREMENT PLANS (CONTINUED)

Components of net periodic pension for the year ended December 31 were:

	199	3	199	2	199:	1
	U.S. PLAN	U.K. PLAN	U.S. PLAN	U.K. PLAN	U.S. PLAN	U.K. PLAN
		(Do	llars in	thousan	ds)	
Service cost - benefits earned						
during the period	\$138	\$195	\$150	\$246	\$151	\$214
Interest cost	171	390	172	440	160	427
Actual return on plan assets	(332)	(435)	(74)	(493)	(268)	(481)
Net amortization and deferral	167	(60)	(101)	(70)	112	(71)
Net periodic pension cost	\$144	\$ 90	\$147	\$123	\$155	\$ 89
	=====	======	======	======	=======	=====

The funded status of these plans at December 31 was:

	1	1993		992
			U.S. PLAN	
Plan assets at fair value	\$2,239	\$4,011	\$2,075	\$4,153
Actuarial present value of benefit obligations: Vested benefits Nonvested	11	533	\$1,510 21	423
Accumulated benefit obligation Additional benefits based on future salary increases	1,764 1,025	4,455 192	1,531 824	3,503 181
Projected benefit obligation			2,355	
Projected benefit obligation (in excess of) less than plan assets Unrecognized net loss (gain) Unrecognized prior service cost Unrecognized transition asset Additional minimum liability recognized	` 20´ 416	1,213 - (533)	(280) (258) 449 (354)	166 - (604)
Net (accrued) prepaid liability at year end	\$ (434) ======	\$ (148) ======	\$ (443) ======	\$ 31 ======

4. RETIREMENT PLANS (CONTINUED)

Key economic assumptions used in these determinations were as follows:

	19	1993		92
	U.S.	U.K.	U.S.	U.K.
	PLAN	PLAN	PLAN	PLAN
Discount rate: January 1 December 31	8.00%	10.50%	8.00%	10.75%
	7.00	8.25	8.00	10.50
Rate of increase in compensation levels Expected long-term rate of return:	4.00	6.50	4.00	6.50
January 1	8.00	10.50	8.00	10.50
December 31	7.00	10.50	8.00	10.50

The Company also sponsors a defined contribution pension, profit-sharing and savings plan covering substantially all of the United States non bargaining unit employees. Contributions are determined each year by the Board of Directors. Expense for the years ended December 31, 1993, 1992 and 1991 was \$490,000 \$556,000 and \$515,000, respectively.

5. INCOME TAXES

Income taxes have been provided as follows:

	1993 (Ti	1992 n Thousands	1991 \
Expected tax (benefit) at statutory tax rate	\$(2,077)	\$1,080	\$446
State taxes net of federal benefit	274	175	68
Net foreign losses not available to offset			
domestic taxes	-	109	45
Permanent differences	58	31	13
Amortization of foreign goodwill (a)	3,831	273	276
FSC Benefit	(48)	(49)	(20)
Foreign tax refund (b)	(56)	(495)	-
Other	(41)	111	35
	\$ 1,941	\$1,235	\$863
	========		======

- (a) In 1993, the Company wrote off all goodwill relating to GPL. The non-recurring charge of \$11,716,000 provided no tax benefit.
- (b) In 1992, The Company received a tax refund related to the pre-acquisition period of its subsidiary in the United Kingdom. In accordance with the original purchase agreement, part of the proceeds were payable to the previous owner and the remainder has been shown as a reduction to the 1992 tax provision.

5. INCOME TAXES (CONTINUED)

19	93				
_	_	_	_	_	_

		DEFERRED	
		In Thousands)	
Federal State Foreign	\$1,052 317 427	\$120 25 -	\$1,172 342 427
	\$1,796	\$145	
1992	=======	========	
		DEFERRED	
		In Thousands)	
Federal State Foreign	\$1,155 273 (148)	\$(34) (8) (3)	\$1,121 265 (151)
	,	\$(45)	•
1991			
	CURRENT	DEFERRED	TOTAL
	(In Thousands)	
Federal State Foreign	51	\$215 52 273	\$414 103 346
	\$323	\$540	\$863

At December 31, 1993, the Company had approximately \$904,000 of foreign net operating loss (NOL) carryforwards available to reduce future foreign taxable income. The NOL carryforwards have no expiration date.

6. EXTRAORDINARY ITEMS

At the time of the original acquisition of GPL by the Company, an agreement had been entered into with the seller to fund the acquired subsidiary's pension plan based on an actuarial valuation to be performed subsequent to the acquisition. The Company had also entered into an agreement to share tax refunds originating prior to the acquisition. In 1992, the Company received a tax refund, of which the seller was entitled to \$929,000. In 1993, the actuarial valuation was finalized and the pension funding became due, however, the Company became concerned with the seller's ability to fund the pension plan and entered into an agreement by which the Company

6. EXTRAORDINARY ITEMS (CONTINUED)

would settle the seller's pension liability in exchange for the Company keeping the full tax refund. The pension funding, including related professional fees, is \$1,607,000. The Company received a tax benefit related to the expense of \$534,000 and since the \$929,000 credit arose as the result of a tax refund, there was no tax effect.

In connection with the refinancing of substantially all of the Company's long-term debt, the Company incurred an extraordinary loss of \$709,000 which is net of a tax benefit of \$411,000. The charge resulted from the prepayment penalties on extinguishing the refinanced debt and the write-off of unamortized debt discount.

7. LONG-TERM DEBT

	1993	1992
Senior Notes due 2003 Various notes, refinanced March 1993 Capitalized lease obligations Unsecured, interest-free note payable to vendor due September 1994	\$36,000 - 460 740	\$ - 33,223 423 755
Less amounts due within one year	37,200 844 \$36,356	34, 401 995 \$33, 406
	========	

On March 26, 1993, the Company refinanced substantially all of its long-term debt with the proceeds from \$36,000,000 of Senior Notes due 2003. The notes bear interest at rates between 9.28% and 9.86%. Principal is payable in annual installments of \$5,775,000 beginning in March 1996 and interest is payable semiannually in arrears, each March and September.

The Company has established a \$4,500,000 revolving credit facility with a bank. Borrowings under the revolving credit facility bear interest at the bank's prime rate plus 1.5%. There were no borrowings during 1993. The Company is charged a fee of .05% on the unused balance, as defined.

The Senior Notes and revolving credit facility require certain levels of net worth, working capital, and the maintenance of certain financial ratios.

7. LONG-TERM DEBT (CONTINUED)

Aggregate scheduled maturities of long-term debt during the next five years and thereafter are as follows (in thousands):

1994	\$	844
1995		124
1996	5	,901
1997	5	,841
1998	5	, 815
Thereafter	18	675
	\$37	,200

Standby letters of credit are issued by the Company during the ordinary course of business through banks as required by certain vendor contracts. At December 31, 1993, the Company had outstanding standby letters of credit for \$1,093,000.

8. OPERATING LEASES

The Company has operating leases for supplemental office space, machinery and equipment which expire at various dates. Rental expense incurred for all operating leases was \$644,000, \$621,000 and \$641,000 in 1993, 1992, and 1991, respectively.

1994	\$387
1995	255
1996	144
1997	95
1998	7:
	\$952
	====

9. STOCK OPTIONS AND WARRANTS

The Company has a stock option plan which provides for the granting of incentive stock options to officers and key employees to purchase shares of common stock at prices ranging from \$5.00 to \$10.26. Transactions involving the plan are summarized as follows:

	1993	1992
Shares outstanding at January 1 Granted Exercised Expired	1,000 - (1,000)	5,759 2,500 (5,767) (1,492)
Shares outstanding at December 31	-	1,000

Hale Products, Inc.

Notes to Consolidated Financial Statements (continued)

9. STOCK OPTIONS AND WARRANTS (CONTINUED)

In connection with the acquisition of HFP, the Company issued warrants which enable the purchase of 100,000 shares of the Company's common stock for an aggregate price of \$1,000. The warrants expire March 26, 2000 and all 100,000 were outstanding at December 31, 1993.

10. RELATED PARTY TRANSACTIONS

The Company receives management services from a merchant banking group. Management fees charged to the Company for such services were \$425,000 in 1993 (of which \$175,000 related to the refinancing of the Company's debt in March 1993) and \$200,000 in 1992 and 1991.

11. RESEARCH AND DEVELOPMENT COSTS

The Company incurred total research and development expense of \$1,109,000, \$1,065,000 and \$1,108,000 for the years ended December 31, 1993, 1992 and 1991, respectively.

During 1992, the Company received fees totaling \$154,000 for reimbursement of research and development performed for others.

IDEX CORPORATION AND HALE PRODUCTS
UNAUDITED PRO FORMA STATEMENTS OF OPERATIONS
FOR THE TWELVE MONTHS ENDED DECEMBER 31, 1993 AND THE SIX
MONTHS ENDED JUNE 30, 1994

The following unaudited pro forma combined statements of operations for the twelve months ended December 31, 1993 and the six months ended June 30, 1994 give effect to the acquisition by IDEX of the common stock of Hale Products, Inc. ("Hale") as if the acquisition had occurred on January 1, 1993. The transaction was accounted for as a purchase in accordance with the provisions of Accounting Principles Board Opinion No. 16.

The historical financial data included in the pro forma statements is as of the periods presented. The historical financial data of Hale included in the pro forma statement of operations for the twelve months ended December 31, 1993 was derived from audited financial statements for the year ended December 31, 1993. The historical financial data of Hale for the six months ended June 30, 1994 was derived from unaudited financial statements for the period January 1 to May 25, 1994.

The unaudited pro forma financial data is based on management's best estimate of the effects of the acquisition of Hale. Pro forma adjustments are based on currently available information; however, the actual adjustments will be based on more precise appraisals, evaluations and estimates of fair values. It is possible that the actual adjustments could differ substantially from those presented in the unaudited pro forma combined financial statements.

The unaudited pro forma statement of operations for the twelve months ended December 31, 1993, and the six months ended June 30, 1994, are not necessarily indicative of the results of operations that actually would have been achieved had the acquisition of Hale been consummated as of the dates indicated, or that may be achieved in the future. The unaudited pro forma financial statements should be read in conjunction with the accompanying notes and historical financial statements and notes thereto.

Hale's financial performance for the period January 1 to May 25, 1994, prior to acquisition by IDEX, was adversely affected by several factors. Customarily, Hale's shipments are stronger in the second half of a calendar year than the first half due to the purchasing practices of customers in industries that it serves. In 1994, shipments were further reduced by production curtailments at the Conshohocken facilities because of severe winter weather and unexpected facility repairs at its foundry. In addition, Hale was in the process of moving production of certain products between its Conshohocken, Pennsylvania and St. Joseph, Tennessee facilities during this period which created certain temporary inefficiencies and loss of overhead absorption. Higher than normal selling, general and administrative expenses were incurred during this period due to Hale's participation in a major international fire and rescue trade show which is held every six years in Germany. During the period January 1 through May 25, 1994 order activity remained strong as sales backlogs increased by \$5.5 million.

IDEX CORPORATION UNAUDITED PRO FORMA STATEMENT OF OPERATIONS FOR THE TWELVE MONTHS ENDED DECEMBER 31, 1993 (IN THOUSANDS EXCEPT PER SHARE INFORMATION)

	IDEX HISTORICAL	HALE HISTORICAL	PRO FORMA ADJUST- MENTS	PRO FORMA
Net sales Cost of sales	\$ 308,638 190,286	\$ 68,868 43,178	\$ 1,292 (1)	\$377,506 234,756
Gross profit	118,352		(1,292)	
Selling, general, and administrative expenses Depreciation Amortization and other charges Goodwill write-off	68,217	14,711 1,530 1,437 11,716	60 (2) (1,530)(3) (1,437)(4) ***	- - 11,716
Income (loss) from operations Other expensenet	50,135 830		1,615 1,538 (4)	48,046
Income (loss) before interest Interest expense		(3,667)	77 1,053 (5)	45,715
Income (loss) before income taxes, extraordinary item, and cumulative effect of change in accounting for income taxes Provision for income taxes	38,298 12,972	(7,315) 1,941	(976) (4,110)(6)	10,803
Income from continuing operations	. ,	,	\$ 3,134	. ,
Earnings per common share	======= \$ 1.97 =======	=====	=====	\$ 1.49
Weighted average shares outstanding	12,878 ======			12,878 ======

***The Hale 1993 financial statements include a one-time charge of \$11.7 million to write-off the unamortized balance of goodwill deemed unrecoverable relating to Godiva Products Limited, the United Kingdom subsidiary of Hale. This charge, while not an extraordinary item that would be reflected as an adjustment to these pro forma financial statements, is a nonrecurring charge having no effect on the continuing operations of Hale. Had this charge been a pro forma adjustment, the 1993 pro forma financial statements would have reflected the following:

Income from continuing operations \$26,703
Earnings per share \$ 2.07

IDEX CORPORATION UNAUDITED PRO FORMA STATEMENT OF OPERATIONS FOR THE SIX MONTHS ENDED JUNE 30, 1994 (IN THOUSANDS EXCEPT PER SHARE INFORMATION)

		HALE HISTORICAL		PRO FORMA
Net sales Cost of sales	,	\$ 22,586 15,687	\$ 646 (1)	\$202,019 126,219
Gross profit Selling, general, and administrative expenses Depreciation Amortization	,	6,899 6,475	(646) 55 (2) (715)(3) (159)(4)	75,800 45,311 -
Income (loss) from operations Other expensenet	30,766 1,029	(450)	173 769 (4)	•
Income (loss) before interest Interest expense	29,737 5,746	(450) 1,399	(596) 905 (5)	
Income (loss) before income taxes Provision for income taxes		(1,849) (586)		20,641
Income from continuing operations	\$ 15,525 ======		(1,052) =====	
Earnings per common share	\$ 1.19 ======			\$ 1.01 ======
Weighted average shares outstanding	13,035 ======			13,035 ======

IDEX CORPORATION UNAUDITED PRO FORMA NOTES TO STATEMENTS OF OPERATIONS (IN THOUSANDS EXCEPT PER SHARE INFORMATION)

NOTES:

- (1) Represents depreciation expense on Hale's stepped up value of property, plant and equipment classified in cost of sales.
- (2) Represents the estimated effect from the acquisition of Hale relating to depreciation expense (\$431 for the twelve months and \$215 for the six months) on stepped up value of property, plant and equipment classified in selling, general and administrative expenses. Elimination of certain Hale corporate operating expenses (\$500 for the twelve months and \$225 for the six months) which will no longer be incurred as a result of the Hale acquisition. Amortization on \$4.5 million of other intangibles (\$129 for the twelve months and \$65 for the six months) arising from the acquisition of Hale.
- (3) Represents the elimination of Hale's historical depreciation expense.
- (4) Represents the reclassification and net increase of amortization expense and other charges resulting from the acquisition of Hale. The excess of the purchase price over the fair value of the net assets acquired of \$61.5 million will be amortized over 40 years.
- (5) Represents the estimated effect on interest expense (\$4,740 for the twelve months and \$2,370 for the six months) from the \$94.8 million of borrowings under the IDEX Credit Agreement to finance the acquisition of Hale at an effective borrowing cost of approximately 5%. The elimination of interest expense (\$3,487 for the twelve months and \$1,365 for the six months) on Hale's \$36 million Senior Notes retired by IDEX. Reduction of interest expense (\$200 for the twelve months and \$100 for the six months) from Hale's application of cash flow from operations to reduce indebtedness.
- (6) Represents the tax effect of the pro forma adjustments described above.

Item 6. Exhibits and Reports on Form 8-K

(a) Exhibits

The exhibits listed in the accompanying "Exhibit Index" are filed as part of this report.

(b) Reports on Form 8-K

On June 6, 1994, the company filed a current report on Form 8-K (item 2 and 7) stating that the Company purchased for cash all of the outstanding shares of common stock of Hale Products, Inc., on May 26, 1994.

Under Item 5, Other Information, of this Form 10-Q, the Company filed amended items 7(a) and 7(b) to Form 8-K with respect to audited financial statements of Hale Products, Inc. for the years ended December 31, 1993, 1992 and 1991, unaudited interim financial statements of Hale products, Inc. for the period January 1 to May 25, 1994 and for the six months ended June 30, 1993, and unaudited pro forma financial statements of IDEX Corporation for the twelve months ended December 31, 1993 and for the six months ended June 30, 1994, giving effect to the Hale Products, Inc. acquisition as provided therein.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this Report to be signed on its behalf by the undersigned, thereunto duly authorized in the capacity and on the date indicated.

IDEX CORPORATION

August 9, 1994

/s/Wayne P. Sayatovic

Wayne P. Sayatovic Vice President-Finance, Chief Financial Officer and Secretary (Duly Authorized and Principal Financial Officer)

Exhibit Number	Description
4.1	Restated Certificate of Incorporation of IDEX (formerly HI, Inc.) (incorporated by reference to Exhibit No. 3.1 to the Registration Statement on Form S-1 of IDEX Corporation, et al., Registration No. 33-21205, as filed on April 21, 1988).
4.1(a)	Amendment to Restated Certificate of Incorporation of IDEX (incorporated by reference to Exhibit No. 3.2 to Amendment No. 1 to the Registration Statement on Form S-1 of IDEX Corporation, Registration No. 33-28317, as filed on June 1, 1989).
4.2	Amended and Restated Bylaws of IDEX (incorporated by reference to Exhibit No. 3.2 to Post-Effective Amendment No. 2 to the Registration Statement on Form S-1 of IDEX Corporation, et al., Registration No. 33-21205, as filed on July 17, 1989).
4.2(a)	Amended and Restated Article III, Section 13 of the Amended and Restated Bylaws of IDEX (incorporated by reference to Exhibit No. 3.2(a) to Post-Effective Amendment No. 3 to the Registration Statement on Form S-1 of IDEX Corporation, et al., Registration No. 33-21205, as filed on February 12, 1990).
4.3	Indenture, dated as of September 15, 1992, among IDEX, the Subsidiaries and the Connecticut National Bank, as Trustee, relating to the 9-3/4% Senior Subordinated Notes of IDEX due 2002 (incorporated by reference to Exhibit No. 4.2 of the Annual Report of IDEX Corporation on Form 10-K for the fiscal year ended December 31, 1992, Commission File No. 1-10235).
4.4	Specimen Senior Subordinated Note including specimen guarantee (incorporated by reference to Exhibit No. 4.3 of the Annual Report of IDEX Corporation on Form 10-K for the fiscal year ended December 31, 1992, Commission File No. 1-10235).
4.5	Specimen certificate of Common Stock (incorporated by reference to Exhibit 4.3 to the Registration Statement on Form S-2 of IDEX Corporation, et al., Registration No. 33-42208, as filed on September 16, 1991).
10.1	Second Amended and Restated Credit Agreement dated as of January 29, 1993 among IDEX, various banks named therein and Continental Bank N.A., as Agent (incorporated by reference to Exhibit 10.1 to the Annual Report of IDEX on Form 10-K for the fiscal year ending December 31, 1992, Commission File No. 1-10235).
10.2	Pledge Agreement, dated January 22, 1988, between IDEX and the Bank Agent (incorporated by reference to Exhibit No. 10.3 to the Registration Statement on Form S-1 of IDEX Corporation, et al., Registration No. 33-21205, as filed on April 21, 1988).
10.3	Guaranty Agreement, dated January 22, 1988, between each of the Guarantors named therein and the Bank Agent (incorporated by reference to Exhibit No. 10.4 to the Registration Statement on Form S-1 of IDEX Corporation, et al., Registration No. 33-21205, as filed on April 21, 1988).

Page

- 10.3(a) Guaranty Agreement, dated May 7, 1991, by CIC Acquisition Corporation in favor of the Bank Agent (incorporated by reference to Exhibit No. 10.3(a) to the Registration Statement on Form S-1 of IDEX Corporation, et al., Registration No. 33-50220, as filed on July 29, 1992).
- 10.3(b) Guaranty Agreement, dated May 4, 1992, by PLF Acquisition Corporation and MCL Acquisition Corporation in favor of the Bank Agent (incorporated by reference to Exhibit No. 10.3(b) to the Registration Statement on Form S-1 of IDEX Corporation, et al., Registration No. 33-50220, as filed on July 29, 1992).
- Inter-Guarantor Agreement, dated as of January 22, 1988, among the Subsidiaries named therein and the Bank Agent (incorporated by reference to Exhibit 4.8 to the Registration Statement on Form S-1 of IDEX Corporation, et al., Registration No. 33-21205, as filed on April 21, 1988).
- 10.4(a) First Amendment to Inter-Guarantor Agreement, dated as of May 7, 1991, among IDEX Corporation and the Subsidiaries named therein (incorporated by reference to Exhibit No. 10.6(a) to the Registration Statement on Form S-1 of IDEX Corporation, et al., Registration No. 33-50220, as filed on July 29, 1992).
- Amended and Restated Employment Agreement between IDEX Corporation and Donald N. Boyce, dated as of January 22, 1988 (incorporated by reference to Exhibit No. 10.15 to Amendment No. 1 to the Registration Statement on Form S-1 of IDEX Corporation, Registration No. 33-28317, as filed on June 1, 1989).
- 10.5(a) First Amendment to the Amended and Restated Employment Agreement between IDEX Corporation and Donald N. Boyce, dated as of January 13, 1993 (incorporated by reference to Exhibit 10.5(a) to the Annual Report of IDEX on Form 10-K for the fiscal year ending December 31, 1992, Commission File No. 1-10235).
- Amended and Restated Employment Agreement between IDEX Corporation and Wayne P. Sayatovic, dated as of January 22, 1988 (incorporated by reference to Exhibit No. 10.17 to Amendment No. 1 to the Registration Statement on Form S-1 of IDEX Corporation, Registration No. 33-28317, as filed on June 1, 1989).
- 10.6(a) First Amendment to the Amended and Restated Employment Agreement between IDEX Corporation and Wayne P. Sayatovic, dated as of January 13, 1993 (incorporated by reference to Exhibit 10.7(a) to the Annual Report of IDEX on Form 10-K for the fiscal year ending December 31, 1992, Commission File No. 1-10235).
- Management Incentive Compensation Plan (incorporated by reference to Exhibit No. 10.21 to Amendment No. 1 to the Registration Statement on Form S-1 of IDEX Corporation, Registration No. 33-28317, as filed on June 1, 1989).
- 10.8 Form of Indemnification Agreement (incorporated by reference to Exhibit No. 10.23 to the Registration Statement on Form S-1 of IDEX Corporation, Registration No. 33-28317, as filed on April 26, 1989).

- 10.9 Form of Shareholder Purchase and Sale Agreement (incorporated by reference to Exhibit No. 10.24 to Amendment No. 1 to the Registration Statement on Form S-1 of IDEX Corporation, Registration No. 33-28317, as filed on June 1, 1989).
- 10.10 Revised Form of IDEX Corporation Stock Option Plan for Outside Directors (incorporated by reference to Exhibit No. 10.22(a) to Post-Effective Amendment No. 4 to the Registration Statement on Form S-1 of IDEX Corporation, et al., Registration No. 33-21205, as filed on March 2, 1990).
- Amendment to the IDEX Corporation Stock Option Plan for Outside Directors, adopted by resolution of the Board of Directors dated as of January 28, 1992 (incorporated by reference to Exhibit 10.21(a) of the Annual Report of IDEX on Form 10-K for the fiscal year ended December 31, 1991, Commission File No. 1-10235).
- 10.12 Non-Qualified Stock Option Plan for Non-Officer Key Employees of IDEX Corporation (incorporated by reference to Exhibit 10.15 to the Annual Report of IDEX on Form 10-K for the fiscal year ending December 31, 1992, Commission File No. 1-102351).
- 10.13 Non-Qualified Stock Option Plan for Officers of IDEX Corporation (incorporated by reference to Exhibit 10.16 to the Annual Report of IDEX on Form 10-K for the fiscal year ending December 31, 1992, Commission File No. 1-102351).
- 10.14 IDEX Corporation Supplemental Executive Retirement Plan (incorporated by reference to Exhibit 10.17 to the Annual Report of IDEX on Form 10-K for the fiscal year ending December 31, 1992, Commission File No. 1-102351).
- Asset Purchase Agreement, dated as of February 19, 1991, by and among Corken International Corporation, Corken Properties, Inc., Hinderliter Industries, Inc., CIC Acquisition Corporation and IDEX (incorporated by reference to Exhibit 10.23 to the Annual Report of IDEX on Form 10-K for the fiscal year ended December 31, 1990, Commission file number 33-21205).
- Asset Purchase Agreement, dated as of April 9, 1992 by and among PLF Acquisition Corporation, O.D.E. Manufacturing, Inc., Pulsafeeder, Inc., Morr Control, Inc., Pulsafeeder Far East Pte. Ltd. and PAC, Inc. (incorporated by reference to Exhibit No. 2.1 to the Current Report of IDEX on Form 8-K filed with the Commission on May 20, 1992, Commission File No. 1-10235).
- *10.17 Stock Purchase Agreement, dated as of May 6, 1994 by and among HPI Acquisition Corp., HFP Partners, L.P., HMTC Partners L.P., the persons listed on Schedule A and Hale Products, Inc.
- *10.18 First Amendment dated as of May 23, 1994 to second amended and restated Credit Agreement dated as of January 29, 1993 by and among IDEX Corporation, various banks named therein and Continental Bank N.A., as agent.
- *24.1 Consent of Ernst & Young

*Filed herewith

1

STOCK PURCHASE AGREEMENT

Dated as of May 6, 1994

By and Among

HPI ACQUISITION CORP. (a Delaware Corporation),

HFP PARTNERS, L.P. (a Delaware limited partnership),

HMTC PARTNERS, L.P. (a Delaware limited partnership),

THE PERSONS LISTED ON SCHEDULE A

and

HALE PRODUCTS, INC. (a Delaware Corporation)

									Page
	DEF	RTICLE :	NS						
1.1	Defined Terms				 	 	 	 	. 1
1.2	Other Defined Terms				 	 	 	 	. 7
1.3	Usage of Terms				 	 	 	 	. 8
1.4	References to Articles, Sections, Exhibits and								
	Schedules				 	 	 	 	. 8
1.5	 Appointment of Management Agent and Seller Agen	nt			 	 	 	 	. 8
1.6	Conversion Rates								
		RTICLE 2							
	PURCHASE AN								
2.1	Transfer of Shares				 	 	 	 	. 10
2.2	Purchase Price				 	 	 	 	. 10
2.3	Pre-Acquisition Financial Review				 	 	 	 	. 10
2.4	Payment of Purchase Price				 	 	 	 	. 11
2.5	Taxes				 	 	 	 	. 11
	C	RTICLE S CLOSING	3						
3.1	Closing				 	 	 	 	. 11
3.2	Stock Certificates and Instruments of Assignmen				 	 	 	 	. 11
3.3	Warrants				 	 	 	 	. 11
3.4	Purchase Price; Certificates				 	 	 	 	. 11
	AR REPRESENTATI OF SELLERS A	RTICLE 4 CONS AND AND THE) WARRA	NTIES ATION	 	 	 	 	. 12
4.1	Organization, Good Standing and Authority of th								
	Business							 	. 12
4.2	Power and Authority; Authorization; Binding								
	Effect				 	 	 	 	. 12
4.3	No Conflict or Violation				 	 	 	 	. 13
4.4	Consents and Approvals				 	 	 	 	. 13
4.5	No Proceedings				 	 	 	 	. 14
4.6					 	 	 	 	. 14
4.7	Corporate Records				 	 	 	 	. 14
4.8	Financial Statements								
4.9	Real Property								
4.10	Tangible Personal Property								
4.11	Intangible Property								
4.12	Compliance with Laws; Permits								
4.12	Litigation								
+.⊥3					 	 	 	 	. 21

	Pa	age
4.14	Labor Matters; Employee Contracts	22
4.15A	United States Employee Benefit Plan	25
4.15B	United Kingdom Employee Benefit Plans	29
4.16	Transactions with Certain Persons	32
4.17A	Tax Matters - General	33
4.17B	Tax Matters - United Kingdom	34
4.18	Insurance	35
4.19	Inventory	36
4.20	Accounts Receivable	36
4.21		36
4.22	Suppliers and Customers	38
4.23	Business Records	38
4.24	Bank Accounts, Directors and Officers	39
4.25	Environmental Matters	39
4.26		40
4.27	 No Brokers	41
4.28	 Absence of Certain Payments	41
4.29	Products; Product Warranties	41
4.30	Material Misstatements or Omissions	
4.31	Power and Authority; Authorized; Binding Effect	43
4.32	No Conflict or Violation	
4.33	Consents and Approvals	43
4.34	No Proceedings	
4.35	No Encumbrances	
	ARTICLE 5	
	REPRESENTATIONS AND WARRANTIES OF BUYER	
5.1	Organization and Good Standing	
5.2	Authorization; Binding Effect	
5.3	No Conflict or Violation	44
5.4	Consents and Approvals	45
5.5	No Proceedings	45
5.6	No Brokers	45
5.7	Acquisition for Investment	45
	ARTICLE 6	
	COVENANTS AND CONDUCT OF THE PARTIES PRIOR TO CLOSING	46
6.1	Investigation by Buyer	46
6.2	Environmental Audits	46
6.3	Matters Relating to Real Property	47
6.4	Consents and Best Efforts	47
6.5	Conduct of Business by the Corporation and Sellers Pending Closing	47
6.6	Notification of Certain Matters	48
6.7	Delivery of Interim Financial Statements	48

	Page	
6.8	Covenants of Certain Others	9
5.9	Distributions; Related Person Transactions;	
	Brokers	9
	ARTICLE 7 CONDITIONS TO SELLERS' OBLIGATIONS	
7.1	Representations, Warranties and Covenants	0
7.2	Consents, Approvals and Filings	0
7.3	No Proceedings	1
7.4	Closing Certificate	1
7.5	Legal Opinion	1
7.6	HSR Act	1
7.7	TCW Warrants	1
	ARTICLE 8 CONDITIONS TO BUYER'S OBLIGATIONS	
	CONDITIONS TO BUTER S OBLIGATIONS	
8.1	Representations, Warranties and Covenants	1
8.2	Consents, Approvals and Filings	2
8.3	No Proceedings	2
8.4	No Interruption or Adverse Change	2
8.5	Closing Certificate	2
8.6	Legal Opinion	2
8.7	Resignations	
8.8	HSR Act	3
8.9	[Reserved]	
8.10	Restrictive Covenants	3
8.11	Terms of Closing Date Funded Debt	3
8.12	TCW Warrants	3
8.13	Certificates Pursuant to Article 4(B) and Additional Matters	3
	ARTICLE 9	
	CONFIDENTIALITY; INDEMNIFICATION; FURTHER ASSURANCES	
9.1	Confidentiality Agreement	4
9.2	Survival and Indemnifications	
9.3	Further Assurances	8
9.4	Liability of Sellers in Certain Circumstances	
	ARTICLE 10 MISCELLANEOUS	
10.1	Termination	8
10.2	Notices	9
10.3	Arbitration	0

																					Page
10.4	Knowledge																				 66
10.5	Public Statements																				 61
10.6	Choice of Law																				 61
10.7	Equitable Remedies																				 61
10.8	Expenses																				 61
	Titles																				
	Waiver																				
	Binding																				
10.12	Entire Agreement .																				 62
10.13	Modification																				 62
10.14	Counterparts																				 63
10.15	Consent to Jurisdi	cti	Lon																		 63

SCHEDULES

Schedule	Description
A 4.1 4.3 4.4 4.5 4.6 4.7 4.8 4.9	Sellers Subsidiaries; Qualification; Assumed Names; Equity Interests Contractual Conflicts Consents, Approvals Proceedings Capitalization Corporate Records Financial Statements Real Property, Liens, Encumbrances and Proceedings
4.10(a) 4.10(b) 4.11 4.12 4.13 4.14 4.15A(a)(1) 4.15A(a)(2)	Owned Tangible Personal Property Leased Tangible Personal Property Intangible Property Permits, Licenses Litigation Labor Matters; Employee Contracts Pension Plans Welfare Plans
4.15A(g) 4.15B 4.16 4.17A 4.17B 4.18 4.19	Post-Retirement Benefits U.K. Employee Benefit Plans Transactions with Certain Persons Tax Matters - General Tax Matters - U.K. Insurance Inventory Not Located at the Real Property; Inventory Not Owned by the
4.20 4.21 4.22 4.24 4.25 4.26 4.29 5.4	Corporation Encumbrances on Accounts Receivable Contracts Supplier/Customer Relationships Bank Accounts; Directors and Officers Environmental Matters Certain Changes Product Warranties Consents, Approvals
	EXHIBITS
Exhibit	Description
7.5 8.6(a) 8.6(b)	Legal Opinion of Hodgson, Russ, Andrews, Woods & Goodyear Legal Opinion of Latham & Watkins Form of Opinions to be delivered by
8.10	local counsel Restrictive Covenants Agreements

STOCK PURCHASE AGREEMENT

THIS STOCK PURCHASE AGREEMENT, dated as of May 6, 1994 is by and among HPI ACQUISITION CORP., a Delaware corporation with its principal place of business at 630 Dundee Road, Suite 400, Northbrook, Illinois 60065 ("Buyer"); HFP PARTNERS, L.P. ("HFP") and HMTC PARTNERS, L.P. ("HMTC"), each a Delaware limited partnership with its principal place of business at 355 South Grand Avenue, 42nd Floor, Los Angeles, California 90071, and the individuals and entities listed on Schedule A attached to this Agreement, residing at the addresses set forth opposite each individual's or entity's name on Schedule A (HFP, HMTC and such individuals and entities being individually a "Seller" and collectively "Sellers"); and HALE PRODUCTS, INC., a Delaware corporation with its principal place of business at 700 Spring Mill Avenue, Conshohocken, Pennsylvania 19428 (the "Corporation").

 $\,$ WHEREAS, Sellers own all of the issued and outstanding common shares of the Corporation;

WHEREAS, Buyer desires to purchase from Sellers, and Sellers desire to sell to Buyer, such shares upon the terms and conditions contained in this Agreement; and

WHEREAS, IDEX Corporation has guaranteed the obligations of its direct, wholly-owned subsidiary, Buyer, pursuant to the Guaranty Agreement executed and delivered by IDEX Corporation to Sellers simultaneously herewith.

NOW, THEREFORE, in consideration of the premises and covenants contained in this Agreement, and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, Sellers, Buyer and the Corporation agree as follows:

ARTICLE 1 DEFINITIONS

- ${\tt 1.1}$ ${\tt Defined\ Terms.}$ As used in this Agreement, the terms below shall have the following meanings:
- (a) "Affiliate" shall mean, as to any Person, any other Person which directly or indirectly controls, or is under common control with, or is controlled by, such Person. As used in this definition "control" (including, with its correlative meanings, "controlled by" and "under common control with") shall mean possession, directly or indirectly, of power to direct or cause the direction of management or policies (whether through ownership of securities or partnership or other ownership interest, by contract or otherwise).

- 2 -

- (b) "Agreement" shall mean this Stock Purchase Agreement together with the Schedules and Exhibits attached hereto and the certificates and instruments to be executed and delivered in connection herewith.
- (c) "Business" shall mean the manufacture, distribution, sales and service of pumps (primarily, but not exclusively, for use in fire fighting equipment), rescue equipment and tools, and other related products, and all other activities conducted by the Corporation or the Subsidiaries on the date of this Agreement.
 - (d) "Buyer's Accountants" shall mean the firm of Deloitte

& Touche.

- (e) "Change of Control" shall mean a change in beneficial ownership of 50% or more of the outstanding voting securities (or other equity interests having the power to vote for or control the decision of the management of) any Person, or as that term is otherwise defined in any agreement in which it appears.
- (f) "Closing Date" shall mean the later of (i) May 23, 1994 or (ii) the third business day after all required approvals under the HSR Act have been received, or any other date as Sellers and Buyer shall mutually agree.
- (g) "Closing Date Cash" shall mean (i) the cash of the Corporation (calculated in a manner consistent with the preparation of the Financial Statements) on the Closing Date, before giving effect to payment by the Corporation on the Closing Date of the Management Bonuses and amounts due under the Godiva Commitment, and before giving effect to payment by the Corporation of any Section 10.8 expenses, regardless of when paid, minus (ii) the balance of any sum outstanding under any overdraft facility of the Corporation or any Subsidiary.
- (h) "Closing Date Funded Debt" shall mean the sum of (i) the aggregate outstanding principal owing by the Corporation as of the Closing Date under its \$17,000,000 Series A 9.28% Senior Notes due March 26, 2000 and its \$19,000,000 Series B 9.86% Senior Notes due March 26, 2003, issued pursuant to a Note Agreement dated March 26, 1993 and (ii) the aggregate principal amount of all other indebtedness of the Corporation and the Subsidiaries for money borrowed from any financial institution (not to include any sums outstanding under any letters of credit or overdraft facility as of the Closing Date).

- 3 -

- (i) "Code" shall mean the Internal Revenue Code of 1986, as amended.
- (i)
- $\mbox{\ensuremath{\mbox{(j)}}}$ "Corporation" shall mean Hale Products, Inc., a Delaware corporation.
- $\hbox{(k)} \qquad \hbox{"Current Real Property" shall mean all Real Property currently owned or leased by the Corporation or any Subsidiary.}$
- (1) "Encumbrance" shall mean any claim, lien, pledge, option, charge, easement, security interest, right-of-way, encroachment, reservation, restriction, encumbrance, or other right of any Person, or any other restriction or limitation of any nature whatsoever, affecting title to the Shares, the Current Real Property, the Tangible Personal Property, the Intangible Property or any other assets of the Corporation or any Subsidiary.
- (m) "Environmental Claims" shall mean any notice of violation, notice of potential or actual responsibility or liability, claim, suit, action, demand, directive or order by any Person for any damage (including, but not limited to, personal injury, tangible or intangible property damage, contribution, indemnity, indirect or consequential damages, damage to the environment, environmental removal, response or remediation costs, nuisance, pollution, contamination or other adverse effects on the environment or for fines, penalties or restrictions on existing environmental permits or licenses) resulting from or relating to (i) the presence of, the Release or threatened Release into the environment of, or exposure to, any Hazardous Substance, (ii) the generation, manufacture, processing, distribution, use, handling, transportation, storage, treatment or disposal of any Hazardous Substance, (iii) the violation, or alleged violation, of any Environmental Laws or (iv) the non-compliance or alleged non-compliance with any Environmental Laws.
- (n) "Environmental Laws" shall mean any applicable statutes, ordinances, directives or other laws, any rules or regulations, orders, and any licenses, permits, orders, judgments, notices or other requirements issued pursuant thereto, enacted, promulgated or issued by any Governmental Authority, relating to pollution or protection of public health or the environment (including, but not limited to, any air, surface water, groundwater, land surface or sub-surface strata, whether outside, inside or under any structure), or to the identification, reporting, generation, manufacture, processing, distribution, use, handling, treatment, storage, disposal,

- 4 -

labelling, deposit, transporting, presence, Release or threatened Release of, any Hazardous Substances, pollutants, contaminants, wastes or any other substances or materials. Without limiting the generality of the foregoing, Environmental Laws shall include (i) in the United States, the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended, the Toxic Substances Control Act, as amended, the Hazardous Materials Transportation Act, as amended, the Resource Conservation and Recovery Act, as amended, the Clean Water Act, as amended, the Safe Drinking Water Act, as amended, the Clean Air Act, as amended, and the Occupational Safety and Health Act, as amended, (ii) in England and Wales, the Environmental Protection Act 1990, the Water Resources Act 1991, the Health and Safety at Work Act 1974, the Control of Substances Hazardous to Health Regulations 1988, the Control of Pollution Act 1974, the Clean Air Act 1956, the Clean Air Act 1968 and the Planning (Hazardous Substances) Act 1990, and (iii) all analogous laws enacted, promulgated or lawfully issued by any Governmental Authority.

- (o) "ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended.
- (p) "GAAP" shall mean, with respect to all United States accounting matters and issues, generally accepted accounting principles.
- (q) "Godiva Commitment" shall mean amounts owed pursuant to a verbal commitment by the Corporation in lieu of employees receiving stock which amounts shall not exceed \$180,000.
- (r) "Governmental Authority" shall mean any federal, state, local or foreign government (including, without limitation, the European Union), or any political subdivision of any of the foregoing, or any court, agency or other entity, body, organization or group, exercising any executive, legislative, judicial, quasi-judicial, regulatory or administrative function of government.
- (s) "Governmental Requirement" shall mean, with respect to the subject matter and context in which such term appears, all applicable laws, statutes, ordinances, directives, orders, rules and regulations of any Governmental Authority including, without limitation, ERISA, the Federal Occupational Safety and Health Act, the National Labor Relations Act, the Civil Rights Act, and the Age Discrimination in Employment Act.
- (t) "Hazardous Substances" shall mean any pollutants, contaminants, substances, chemicals, carcinogens, wastes and any

- 5 -

ignitable, corrosive, reactive, toxic or other hazardous substances or materials, whether solids, liquids or gases (including, but not limited to, petroleum and its derivatives, PCBs, asbestos, radioactive materials, waste waters, sludge, slag and any other substance, material or waste), as defined in or regulated by any Environmental Laws or as determined by any Governmental Authority.

- (u) "HSR Act" shall mean the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended from time to time.
- $\mbox{\ensuremath{(v)}}$ "IDEX" shall mean IDEX Corporation, a Delaware corporation.
- (w) "Intangible Property" shall mean (a) all United States federal and state and all foreign patents, trademarks, service marks and (with respect to the United States only) trade names and copyrights and all other intellectual property rights owned or used, pursuant to license agreements or otherwise, by the Corporation and each Subsidiary, and (b) all other unregistered intellectual property rights owned or used by the Corporation and each Subsidiary, including, without limitation, unregistered designs, design rights, unregistered trademarks and service marks, trade names, copyrights, moral rights, topography rights, rights in the nature of unfair competition rights and rights to sue for passing off (and equivalent rights), trade secrets and other proprietary rights.
- (x) "Management Bonuses" shall mean the aggregate amount of bonuses, payable to executives of the Corporation for the calendar year 1993 and through the Closing Date which amount shall not exceed \$452,580.
- (y) "Management Agent" shall mean Peter J. Andrews, Dale M. Clements and John J. O'Grady, Jr., each of whom, acting singly, shall be authorized to act for and on behalf of the Management Sellers pursuant to this Agreement.
- (z) "Management Sellers" shall mean all individual Sellers other than Walter B. Rose.
- (aa) "Permitted Exceptions" shall have the meaning given to such term on Schedule 4.9.
- (ab) "Person" shall mean any corporation, Governmental Authority, individual, partnership, trust or other entity.
- (ac) "Predecessor" shall mean any Person to which the Corporation or any Subsidiary is or is deemed to be a successor

in interest, whether directly or indirectly, by merger or otherwise.

- (ad) "Prohibited Exceptions" shall have the meaning given to such term on Schedule 4.9.
- (ae) "Purchase Price" shall mean \$90,200,000, less the Closing Date Funded Debt, less the amount by which the Closing Date Cash is less then \$2,500,000 less the amount of the Management Bonuses and the amounts due under the Godiva Commitment.
- (af) "Real Property" shall mean (i) all real property currently owned or leased by the Corporation or any Subsidiary or in which the Corporation or any Subsidiary currently has any interest, (ii) the real property formerly owned by American Godiva, Inc. and located at Main Street Road, Battle Creek, Michigan and (iii) the real property formerly leased by Macomson Company, a Predecessor of Hale Fire Pump Company, and located at 1009 Buffalo Street, Shelby, North Carolina, together with (x) all buildings and improvements located thereon and (y) all rights, privileges, interests, easements, hereditaments and appurtenances thereunto in any way incident, appertaining or belonging.
- (ag) "Release" shall mean any spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, dumping or disposing into the environment.
- (ah) "Representative" shall mean any officer, director, principal, attorney, accountant, agent, employee or other representative of any Person.
- (ai) "Seller's Accountants" shall mean the firm of Ernst & Young acting as accountants for the Corporation and the Subsidiaries prior to Closing.
- (aj) "Seller Agent" shall mean McBain, Rose Partners, a California general partnership.
- (ak) "Shares" shall mean all of the issued and outstanding shares of common stock of the Corporation, which as of the date of this Agreement consists of 479,171 shares.
- (al) "Subsidiary" shall mean any corporation or other entity of which a majority of the voting power of the voting equity securities or equity interest is owned, directly or indirectly, by the Corporation.

- 7 -

- (am) "TCW" shall mean TCW Special Placements Fund II, a California limited partnership.
- (an) "Transaction Costs" shall mean all of the costs and expenses incurred by or on behalf of Sellers in connection with this Agreement and the consummation of the transactions contemplated hereby, to the extent not reimbursed by the Corporation under Section 10.8, including transaction fees payable to McBain Rose Partners, which costs and expenses shall not exceed \$927,000 in the aggregate.
- (ao) "U.K. Subsidiaries" shall mean those Subsidiaries identified as "U.K. Subsidiaries" on Schedule 4.1.
- (ap) "U.S. Current Real Property" shall mean all Current Real Property other than U.K. Current Real Property.
- (aq) "U.S. Subsidiaries" shall mean those Subsidiaries identified as "U.S. Subsidiaries" on Schedule 4.1.
- (ar) "Warrants" shall mean the issued and outstanding warrants for the purchase of 100,000 shares of common stock of the Corporation held by TCW pursuant to (i) a Warrant Certificate issued by the Corporation to TCW on April 6, 1987 in the amount of 75,000 warrants and (ii) a Warrant Certificate issued by the Corporation to TCW on April 6, 1987 in the amount of 25,000 warrants.
- $1.2\,$ Other Defined Terms. The following terms shall have the meanings defined for such terms in the Sections set forth below:

Term 	Section
Accounts Receivable Closing COBRA Contracts Employee Contracts Employee Plan ERISA ERISA Affiliate Financial Statements Godiva Financial Statements Insurance Intangible Property Interim Financial Statements Inventory PBGC	4.20 3.1 4.15A(h) 4.21 4.14(b) 4.15A(a) 4.15A(e) 4.8 4.8 4.11 4.8 4.11 4.8 4.19 4.15A(k)

Pension Plans	4.15A(a)
Permits	4.12
Pre-Acquisition Financial Review	2.3(a)
Related Person	4.16
Tangible Personal Property	4.10
U.K. Current Real Property	4.9
Welfare Plans	4.15A(a)

For definitions applicable to United Kingdom employee benefit plans and provisions relating thereto, see Section 4.15B. For definitions applicable to United Kingdom tax matters, see Section 4.17B.

- 1.3 Usage of Terms. Except where the context otherwise requires, words importing the singular number shall include the plural number and vice versa.
- 1.4 References to Articles, Sections, Exhibits and Schedules. All references in this Agreement to Articles, Sections (and other subdivisions), Exhibits and Schedules refer to the corresponding Articles, Sections (and other subdivisions), Exhibits and Schedules of or attached to this Agreement, unless the context expressly, or by necessary implication, otherwise requires.
- 1.5 Appointment of Management Agent and Seller Agent.

 (a) Each of the Management Sellers hereby irrevocably appoints the Management Agent as the agent of such Management Seller to act for and on behalf of such Management Seller in any capacity in connection with this Agreement and all other agreements, documents and instruments executed and delivered by such Management Seller in connection with this Agreement. Each such Management Seller agrees to deliver this Agreement and all certificates representing any Shares owned by such Management Seller, together with blank stock powers with respect to such Shares duly executed and guaranteed, to the Management Agent and hereby authorizes the Management Agent to (i) amend or modify this Agreement on behalf of such Management Seller, (ii) attend the Closing on behalf of such Management Seller, (iii) deliver this Agreement and said stock certificates and blank stock powers to Buyer as required hereby against receipt of payment as provided in Section 2.4, (iv) execute, deliver and endorse any other agreements, documents or instruments required to be executed, delivered or endorsed on behalf of such Management Seller to accomplish the sale of the Shares pursuant to this Agreement, including, without limitation, any certificates with respect to representations, warranties, covenants and conditions required pursuant to Article 8 and (v) take all other actions for and on behalf of such Management Seller, whether before or after

- 9 -

the Closing, which may be necessary or desirable in connection with the consummation of the transactions contemplated by this Agreement or in connection with any matters arising after the Closing relating to this Agreement (including, without limitation, matters pursuant to Article 9). Except as otherwise specifically provided in this Agreement, each Management Seller agrees to act only by and through the Management Agent for all purposes in connection with this Agreement.

- (b) Each Seller hereby irrevocably appoints the Seller Agent as the agent of such Seller to act for and on behalf of such Seller for purposes of Sections 2.2 and 2.4. In furtherance thereof, each Seller hereby authorizes the Seller Agent to receive payment of the Purchase Price for and on behalf of such Seller, to pay from such Seller's pro rata share of the Purchase Price his or its pro rata share of the Transaction Costs, and to distribute to such Seller, after calculation of all such Transaction Costs, his or its remaining net pro rata share of the Purchase Price. Each Seller's pro rata share of the Purchase Price and the Transaction Costs shall be based upon (i) the number of Shares owned by such Seller compared to (ii) the number of Shares owned by all Sellers and the number of Warrants (or common shares which would be acquired upon exercise thereof) owned by TCW.
- Each Management Seller hereby agrees, severally and (c) not jointly, to indemnify the Management Agent and to hold him harmless against any loss, liability or expense incurred without grossly negligent conduct or bad faith on the part of the Management Agent and arising out of or in connection with his duties as Management Agent, including the costs and expenses incurred by such Management Agent in defending against any claim of liability in connection herewith and administering to the responsibilities of acting as Management Agent. Each Seller hereby agrees, severally and not jointly, to indemnify the Seller Agent and to hold it harmless against any loss, liability or expense incurred without grossly negligent conduct or bad faith on the part of the Seller Agent and arising out of or in connection with its duties as Seller Agent, including the costs and expenses incurred by such Seller Agent in defending against any claim of liability in connection herewith and administering to the responsibilities of acting as Seller Agent. Each Seller expressly acknowledges and accepts payment of the Purchase Price to the Seller Agent in accordance with Section 2.4 and, upon payment of such Purchase Price to Seller Agent by Buyer, releases and forever discharges Buyer from any claims which such Seller may have with respect to payment of such Seller's share of the Purchase Price.

1.6 Conversion Rates. Solely for purpose of applying any monetary tests under representations and warranties of the Corporation contained in this Agreement (but not for purposes of foreign currency translation under the Corporation's books, records or financial statements), the established foreign currency conversion rates shall be as follows:

U.S. \$1.00 = 0.67 British Pounds Sterling

U.S. \$1.00 = 1.70 German Marks

U.S. \$1.00 = 1.38 Canadian Dollars

ARTICLE 2 PURCHASE AND SALE OF SHARES

- 2.1 Transfer of Shares. Subject to the terms and conditions contained in this Agreement, on the Closing Date Sellers shall sell, convey, transfer, assign and deliver to Buyer, and Buyer shall acquire from Sellers, the Shares, free and clear of all Encumbrances.
- 2.2 Purchase Price. At the Closing, Buyer shall pay to the Seller Agent, as provided in Section 2.4, for the sale, transfer, assignment, conveyance and delivery of the Shares and Warrants an aggregate amount equal to the Purchase Price. None of the Purchase Price shall be allocated by any party for income tax reporting purposes to the Restrictive Covenants Agreements attached as Exhibit 8.10 to this Agreement.
- 2.3 Pre-Acquisition Financial Review. (a) Beginning on the date of this Agreement, Buyer and Buyer's Accountants shall have access to the Corporation and each Subsidiary, and to Sellers' Accountants and their work papers, for the purpose of conducting a pre-acquisition financial review (although not constituting a full financial audit) of the Corporation and each Subsidiary (the "Pre-Acquisition Financial Review").
- (b) In connection with the Pre-Acquisition Financial Review, Buyer shall have the right to require the Corporation and Sellers' Accountants to conduct, prior to Closing, a physical taking of inventory of the Corporation and its Subsidiaries, consistent with past practices, in accordance with GAAP, to be observed by Buyer and Buyer's Accountants, at any one or more or all of the facilities located at 700 Spring Mill Avenue and Washington Street, Conshohocken, Pennsylvania; 711 North Post Road, Shelby, North Carolina; 750 American Boulevard, St. Joseph, Tennessee; Charles Street, Warwick, Warwickshire, England; Old Whittington Road, Gobowen, Oswestry, Shropshire, England; and Industriegebiet-Nord, Benzstrasse 4, D-64807 Dieburg, Germany.

Any valuation of inventory pursuant to this Section 2.3 shall be computed by Sellers' Accountants in accordance with GAAP.

- 2.4 Payment of Purchase Price. On the Closing Date, Buyer shall pay to the Seller Agent an amount equal to the Purchase Price in cash by wire transfer to an account designated by the Seller Agent, such amount to be distributed by the Seller Agent to each of the Sellers and TCW pro rata in proportion to their respective interests (after giving effect to the exercise by TCW of the Warrants), but subject to reduction in the amount of the Transactions Costs, as provided in Section 1.5(b).
- 2.5 Taxes. Sellers shall be responsible for the payment of any transfer, sales, use or other similar taxes imposed by reason of the transfer of the Shares pursuant to this Agreement and any deficiency, interest or penalty with respect to such taxes.

ARTICLE 3 CLOSING

- 3.1 Closing. The closing of the transaction contemplated by this Agreement shall be held at 10:00 a.m. local time on the Closing Date at the offices of Latham & Watkins, 233 South Wacker Drive, Sears Tower, Suite 5800, Chicago, Illinois, or any other place as Buyer and Sellers mutually agree ("Closing"). The Closing shall be effective as of the commencement of business on the Closing Date.
- 3.2 Stock Certificates and Instruments of Assignment. To effect the transfer referred to in Section 2.1 on the Closing Date, the Management Agent shall deliver to Buyer with respect to each Management Seller, and each other Seller shall deliver to Buyer, each certificate representing any of the Shares held by each such Seller and all stock powers or other instruments of assignment reasonably requested by Buyer. Such instruments of assignment shall be in form and substance, and shall be executed and delivered in a manner, satisfactory to Buyer.
- 3.3 Warrants. The Warrants shall have been fully exercised or otherwise cancelled or extinguished or transferred to Buyer on or before the Closing Date and, to the extent so exercised, all Shares issued as a result thereof shall be transferred to Buyer on the Closing Date in accordance with Section 3.2.
- 3.4 Purchase Price; Certificates. On the Closing Date, Buyer shall deliver and tender the Purchase Price. Buyer

and Sellers shall deliver the certificates, agreements and other items described in Articles 7 and 8 of this Agreement.

ARTICLE 4 REPRESENTATIONS AND WARRANTIES OF SELLERS AND THE CORPORATION

- A. Representations and Warranties of the Corporation The Corporation hereby represents and warrants to Buyer as follows:
- Organization, Good Standing and Authority of the Corporation and each Subsidiary to Conduct Business. The Corporation is a corporation duly organized, validly existing and in good standing under the laws of the state of Delaware. Each Subsidiary is duly organized, validly existing and in good standing under the laws of the jurisdiction in which it is organized. Schedule 4.1 sets forth the identity of each Subsidiary, the jurisdiction of organization of each Subsidiary, each jurisdiction other than the jurisdiction of organization where the Corporation and each such Subsidiary is qualified to do business and each tradename or assumed name used by the Corporation and each Subsidiary in the conduct of the Business. The Corporation is, and each Subsidiary is, duly qualified to do business in, and is in good standing under the laws of, each jurisdiction in which such qualification is necessary under the applicable law as a result of the conduct $\ensuremath{\mathsf{I}}$ of its respective business or the ownership of its respective properties. The Corporation has, and each Subsidiary has, full power and authority to conduct its business as it is presently being conducted and to own and lease its properties and assets. None of the U.K. Subsidiaries is insolvent or unable to pay its debts for the purposes of Section 123 of the Insolvency Act 1986 and no administrative receiver or receiver or receiver and manager has been appointed by any Person of its Business or assets or any part thereof and no power to make any such appointment is currently exercisable. Except as set forth on Schedule 4.1, neither the Corporation nor any Subsidiary has any stock, partnership or other equity interest in any corporation, partnership, joint venture, firm or organization, and the Corporation and each Subsidiary conducts its business directly and not through any association, joint venture, partnership or other business entity.
- 4.2 Power and Authority; Authorization; Binding Effect. The Corporation has all necessary power and authority and has taken all action necessary to execute and deliver this Agreement, to consummate the transactions contemplated by this Agreement and to perform its obligations under this Agreement.

Copies of all resolutions of the board of directors of the Corporation with respect to the transactions contemplated by this Agreement, certified by the Secretary or an Assistant Secretary of the Corporation, in form satisfactory to counsel for Buyer, have been delivered to Buyer. This Agreement has been duly executed and delivered by the Corporation and constitutes a legal, valid and binding obligation of the Corporation enforceable against the Corporation in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium or other similar laws relating to or affecting the rights of creditors generally, or by general principles of equity. Each of Mr. Peter J. Andrews, chief executive officer of the Corporation, and Mr. John J. O'Grady, Jr., Secretary of the Corporation, is, and at all times through the Closing Date will be, duly authorized to execute and deliver, for and on behalf of the Corporation, this Agreement and all other agreements, instruments, certificates and other documents incident or related hereto.

- No Conflict or Violation. The execution and delivery of this Agreement, the consummation of the transactions contemplated by this Agreement and the fulfillment of the terms of this Agreement (a) do not and will not result in a violation of or conflict with any provision of the certificate or articles of incorporation, bylaws or other organization charters, certificates or documents of the Corporation or any Subsidiary, (b) except as set forth on Schedule 4.3, do not and will not constitute a breach of, or constitute an event, occurrence, condition or act which is or, with the giving of notice or the lapse of time would become, a default under, or result in the acceleration of, any obligations under, any term or provision of, any contract, agreement, indebtedness, encumbrance, commitment, license, franchise, permit, authorization or concession to which the Corporation or any Subsidiary is a party, (c) do not and will not result in a violation by the Corporation or any Subsidiary of any statute, rule, regulation, ordinance, code, order, judgment, writ, injunction, decree or award, or (d) do not and will not result in an imposition of any Encumbrance on the Shares or any assets of the Corporation or any Subsidiary.
- 4.4 Consents and Approvals. Except for (a) any approvals required under the HSR Act and (b) as otherwise set forth on Schedule 4.4, no consent, approval or authorization of, or declaration, filing or registration with, any Governmental Authority or other Person is required to be made or obtained by the Corporation, any Subsidiary or any Seller in connection with the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby.

- 14 -

- 4.5 No Proceedings. Except as set forth on Schedules 4.5 and 4.13, there is no action, order, writ, injunction, judgment or decree outstanding, or claim, suit, litigation, proceeding, arbitral action or investigation pending, or to the knowledge of the Corporation or any Subsidiary threatened or anticipated, relating to or affecting in any adverse manner the transactions contemplated by this Agreement.
- Capitalization. Schedule 4.6 sets forth the authorized, issued and outstanding shares of capital stock of the Corporation and each Subsidiary, the legal and beneficial ownership thereof and any Encumbrances thereon. All of the Shares are duly authorized, validly issued, fully paid and nonassessable, and were issued in compliance with all applicable laws. All voting rights with respect to the Corporation are vested in the Shares. Except as set forth in Schedule 4.6, (a) there are no outstanding shares of capital stock of the Corporation or any Subsidiary, or outstanding securities convertible into or exchangeable or exercisable for shares of capital stock of the Corporation or any Subsidiary, (b) there are no bonds, debentures, notes, or other indebtedness having the right to vote on any matters on which the Corporation's or any Subsidiary's shareholders may vote, (c) there are no outstanding options, warrants, rights, contracts, commitments, understandings or arrangements by which the Corporation or any Subsidiary is bound to issue, repurchase or otherwise acquire or retire any capital stock of the Corporation or any Subsidiary, (d) there are no voting agreements, voting trusts, buy-sell agreements, options or rights or obligations relating to the shareholders or the capital stock of the Corporation or any Subsidiary, and (e) except for certain provisions of this Agreement, there are no agreements between any Seller and the Corporation or any Subsidiary which will survive the Closing. Upon consummation of the transactions contemplated by this Agreement, Buyer will acquire the Shares, free of any Encumbrance.
- 4.7 Corporate Records. Except as set forth on Schedule 4.7, the minute books of the Corporation and each Subsidiary are complete and accurate and contain a complete and accurate record of all meetings and actions of shareholders and directors and of any executive committee or other committee of the shareholders or board of directors. The stock record book of the Corporation and each Subsidiary is complete and accurate and contains a complete and accurate record of all share transactions for the Corporation and each Subsidiary from the date of its incorporation. True and complete copies of the articles of association and other similar governing and organizational documents of the Corporation, each Subsidiary and each other entity listed on Schedule 4.1 have been delivered to Buyer, and

true and complete copies of the minute book and stock record book of the Corporation and each Subsidiary have been made available for review by Buyer.

Financial Statements. The Corporation has delivered to Buyer (a) consolidated audited financial statements of the Corporation and the Subsidiaries for each of the years in the four-year period ended December 31, 1993 (consisting of a balance sheet, statement of income, profit and loss and a statement of cash flows), certified by the Sellers' Accountants (the "Financial Statements"), (b) consolidated unaudited interim financial statements of the Corporation and the Subsidiaries (consisting of a balance sheet and a statement of income, profit and loss) for the three-month period ended March 31, 1994 (the "Interim Financial Statements"), (c) consolidating financial statements of the Corporation and each Subsidiary for each of the years in the four-year period ended December 31, 1993 and consolidating interim financial statements for the period ended March 31, 1994 (consisting of balance sheets, statements of income, profit and loss and, for 1994 only, statements of cash flows) and (d) audited financial statements of Godiva Products Limited for the period ended December 31, 1993 (consisting of balance sheets, statement of profit and loss and statement of cash flows), certified by Sellers' Accountants (the "Godiva Financial Statements"). Copies of the Financial Statements, the Interim Financial Statements and the Godiva Financial Statements are attached to this Agreement as Schedule 4.8. Except as set forth on Schedule 4.8, the Financial Statements and the Interim Financial Statements fairly present the financial condition and the results of operations of the Corporation and its Subsidiaries as of their respective dates and for the periods then ended and the Financial Statements have been prepared in accordance with GAAP applied on a consistent basis. The books and records of the Corporation and its Subsidiaries fairly reflect the assets, liabilities and operations of the Corporation and its Subsidiaries in accordance with GAAP and the Financial Statements and the Interim Financial Statements are in conformity therewith. The Financial Statements and the Interim Financial Statements (with respect to the U.S. Subsidiaries only) provide for all material fixed and non-contingent liabilities of the Corporation and its Subsidiaries, and disclose or provide for all contingent liabilities of the Corporation and its Subsidiaries of a type required to be disclosed or provided for in financial statements in accordance with GAAP. Except as set forth on Schedule 4.8, all reserves reflected in the Financial Statements and the Interim Financial Statements were and are adequate, appropriate and reasonable in accordance with GAAP. To the knowledge of the Corporation and any Subsidiary, except as disclosed on Schedule 4.8, there are no liabilities or obligations of any

nature, whether absolute, accrued, contingent, matured, unmatured or otherwise, and whether or not required to be disclosed or provided for in financial statements in accordance with GAAP, of the Corporation or any Subsidiary (including, without limitation, liabilities relating to any Employee Plans) except (a) liabilities and obligations reflected or reserved for in the Financial Statements and the Interim Financial Statements, (b) as otherwise specifically disclosed in this Agreement, (c) liabilities and obligations incurred between March 31, 1994 and the Closing Date in the ordinary course of the business of the Corporation and its Subsidiaries, consistent with past practice, and as permitted by this Agreement and (d) contractual liabilities and obligations not required to be disclosed pursuant to Section 4.21. The adjusted pro forma net income of the Corporation and its Subsidiaries for the fiscal year ended December 31, 1993 is at least \$2,550,000, after adjustment for the write-off of goodwill and non-recurring charges for debt restructuring and corporate finance fees. The available cash flow of the Corporation (defined as gross margin less selling, general and administrative operating costs, exclusive of depreciation, amortization and McBain, Rose Partners management fees) for the calendar year ended December 31, 1993 was at least \$11,200,000.

Real Property. Schedule 4.9 contains a true, 4.9 complete and correct list of the Real Property and identifies all Real Property which constitutes Current Real Property. Except as set forth on Schedule 4.9, (a) the Corporation or the applicable Subsidiary, as the case may be, has good and marketable title to the Current Real Property owned by the Corporation or such Subsidiary, (b) the Corporation or the applicable Subsidiary, as the case may be, enjoys peaceful and undisturbed possession of the Current Real Property (other than the U.K. Current Real Property) leased by the Corporation or such Subsidiary, (c) none of the Current Real Property is subject to any commitment for sale or use by any Person other than the Corporation or its Subsidiaries, (d) none of the Current Real Property is subject to any Encumbrance (other than the Permitted Exceptions) which in any material respect interferes with or impairs the transferability or present and continued use thereof in the usual and normal conduct of the Business, (e) the Current Real Property (other than the U.K. Current Real Property), and each user thereof, is in compliance with all Governmental Requirements and (f) no default or breach exists with respect to, and neither the Corporation nor any Subsidiary has received any notice of any default or breach under, any Encumbrance affecting any of the Current Real Property. There are no condemnation or eminent domain proceedings pending, or to the knowledge of the Corporation or any Subsidiary, contemplated or threatened,

against the Current Real Property (other than the U.K. Current Real Property) or any part thereof, and neither the Corporation nor any Subsidiary knows of any desire of any Governmental Authority to take or use the Current Real Property (other than the U.K. Current Real Property) or any part thereof. There are no existing, or to the knowledge of the Corporation or any Subsidiary, contemplated or threatened, general or special assessments affecting the Current Real Property (other than the U.K. Current Real Property) or any portion thereof. Neither the Corporation nor any Subsidiary has received notice of, nor does the Corporation or any Subsidiary have any knowledge of, any pending or threatened action, suit, proceeding or investigation before any Governmental Authority which relates to the ownership, maintenance, use or operation of the Current Real Property (other than periodic general reassessments, which reassessments, if any, are set forth on Schedule 4.9). Except as set forth on Schedule 4.9, the buildings and improvements on the Current Real Property (including, without limitation, the heating, air conditioning, mechanical, electrical and other systems used in connection therewith) are structurally sound and otherwise in a reasonable state of repair, have been well maintained and are free from infestation by termites, other wood destroying insects, vermin and other pests. There are no repairs or replacements exceeding \$500,000 in the aggregate for all Current Real Property or \$100,000 for any single repair or replacement which are currently contemplated by the Corporation or any Subsidiary or which, to the knowledge of the Corporation or any Subsidiary, should be made in order to maintain said buildings and improvements in a reasonable state of repair. The only Current Real Property located in the United Kingdom owned by the Corporation or any Subsidiary are the two freehold sites at Warwick and Gobowen respectively (the "U.K. Current Real Property"). There are no current or former leasehold or licensed properties or freehold properties located in the United Kingdom leaving any residual liability with the Corporation or any Subsidiary. Except as set forth on Schedule 4.9, the proprietor has exclusive and unfettered possession of the U.K. Current Real Property. Except as set forth on Schedule 4.9, the title to the U.K. Current Real Property is properly constituted by and can be deduced from the documents of title which are in the possession and under the control of the Corporation or any Subsidiary. There are no covenants, restrictions, stipulations, conditions, terms or outgoings affecting the U.K. Current Real Property which are of an unusual or onerous nature or which adversely affect the value of the U.K. Current Real Property and there are no outstanding disputes, notices or complaints which affect or might in the future affect the use of any of the U.K. Current Real Property. Neither the Corporation nor any Subsidiary has

received notice that any U.K. Current Real Property fails to comply with all Governmental Requirements.

Tangible Personal Property. Schedule 4.10(a) contains a description of each of the fixed asset depreciation lists of tangible personal property (other than Inventory) owned by the Corporation and each Subsidiary, true and complete copies of which have been previously provided to Buyer (and, if available, attached to Schedule 4.10(a)), and Schedule 4.10(b) lists each item of tangible personal property leased by the Corporation and each Subsidiary (other than individual leases of office equipment having an annual rental of less than \$25,000) (such owned and leased tangible personal property being collectively the "Tangible Personal Property"). The Tangible Personal Property constitutes substantially all of the tangible personal property used in the operation of the Business of the Corporation and each Subsidiary and constitutes substantially all tangible personal property necessary to conduct the Business of the Corporation and each Subsidiary as presently conducted. Except as set forth on Schedule 4.10(a), the Tangible Personal Property and all other personal property, whether tangible or intangible, owned by the Corporation and each Subsidiary is free and clear of all material Encumbrances. With respect to those Encumbrances identified on Schedule 4.10 as Encumbrances for which the Corporation or any Subsidiary is currently seeking releases (including, without limitation, Encumbrances in favor of Fidelity Bank, National Association and Midland Bank plc) all underlying indebtedness has been paid in full. With respect to Encumbrances identified on Schedule 4.10 in favor of Barclays Bank plc, except as set forth on Schedule 4.10, no amounts are currently outstanding under the credit facility secured by such Encumbrances, and no amounts will be outstanding under said credit facility at any time through the Closing Date. Except as set forth on Schedule 4.10, all of the Tangible Personal Property is located at the Current Real Property and there is no tangible personal property located at any of the Current Real Property which is not owned or leased by the Corporation or any Subsidiary. The Tangible Personal Property is in all material respects in good working order, ordinary wear and tear excepted. Neither the Corporation nor any Subsidiary currently intends to incur and, to the knowledge of the Corporation and any Subsidiary, neither the Corporation nor any Subsidiary should incur, costs and expenses exceeding \$300,000 in the aggregate for all Tangible Personal Property or \$50,000 for any single item of Tangible Personal Property to repair or replace any Tangible Personal Property or to maintain the Tangible Personal Property in good working order. Buyer acknowledges that the Corporation has delivered to it a copy of its 1994 capital budget, and that no planned expenditure set forth in such budget shall be deemed a

violation of this Section 4.10 or any other provision of this Agreement.

4.11 Intangible Property. To the knowledge of the Corporation or any Subsidiary, Schedule 4.11 lists (a) all United States federal and state and all foreign registrations and applications pertaining to any of the Intangible Rights, and (b) all applicable grant, registration, application or serial numbers, all other filing or recording information and all renewals and expiration dates pertaining thereto. Except as set forth on Schedule 4.11, (i) the Intangible Property is legally and beneficially owned exclusively by the Corporation or the applicable Subsidiary and is used exclusively by the Corporation and/or the applicable Subsidiaries and is not the subject of any pending or threatened proceeding for opposition, cancellation, reexamination, revocation or rectification and, to the knowledge of the Corporation or any Subsidiary, there are no facts or matters which might give rise to any such proceeding; (ii) the Intangible Property owned by the Corporation and each Subsidiary is free and clear of all Encumbrances; (iii) the use by the Corporation and each Subsidiary of the Intangible Property does not infringe upon or otherwise violate any right of any third Person in or to such Intangible Property; (iv) during the five-year period preceding the date of this Agreement, neither the Corporation nor any Subsidiary has received any notice of, and there is no pending or, to the knowledge of the Corporation or any Subsidiary, threatened action or proceeding by or before any Governmental Authority alleging, any infringement or other violation of any right of any third Person in or to the Intangible Property; (v) there is not now, and there has not been during the past five years, any infringement or other violation of any other intellectual property right of any third Person resulting from the conduct of the business of the Corporation or any Subsidiary, and neither the Corporation nor any Subsidiary is aware that any such infringement or violation exists or is or was alleged or will be alleged; (vi) neither the Corporation nor any Subsidiary knows of any activity by any third Person which does or might constitute an infringement or other violation of the Corporation's or any Subsidiary's rights in or to any Intangible Property which the Corporation or such Subsidiary has not defended or is not defending, except where such infringement would not have a material adverse effect on the Business; (vii) neither the Corporation nor any Subsidiary has entered into any license, consent, indemnification, forbearance to sue, settlement agreement or cross-licensing arrangement with any Person relating to the Intangible Property or any intellectual property right of any third Person; (viii) there are no agreements relating to or affecting any Intangible Property of the Corporation or any Subsidiary or the use or ownership thereof, including, without

limitation, license agreements, confidentiality and non-disclosure agreements, assignments or agreements to assign, development agreements, settlement agreements and other related agreements; (ix) all applicable continuation, maintenance and renewal fees owing with respect to any of the Intangible Property have been paid in full and (x) neither the Corporation nor any Subsidiary is aware of any information which would or might materially adversely affect any of the Intangible Property or render any of the Intangible Property invalid or unenforceable. To the knowledge of the Corporation or any Subsidiary, the Intangible Property constitutes all of the patents, trademarks, service marks, trade names, copyrights, design rights, trade secrets and other intellectual property rights or intangible property (whether registered or not) and wherever in the world enforceable, necessary to operate the Business of the Corporation and each Subsidiary. The consummation of the transactions contemplated hereby will not result in the loss or impairment of any of the Corporation's or any Subsidiary's rights in the Intangible Property. No shareholder, director, officer or employee, or former officer or employee, of the Corporation or any Subsidiary owns, directly or indirectly, in whole or in part, any rights in any of the Intangible Property. Each of the Corporation and each Subsidiary has taken all reasonable steps available to each of them to preserve the Intangible Property including, but not limited to, (A) opposing any application to register any trademark and/or service mark likely to be confused with any of the trademarks in Schedule 4.11 in any of the countries and in relation to the goods and/or services in respect of which the trademarks and/or service marks are registered; (B) taking all reasonable steps to preserve the confidentiality of all confidential information and trade secrets used in the Business, including ensuring that all such information and secrets are held in a secure location, are only disclosed to such employees and other persons to whom disclosure is necessary in the conduct of the business and who are aware of, and accept an obligation to maintain, the confidentiality thereof; and (C) obtaining from any individual other than an employee who is engaged in, or contributes to, the creation of any Intangible Property a written assignment of the rights of that individual to the Corporation or such Subsidiary. Each of the Corporation and each Subsidiary has the sole and exclusive right to use its corporate name and each tradename or assumed name under which it conducts the Business. Except as disclosed on Schedule 4.11, (x) no Person has asserted, or to the knowledge of the Corporation or any Subsidiary threatened to assert, any claim or made any demand to the right to such name or the right to use such name, and (y) no proceeding has been instituted, or is pending, or to the knowledge of the Corporation or any Subsidiary threatened, which challenges the exclusive right of the Corporation or any

Subsidiary with respect thereto. To the knowledge of the Corporation or any Subsidiary, no other Person is using such names in connection with any business or commercial activity similar to or competitive with the Business.

4.12 Compliance with Laws; Permits. Except as set forth in Schedule 4.12, the Corporation and each Subsidiary and the conduct of the Business of the Corporation and each Subsidiary has duly complied with and is in compliance with all Governmental Requirements. Neither the Corporation nor any Subsidiary has received any notice to the effect that, or otherwise been advised that, the Corporation or any Subsidiary is not in compliance with any Governmental Requirement, and neither the Corporation nor any Subsidiary reasonably anticipates that any presently existing circumstances are likely to result in violations of any Governmental Requirement. The permits, consents, licenses, franchises, authorizations and approvals issued to the Corporation and each Subsidiary by any Governmental Authority which are currently in effect (the "Permits") constitute all material permits, consents, licenses, franchises, authorizations and approvals of any Governmental Authority or other Person (a) which are used in the operation of the Business of the Corporation and each Subsidiary and (b) which are necessary to conduct the Business as presently conducted, other than those the failure of which to obtain would not have a material adverse effect on the Business, assets or financial condition of the Corporation or any Subsidiary. All of the Permits are valid and in full force and effect, no violations thereof have been issued or are anticipated and no proceeding is pending, or to the knowledge of the Corporation or any Subsidiary threatened, to revoke or limit any of them. Except as set forth on Schedule 4.12, the consummation of the transactions contemplated by this Agreement do not and will not violate or render any of the Permits invalid, require any amendment or reissuance of any of the Permits or require the consent of the Governmental Authority which has issued any of the Permits.

4.13 Litigation. To the best of the Corporation's knowledge, except as set forth in Schedule 4.13, there is no claim, legal action, suit, arbitration, Governmental Authority investigation or other legal or administrative proceeding, or any order, decree, or judgment pending, or to the knowledge of the Corporation or any Subsidiary threatened, against or relating to the Corporation or any Subsidiary, its officers, directors or employees, or its properties, assets or Business. Neither the Corporation nor any Subsidiary knows of any basis or grounds for any such claim, legal action, suit, arbitration, Governmental Authority investigation or other legal or administrative proceeding. None of the matters disclosed in Schedule 4.13 has

or will have a material adverse affect on the business or financial condition of the Corporation or any Subsidiary.

4.14 Labor Matters; Employee Contracts. (a) Schedule 4.14 identifies the name of each current employee of the Corporation and each Subsidiary whose total compensation (including bonuses) earned in 1993 exceeded The current compensation of each employee listed on Schedule 4.14 has been provided to Buyer. Except as set forth in Schedule 4.14, (a) neither the Corporation nor any Subsidiary has any obligations under or is a party to any written or oral labor agreement, collective agreement, collective bargaining agreement or other agreement with any labor organization or employee group and neither the Corporation nor any Subsidiary has a works council, (b) to the knowledge of the Corporation or any Subsidiary, neither the Corporation nor any Subsidiary is currently engaged in any unfair labor practice and there is no unfair labor practice charge or other employee-related or employment-related complaint against the Corporation or any Subsidiary pending or threatened before any Governmental Authority, (c) there is currently no labor strike, labor disturbance, slowdown, work stoppage or other material labor dispute or arbitration pending or threatened against the Corporation or any Subsidiary nor is any material grievance currently being asserted, (d) neither the Corporation nor any Subsidiary has experienced a labor strike, labor disturbance, slowdown, work stoppage or other material labor dispute at any time during the three years immediately preceding the date of this Agreement and (e) there is no labor union, trade union or staff association expressly or impliedly recognized by the Corporation or any Subsidiary and, to the knowledge of the Corporation or any Subsidiary, there is no organizational campaign being conducted or contemplated and there is no pending or threatened petition before any Governmental Authority or other dispute as to the representation of any employees of the Corporation or any Subsidiary. Except as set forth on Schedule 4.14, each of the Corporation and each Subsidiary has complied with, and is currently in compliance in all material respects with, all Governmental Requirements relating to any of its employees or consultants (including, without limitation, any Governmental Requirement of the Occupational Safety and Health Administration), and neither the Corporation nor any Subsidiary has received, within the past three years, any written notice of failure to comply with any such Governmental Requirement.

(b) Schedule 4.14 contains a list identifying each currently binding written or oral contract, agreement, arrangement, policy, program, plan or practice (exclusive of any such contract which is terminable within thirty (30) days without

liability to Sellers, the Corporation or any Subsidiary) directly or indirectly providing for or relating to any employment, consulting, remuneration, compensation or benefit, severance or other similar arrangement, insurance coverage (including any self-insured arrangements), medical-surgical-hospital or other health benefits, workers' compensation, disability benefits, supplemental employment benefits, vacation benefits and other forms of paid or unpaid leave, retirement benefits, deferred compensation, savings or bonus plans, profit-sharing, stock options, stock appreciation rights, or other forms of incentive compensation or post-retirement compensation or benefit, employment guarantee or security, or limitation on right to discipline or discharge, which (i) is not an Employee Plan, (ii) has been entered into or maintained, as the case may be, by Sellers, the Corporation or any Subsidiary, and (iii) covers any one or more current or former director, officer, employee or consultant of the Corporation or any Subsidiary (collectively, "Employee Contracts"). All Employee Contracts are valid and binding on all parties thereto, are in full force and effect, and no party to any such Employee Contract is in material default thereunder. Except as provided on Schedule 4.14 or reflected or reserved for on the Financial Statements or the Interim Financial Statements, neither the Corporation nor any Subsidiary has any liability for unpaid wages, salaries, bonuses, commissions, vacation pay severance pay, health insurance, life insurance, or other form of employee compensation. Except as set forth on Schedule 4.14, there is no amount in excess of \$500 owing to the Corporation or any Subsidiary from any current or former director, officer, employee or consultant or owing by the Corporation or any Subsidiary to any current or former director, officer, employee or consultant. Except as set forth on Schedule 4.14, neither the Corporation nor any Subsidiary is a party to any Employee Contract which provides that the terms and conditions that would otherwise govern the relationship of the parties thereto will be altered upon a Change of Control. True and complete copies or descriptions of the Employee Contracts have been delivered to Buyer. Each Employee Contract has been maintained in substantial compliance with the requirements prescribed by any and all statutes, orders, rules and regulations which are applicable to such Employee Contract.

(c) Each of the Corporation and each U.S. Subsidiary has completed a valid Form I-9 for each employee hired on or after November 7, 1986. To the knowledge of the Corporation and any Subsidiary, all of the Corporation's and each U.S. Subsidiary's employees are (i) United States citizens, or lawful permanent residents of the United States, (ii) aliens whose right to work in the United States is unrestricted, (iii) aliens who have valid, unexpired work authorization issued by the Attorney

General of the United States (Immigration and Naturalization Service) or (iv) aliens who have been continually employed by the Corporation since November 6, 1986. Neither the Corporation nor any Subsidiary is, or has been, the subject of an immigration compliance or employment visit from, nor has the Corporation or any Subsidiary been assessed any fine or penalty by, or been the subject of any order or directive of, the United States Department of Labor or the Attorney General of the United States (Immigration and Naturalization Service).

- (d) Neither the Corporation nor any Subsidiary has dismissed (within the meaning of Section 55(2) Employment Protection (Consolidation) Act 1978) any Person whose employment is subject to the laws of the United Kingdom from its employment, including but not limited to employees, any casual worker, consultant and representative for any reason directly or indirectly connected with the subject-matter of this Agreement and, for the avoidance of doubt, this warranty shall include but not be limited to claims arising out of the dismissal of any employee where notice of dismissal or resignation is given prior to the date of this Agreement but expires after that
- (e) Except as set forth in Schedule 4.14, (i) there is no pending, existing or, to the knowledge of the Corporation or any Subsidiary, threatened claim or litigation, right of action or formal investigation relating to all or any of the employees of the Corporation and each Subsidiary whose employment is subject to the laws of the United Kingdom; (ii) there is, to the knowledge of the Corporation or any Subsidiary, no matter, act or omission in respect of which any employee, casual worker, consultant or representative of the Corporation or any Subsidiary whose employment is subject to the laws of the United Kingdom could or may bring a claim or commence legal proceedings connected with or arising from his or her employment; and (iii) there are, with respect to any such employees, no additional terms or conditions of employment, employment policies or schemes, special working arrangements, severance or redundancy plans relating to any of the employees of the Corporation and each Subsidiary.
- (f) Each of the Corporation and each Subsidiary which has employees whose employment is subject to the laws of the United Kingdom, has paid to, and/or in respect of, each and all of their employees, all outstanding national insurance contributions, PAYE deductions, salary, wages, pension contributions, holiday pay, expenses, allowances and all other benefits or employments.

- 25 -

- 4.15A United States Employee Benefit Plans. The following representations and warranties set forth in this Section 4.15A shall apply to each employee benefit plan or arrangement maintained by the Corporation or any Subsidiary and to which the laws of the United States apply:
- (a) Attached hereto is (i) as Schedule 4.15A(a)(1), a list identifying each "employee pension benefit plan," as defined in Section 3(2) of the Employee Retirement Income Security Act of 1974, as amended, ("ERISA"), including any "multiemployer plan," as defined in Section 3(37) of ERISA, (the "Pension Plans") and (ii) as Schedule 4.15A(a)(2), a list identifying each "employee welfare benefit plan," as defined in Section 3(1) of ERISA, (the "Welfare Plans") that, in either case, are maintained, administered or contributed to by the Corporation or its Subsidiaries, or which cover any employee or former employee of the Corporation or its Subsidiaries. Collectively, the Pension Plans and the Welfare Plans shall hereafter be referred to as the "Employee Plans." Except as otherwise identified on Schedule 4.15A(a)(1) and (a)(2) and on Schedule 4.14, (i) no Employee Plan or Employee Contract (as defined in Section 4.14(b) of this Agreement) is maintained, administered or contributed to by any entity other than the Corporation or any Subsidiary, and (ii) no Employee Plan is maintained under any trust arrangement which covers any employee benefit arrangement which is not an Employee Plan.
- (b) The Corporation has delivered or caused to be delivered to Buyer true and complete copies of (i) the Employee Plans (and related trust agreements and other funding arrangements, if any, and adoption agreements, if any), (ii) any amendments to the Employee Plans, (iii) written interpretations of the Employee Plans held by the plan administrator (or which the plan administrator has knowledge) of such Employee Plan (iv) material employee communications by the plan administrator of any Employee Plan (including, but not limited to, summary plan descriptions and summaries of material modifications as defined under ERISA), (v) the three most recent annual reports (e.g., the complete Form 5500 series) prepared in connection with each Employee Plan (if any such report was required), including all attachments (including without limitation the actuarial valuation reports) and (vi) the three most recent actuarial valuation reports prepared in connection with each Employee Plan (if any such report was required).
- (c) Except as set forth on Schedule 4.15A(c), each Employee Plan has been maintained in compliance in all material respects with its terms and the requirements prescribed by any and all statutes, orders, rules and regulations, including but

not limited to, ERISA and the Code, which are applicable to such Employee Plan.

- (d) There are no pending or, to the knowledge of the Corporation or any Subsidiary, threatened claims, suits or other proceedings by any employees, former employees or plan participants or the beneficiaries, spouses or representatives of any of them, against any Employee Plan, the assets held thereunder, the trustee of any such assets, or the Corporation or its Subsidiaries relating to any of the Employee Plans, Employee Contract, any other employee benefit plans, contracts or arrangements, other than ordinary and usual claims for benefits by participants, beneficiaries or alternate payees. Furthermore, there are no pending or, to the knowledge of the Corporation or any Subsidiary, threatened suits, investigations or other proceedings by any Governmental Authority of or against any Employee Plan, the trustee of any assets held thereunder, or the Corporation or its Subsidiaries relating to any of the Employee Plans, Employee Contracts, any other employee benefit plans, contracts or arrangements.
- (e) Except as set forth on Schedule 4.15A(c), no liability has been incurred by the Corporation or any Subsidiary or by a trade or business, whether or not incorporated, which is deemed to be under common control or affiliated with the Corporation within the meaning of Section 4001 of ERISA or Sections 414(b), (c), (m) or (o) of the Code (an "ERISA Affiliate") for any tax, penalty or other liability with respect to any Employee Plan and, to the knowledge of the Corporation or any Subsidiary, such Employee Plans do not expect to incur any such liability prior to the date of Closing. The Corporation and its Subsidiaries, for all periods ending on the prior to the date of this Agreement, have administered, and between the date of this Agreement and the date of Closing, will administer each Employee Plan in compliance in all material respects with the reporting, disclosure, fiduciary and all other requirements applicable thereto under ERISA, the Code or any other applicable law.
- (f) Neither the Corporation, any Subsidiary nor any Seller has engaged in any transaction or acted or failed to act in a manner that violates the fiduciary requirements of Section 404 of ERISA with respect to any Employee Plans, and will not so engage, act or fail to act prior to the date of Closing. Neither the Corporation, any Subsidiary nor any Seller has engaged in any "prohibited transaction" within the meaning of Section 406(a) or 406(b) of ERISA, or of Section 4975(c) of the Code with respect to any Employee Plan which is not otherwise exempted by law, regulation or issued prohibited transaction exemption.

Furthermore, to the knowledge of the Corporation or any Subsidiary, no other "party in interest," as defined in Section 3(14) of ERISA, or "disqualified person," as defined in Section 4975(e)(2) of the Code, has engaged in any such "prohibited transaction."

- (g) Except as set forth on Schedule 4.15A(g), no Employee Plan provides benefits, including without limitation, death, disability, or medical benefits (whether or not insured), with respect to current or former employees of the Corporation or any of its Subsidiaries beyond their retirement or other termination of service other than (i) coverage mandated by applicable law, (ii) death, disability or retirement benefits under any Pension Plan, (iii) deferred compensation benefits, or (iv) benefits, the full cost of which is borne by the current or former employee (or his or her beneficiary). Except as set forth on Schedule 4.15A(g), there are no retiree medical or health plans which were, at any time during the past six (6) years, maintained by the Corporation or any Subsidiary and which have been discontinued.
- (h) Except as set forth on Schedule 4.15A(c), the Welfare Plans that are group health plans (as defined for the purposes of Section 4980B of the Code and Part 6 of Subtitle B of Title I of ERISA, and all regulations thereunder, (such provisions of law and regulations are hereinafter referred to "COBRA")) have complied in all material respects at all times, and will continue to comply in all material respects through the date of Closing, with requirements of COBRA. There are no pending, and to the knowledge of the Corporation or any Subsidiary, threatened claims, suits, or other proceedings by any employee, former employee, participants or by the beneficiary, dependent or representative of any such person, involving the failure of any Welfare Plan or of any other group health plan ever maintained by the Corporation or its Subsidiaries to comply with the health care continuation coverage requirements of COBRA.
- (i) Except as set forth on Schedule 4.15A(c), each Pension Plan is intended to be "qualified" within the meaning of Section 401(a) of the Code, and has been intended to be qualified during the period from the date of its adoption to the date of this Agreement, and each trust created thereunder is intended to be tax-exempt under Section 501(a) of the Code. Sellers have delivered or caused to be delivered to Buyers the latest determination letters of the Internal Revenue Service relating to each Pension Plan. Such determination letters have not been revoked. Furthermore, there are no pending proceedings or, to the knowledge of the Corporation or any Subsidiary, threatened proceedings in which the "qualified" status of any Pension Plan

is at issue and in which revocation of the determination letter has been threatened. Each such Pension Plan has not been amended or operated, since the receipt of the most recent determination letter, in a manner that would adversely affect the "qualified" status of the Plan and which cannot be cured without a material adverse effect. To the knowledge of the Corporation or any Subsidiary, there has been no partial termination as defined in Section 411(d) of the Code and the regulations thereunder, of any Pension Plan.

- (j) The Corporation or its Subsidiaries have made all required contributions under each Pension Plan on a timely basis or, if not yet due, adequate accruals therefore have been provided for in the Financial Statements. No Pension Plan has incurred any "accumulated funding deficiency" within the meaning of Section 302 of ERISA or Section 412 of the Code and no Pension Plan has applied for or received a waiver of the minimum funding standards imposed by Section 412 of the Code.
- (k) Except for required premium payments, no liability to the Pension Benefit Guaranty Corporation (the "PBGC") has been incurred by the Corporation or its Subsidiaries with respect to any Pension Plan. The Corporation or its Subsidiaries have complied with all requirements for premium payments, including any interest and penalty charges for late payment, due to PBGC with respect to each Pension Plan for which any premiums are required. No proceedings to terminate, pursuant to Section 4042 of ERISA, have been instituted or, to the knowledge of the Corporation or any Subsidiary, are threatened by the PBGC with respect to any Pension Plan (or any Pension Plan maintained by an ERISA Affiliate). There has been no termination, as defined in Section 411(d) of the Code and the regulations thereunder, of any Pension Plan. No reportable event, within the meaning of Section 4043 of ERISA, the required notice for which has not been waived by regulation, has occurred with respect to any Pension Plan.
- (1) The Corporation and its ERISA Affiliates have not been, nor will they become through the date of Closing, liable to contribute to any "multiemployer plan" (as defined in Section 3(37) of ERISA).
- (m) There has been no amendment to, written interpretation or announcement (whether or not written) by the Corporation or any Subsidiary relating to, or change in employee participation or coverage under, any Employee Plan or Employee Contract that would increase materially the expense of maintaining such Employee Plan or Employee Contract above the level of expense incurred in respect of such Employee Plan or Employee Contract for the most recent plan year with respect to

Employee Plans or the most recent fiscal year with respect to Employer Contracts.

- (n) There is no contract, agreement, plan or arrangement covering any employee or former employee of the Corporation or its Subsidiaries that, individually or in aggregate, could give rise to the payment by the Corporation or any Subsidiary, directly or indirectly, of any amount that would not be deductible pursuant to the terms of Section 280G of the Code.
 - 4.15B United Kingdom Employee Benefit Plans.
- (a) The representations and warranties set forth in this Section 4.15B shall apply to all employee benefit plans or arrangements to which the laws of England and Wales apply. In this Section 4.15B, references to paragraphs are references to paragraphs in this Subsection and the following terms shall have the following meanings:
 - (1) "the Company" means Godiva Products Ltd. and each of the other U.K. Subsidiaries;
 - (2) "Employees" means any current or former director, officer, employee or consultant employee of the Company; and
 - (3) "the Pension Schemes" means each of the Godiva Group Pension Scheme and the Godiva Products Limited Scheme.
 - (i) Neither of the Pension Schemes has been nor shall be placed in full or partial winding-up before the Closing Date and no amendments shall be made to either of the Pension Schemes before the Closing Date (whatever the effective date of such winding-up or amendment).
 - (ii) The Corporation represents and warrants, except as otherwise disclosed on Schedule 4.15B, as follows:
 - (1) that it has disclosed to Buyer details of all benefits payable or prospectively payable under each of the Pension Schemes to or in respect of all active members, pensioners and deferred pensioners, including any augmentations of benefits;
 - (2) that, other than the Pension Schemes, there are no pension, share option, share

incentive or similar schemes for any Employees and none of Sellers, the Corporation and the Company has any obligation (whether legally binding or established by custom) to pay any pension or make any other payment after retirement or death or otherwise to provide "relevant benefits" within the meaning of section 612 of the Income and Corporation Taxes Act 1988 to or in respect of any Employee and that none of Sellers, the Corporation and the Company is a party to any scheme or arrangement having as its purpose or one of its purposes the making of such payments or the provision of such benefits;

- (3) that each of the Pension Schemes complies with and has at all times complied with the provisions of the relevant legislation and the requirements of the Pension Schemes Office and Occupational Pensions Board affecting schemes approved or capable of approval under Chapter I of Part XIV of the Income and Corporation Taxes Act 1988;
- (4) that Company and Sellers have supplied to the Buyer complete and accurate copies of all trust deeds, rules, resolutions, announcements and booklets governing the Pension Schemes;
- (5) that Company and Sellers have supplied to the Buyer copies of the three most recent actuarial valuations and accounts of each of the Pension Schemes, together with copies of any certifications (if appropriate) made to the Inland Revenue under the Finance Act 1986 or Schedule 22 to the Income and Corporation Taxes Act 1988;
- (6) that the Company and the trustees of each of the Pension Schemes have duly complied with their respective obligations under the trust deeds and the rules thereof and under the aforementioned legislation and requirements;

- (7) that all amounts due to the trustees of each of the Pension Schemes or to any insurance company in connection therewith have been paid;
- (8) that no employer other than Godiva Group Limited participates in the Godiva Group Pension Scheme and no employer other than Godiva Products Limited participates in the Godiva Products Limited Scheme; and
- (9) that neither the Company nor the trustees of either of the Pension Schemes is engaged in any litigation or arbitration proceedings in respect of any retirement benefits scheme (as defined in Section 611 of the Income and Corporation Taxes Act 1988) or any benefit provided thereunder in relation to the employees or former employees of the Company and that there are no current submissions or referrals to the Pensions Ombudsman or to the Occupational Pensions Advisory Service in respect of the Company or such Pension Scheme.
- (iii) There are no circumstances which may give rise to a debt which shall be owing or shall become due from the Corporation or any Subsidiary in respect of its participation in a retirement benefits scheme (as defined in section 611 of the Income and Corporation Taxes Act 1988) prior to the Closing Date as a result of the operation of Section 58B of the Social Security Pensions Act 1975, the Pension Schemes Act 1993 or otherwise.
- (b) The actuary appointed by the Buyer (the "Buyer's Actuary") has estimated the ongoing solvency ratio of the Godiva Group Pension Scheme as at January 1, 1994 to be 70%. In estimating this ongoing solvency ratio, the Buyer's Actuary has adopted the actuarial assumptions used by the Godiva Group Pension Scheme Actuary for the July 1, 1992 actuarial valuation and has also used information disclosed to the Buyer by the Corporation and the Subsidiaries and their Representatives. In estimating the ongoing solvency ratio of the Godiva Group Pension Scheme as at January 1, 1994, the Buyer's Actuary has taken account of:
 - (i) the payments totalling (pound)1.02m due under the Agreement dated December 30, 1993 between

Godiva Products Limited, Godiva Group Limited and the Trustees of the Godiva Group Pension Scheme. For the purpose of these calculations, these payments were assumed to be held by the Trustees at January 1, 1994 and to have been dealt with consistently with the invested assets of the Scheme; and

(ii) the insured pensions which are an asset and a liability of the Scheme.

If and to the extent that (i) an inaccuracy or inaccuracies in the information disclosed or (ii) new information learned by the Buyer, shall result in the aggregate in a five per cent (5%) or more worsening of the solvency ratio of the Godiva Group Pension Scheme on an ongoing basis as at January 1, 1994 from that reflected in the preceding paragraph, this will constitute a material breach (but will not result in any liability or any right of indemnification of any kind under this Agreement) but the Buyer shall have the right to terminate this Agreement at any time prior to the Closing Date.

Transactions with Certain Persons. Except as set forth on Schedule 4.16, no Seller nor any other shareholder, officer, director or employee of the Corporation or any Subsidiary, nor any Person related to any Seller or any such shareholder, officer, director or employee by blood or marriage, nor any corporation, partnership, trust or other entity in which any such Person has a substantial interest as a shareholder, officer, director, trustee, partner or otherwise, or any Affiliate of any of the foregoing (each, a "Related Person") is presently or at any time during the past five years has been a party to any transaction with the Corporation or any Subsidiary, including, without limitation, any contract, agreement or other arrangement (a) providing for the furnishing of material services to or by, (b) providing for the rental or sale of real or personal property to or from, or (c) otherwise requiring payments to or from (other than for services as officers, directors or employees of the Corporation or any Subsidiary or for reimbursement of expenses in accordance with established policies), such Related Person. Except as set forth on Schedule 4.16, there is no outstanding amount owing (including, without limitation, pursuant to any advance, note or other indebtedness instrument) from the Corporation or any Subsidiary to any Related Person or from any Related Person to the Corporation or any Subsidiary. Each of the related-party transactions set forth on Schedule 4.16 (i) was entered into between the Corporation or any Subsidiary and the Related Person on an arms length basis on terms no less favorable to the

Corporation or the Subsidiary than could be obtained from an unrelated third party and (ii) shall, to the extent outstanding immediately after the Closing (excluding this Agreement and all related agreements entered into in connection herewith) automatically terminate without action of the parties unless otherwise specifically agreed to by the Buyer, and no party thereto shall have any continuing obligation or liability whatsoever under such agreement.

4.17A Tax Matters - General. Except as set forth on Schedule 4.17A, and except in relation to U.K. Subsidiaries $\dot{\text{(a)}}$ the Corporation and each Subsidiary have duly filed all tax reports and returns required to be filed (including, but not limited to, all federal, state, local and foreign $\tan x$ returns and reports) with any Governmental Authority and all such returns and reports were correct and complete in all material respects; (b) the federal income tax returns of the Corporation and all Subsidiaries (other than any other Subsidiary domiciled in any country other than the United States) are and have been at all times in the past prepared and filed on a consolidated basis for federal income tax purposes; (c) the Corporation and each Subsidiary have paid in full all taxes required to be paid by the Corporation and each Subsidiary before such payment became delinquent, no deficiencies have been or will be assessed with respect thereto for any period through December 31, 1993, and adequate reserves have been accrued on the Interim Financial Statements for all periods after January 1, 1994; (d) all taxes which the Corporation and each Subsidiary have been required to collect or withhold have been duly collected or withheld and, to the extent required when due, have been or will be duly paid to the proper taxing authority; (e) the income tax returns of the Corporation and each Subsidiary have not been examined by any Governmental Authority for any period on or after December 31, 1989, there are no audits known by the Corporation or any Subsidiary to be pending of the Corporation's or any Subsidiary's tax returns, and there are no claims which have been asserted relating to the Corporation's tax returns filed for any year; (f) neither the Corporation nor any Subsidiary is a party to any tax- sharing agreement or similar arrangement with any other party; (g) there are no federal, state, local or foreign tax liens upon any of the properties or assets of the Corporation or any Subsidiary and there are no unpaid taxes which are or could become a lien on the properties or assets of the Corporation, except for current taxes not yet due and payable; and (h) there have been no waivers of statutes of limitations by the Corporation or any Subsidiary with respect to any Governmental Authority. Correct and complete copies of all tax returns and reports of the Corporation and each Subsidiary requested by Buyer or any of Buyer's Representatives have been, or will be, provided

- 34 -

to Buyer. For the purpose of this Agreement, any income, franchise, sales, value added, use, transfer, payroll, social security, personal property, real property, occupancy or other tax, levy, impost, fee, imposition, assessment or similar charge, together with any related addition to tax, interest or penalty thereon, of any Governmental Authority, is referred to as a "tax." Neither the Corporation nor any Subsidiary has filed a consent pursuant to, or made an election under, Section 341(f) of the Code. Neither the Corporation nor any Subsidiary has agreed or been required to make any adjustment under Section 481(a) of the Code by reason of a change in accounting method or otherwise.

4.17B Tax Matters - United Kingdom. Except as set forth on Schedule 4.17B and in relation to U.K. Subsidiaries only, (a) for the past $\sin x$ years each U.K. Subsidiary has made all such returns and provided all such information in relation to tax to any Governmental Authority (including for the avoidance of doubt the Inland Revenue and H.M. Customs & Excise) as are required to be made or provided and all such returns and information were correct and complete in all material respects; (b) each U.K. Subsidiary has paid in full all tax required to be paid by it and is not now and has not been liable to pay any penalty, interest, surcharge or fine with respect thereto in the three years prior to the date of this Agreement; (c) all tax which each U.K. Subsidiary has been required to collect or withhold (including tax required to be withheld under the Pay-As-You-Earn system in respect of income tax and national insurance contributions) has been duly collected or withheld and each U.K. Subsidiary has complied with all requirements to account to the appropriate authority for tax so deducted or withheld; (d) none of the corporation tax returns of any U.K. Subsidiary is disputed by any Governmental Authority, no U.K. Subsidiary is, and no U.K. Subsidiary expects to be, involved in any dispute in relation to tax and no Governmental Authority (including for the avoidance of doubt the Inland Revenue and H.M. Customs & Excise) has investigated or indicated that it intends to investigate the tax affairs of any U.K. subsidiary; (e) all documents by virtue of which any U.K. Subsidiary has any rights which are required to be stamped have been duly stamped and all duty, interest and penalties on those documents have been duly paid; (f) no U.K. Subsidiary has any unsatisfied liability to stamp duty reserve tax or any interest or penalties on stamp duty reserve tax; (g) any U.K. Subsidiary is registered for the purposes of the Value Added Tax Act 1983 and has in all material respect made, given retained and kept complete, correct and up-to-date records, invoices and other documents which are required for the purposes of the Value Added Tax Act 1983 and has not been required by H.M. Customs & Excise to give security; (h) none of the U.K. subsidiaries have at any time within the $\bar{\text{period}}$ of six years

ending with the date of this Agreement acquired any asset from any company which at the time of acquisition was a member of the same group and in relation to which charge to tax might arise under either section 178 or 179 Taxation of Chargeable Gains Act 1992; (i) no U.K. Subsidiary is or may be liable to make any payment to any company (other than another U.K. Subsidiary) in respect of a claim for group relief and no U.K. Subsidiary is or may be liable to refund any such payment (other than to another U.K. Subsidiary) for any period ending prior to the date hereof; (j) no event has occurred and no circumstances exist as a result of which any lien charge or power of sale in favor of any Governmental Authority may exist over any of the shares or assets of any U.K. Subsidiary in respect of unpaid taxes; (k) no act or transaction has been effected by any U.K. Subsidiary, in consequence of which any U.K. Subsidiary is or may be held liable for tax primarily chargeable against some other person (other than another U.K. Subsidiary). Copies of corporation tax returns of each U.K. Subsidiary for the past three years requested by Buyer or any of Buyer's Representatives have been, or will be provided to Buyer.

Insurance. Schedule 4.18 contains a complete and accurate list of all policies or binders of fire, product liability, automobile liability, general liability, worker's compensation and other forms of insurance (showing as to each policy or binder the carrier, policy number coverage limits, expiration dates, annual premiums and a general description of the type of coverage provided) maintained by the Corporation and each Subsidiary and relating to the Corporation's and each Subsidiary's properties and assets or personnel (collectively, the "Insurance"). All of the Insurance is, and at all times prior to the Closing will be, sufficient for compliance in all material respects with all requirements of law and of all contracts to which the Corporation or any Subsidiary is a party. Neither the Corporation nor any Subsidiary is in default in any respect under any of the Insurance, and neither the Corporation nor any Subsidiary has failed to give any notice or to present any claim under any of the Insurance in a due and timely fashion. the knowledge of the Corporation or any Subsidiary, there are no facts upon which any insurer might be justified in reducing coverage or increasing premiums more than is normal or customary on any of the existing Insurance. No notice of cancellation or termination has been received with respect to any of the Insurance, and all premiums with respect to any of the Insurance have been, and will at all times through the Closing Date be, timely paid. The Insurance provides sufficient coverage for the assets and operations of the Corporation and each Subsidiary, is in full force and effect and will be kept in full force and effect by the Corporation and each Subsidiary through the Closing

Date. Except as disclosed on Schedule 4.18, there will be no retrospective adjustments of insurance premiums or other charges on or with respect to any of the Insurance for any period or occurrence through the Closing Date.

- 4.19 Inventory. The inventory of the Corporation and each Subsidiary ("Inventory") consists only of items of quality and quantity commercially usable and salable in the ordinary course of the business of the Corporation and each Subsidiary. The Inventory is reasonably related to the normal demands of the business of the Corporation and each Subsidiary as currently conducted by the Corporation and each Subsidiary. Except as set forth on Schedule 4.19, on the Closing Date, none of the Inventory will be damaged, obsolete, excess or for any other reason unusable or unsalable at normal prices. Except as set forth on Schedule 4.19, none of the Inventory is on consignment or subject to any Encumbrance. For purposes of this Section 4.19, the term "usable and saleable" shall mean Inventory which is in good and merchantable condition and of the quality regularly sold to customers of the Corporation and any Subsidiary in the usual course of business. Except as set forth on Schedule 4.19, (a) all of the Inventory is owned by the Corporation and each Subsidiary free of any Encumbrance and is located at the Current Real Property and (b) the Inventory as reflected in the Financial Statements and Interim Financial Statements has been valued at the lower of cost or market, in a manner consistent with past practices and procedures (including, without limitation, the method of computing overhead and other indirect expenses to be applied to inventory) and in accordance with GAAP.
- 4.20 Accounts Receivable. All of the accounts receivable of the Corporation and each Subsidiary (the "Accounts Receivable") are bona fide receivables, are reflected on the books and records of the Corporation and each Subsidiary, arose in the ordinary course of the business of the Corporation and each Subsidiary and are expected by the Corporation and each Subsidiary to be collected at their full face amount, net of reserves, as reflected on the Interim Financial Statements for the period ended March 31, 1994. Except as set forth on Schedule 4.20, or as reflected on the Financial Statements or the Interim Financial Statements, no Person has any lien or other Encumbrance on the Accounts Receivable, there is no right of offset against any of the Accounts Receivable, and no agreement for deduction or discount has been made with respect to any of the Accounts Receivable.
- 4.21 Contracts. Schedule 4.21 contains a true and correct list of contracts and other agreements to which the Corporation or any Subsidiary is a party or by which the

Corporation or any Subsidiary is bound, including, without limitation, all distributor and sales representative and dealer agreements, leases, licenses, indebtedness instruments, letters of credit, performance bonds, currency contracts, agreements with respect to guaranties, suretyships, mortgages, pledges, Encumbrances, security interests, covenants not to compete or indemnification by or for the benefit of the Corporation or any Subsidiary, and purchase and sale orders, grants and other financial concessions, and all amendments relating to any of the foregoing (the "Contracts"); provided, however, that the term "Contracts" shall not include any blanket inventory purchase order in an aggregate amount of less than \$500,000 annually and of a duration of less than one year, any other purchase and sale order under \$50,000, or any agreements relating to office equipment, production support equipment, maintenance, security or utilities, or other agreements which, in the aggregate for all agreements with any one Person, result in the incurrence of annual expenditures of less than \$50,000. True, correct and complete copies of all of the written Contracts have been delivered to Buyer. Each of the Contracts is valid, binding and enforceable by the Corporation or the applicable Subsidiary in accordance with its terms, was entered into in the ordinary course of business and is not subject to termination except in accordance with its terms or by reason of a Change of Control. Except as set forth in Schedule 4.21, each of the Contracts is in full force and effect, all fees, rents, royalties and other payments due thereunder are current, neither the Corporation, any Subsidiary nor any other party is in default thereunder or in breach thereof, and neither the Corporation nor any Subsidiary has during the past five years sought or obtained any waiver of or under any provision of any Contract (including, without limitation, any waiver from any lender or other creditor of any term, condition or default under any Contract). There exists no event or occurrence, condition or act which constitutes or, with the giving of notice or the lapse of time, would become, a default by the Corporation, any Subsidiary or any other party under any of the Contracts and neither the Corporation nor any Subsidiary has received notice of or knows of a threatened default under any of the Contracts. Except as set forth in Schedule 4.21, neither the Corporation nor any Subsidiary is a party to any contract or commitment which was not entered into in the ordinary course of business, which requires the Corporation and/or any Subsidiary to make any capital expenditure in excess of \$25,000, which has a term of greater than one year (other than contracts which are cancelable without penalty to the Corporation or the applicable Subsidiary in sixty (60) days or less), which contains terms other than on arms-length basis or which are in excess of the normal business requirements of the business of the Corporation or any Subsidiary. There are no practices in which

the Corporation or any Subsidiary is engaged, which are void, illegal, unenforceable, registrable or notifiable under or which contravene the Restrictive Trade Practices Act 1976, the Fair Trading Act 1973, Article 85 of the Treaty of Rome and regulations made thereunder or under EEC Council Regulation 4064/89 and regulations made thereunder or any other anti-trust or similar legislation anywhere in the world (all such legislation and regulations being referred to as the "anti-trust rules" in this Section 4.21). None of the practices of the Corporation or any Subsidiary constitutes an anti-competitive practice within the meaning of the Competition Act 1980. Neither the Corporation nor any Subsidiary has registered any currently binding agreement or arrangements under the Restrictive Trade Practices Act 1976 or filed any notification or application for exemption with the Commission of the European Communities, neither has it received any complaint or threat to complain under or referring to the anti-trust rules from any Person and has not received any request for information, investigation or objections or been the addressee of or party to any decision, judgment, undertaking or settlement relating to the anti-trust rules or to any proceedings in which the anti-trust rules were pleaded or relied upon.

- 4.22 Suppliers and Customers. Except as set forth on Schedule 4.22, no substantial supplier or customer (accounting for more than 2% of aggregate annual purchases or more than 2% of aggregate annual revenues, as the case may be, of the Corporation or any Subsidiary) has indicated to the Corporation or any Subsidiary that it intends to terminate its relationship with the Corporation or any Subsidiary and neither the Corporation nor any Subsidiary is aware of any such supplier or customer that intends to terminate such relationship or of any material problem or dispute with any such supplier or customer. To the knowledge of the Corporation or any Subsidiary, the Corporation and each Subsidiary has good business relationships with each such supplier and customer. Neither the Corporation nor any Subsidiary has any reason to believe that the consummation of the transactions contemplated hereby will or might disrupt the existing relationships with any such supplier or customer.
- 4.23 Business Records. All material records of accounts, personnel records and other business records (a) for the past seven years relating to the Corporation or any U.S. Subsidiary and (b) for all periods on or after September, 1989 relating to any U.K. Subsidiary, have not been destroyed and are available upon request. In addition, all material business records relating to periods prior to such seven-year with respect to the Corporation or any Subsidiary (other than any U.K.

Subsidiary) which are required to be maintained have not been destroyed and are available upon request.

- 4.24 Bank Accounts, Directors and Officers. Schedule 4.24 contains (a) a true, complete and correct list of all bank accounts and safe deposit boxes maintained by the Corporation and each Subsidiary and all persons entitled to draw thereon, to withdraw therefrom or with access thereto, (b) a description of all lock box arrangements for the Corporation and each Subsidiary, (c) the names of all the directors and officers of the Corporation and each Subsidiary and (d) a true, complete and correct list of all currently valid powers of attorney executed by the Corporation and each Subsidiary.
- 4.25 Environmental Matters. Except as disclosed in Schedule 4.25, to the best of the Corporation's knowledge, the Corporation, each Subsidiary, each Predecessor and their assets, properties and operations are now and, at all times prior to the Closing Date, have been in compliance with all Environmental Laws. There has been and is no Release or threatened Release of any Hazardous Substance at, on, under, in, to or from any of the Current Real Property (or, to the knowledge of the Corporation or any Subsidiary, at, on, under, in, to or from any of the Real Property) whether as a result of or in connection with the operations and activities at the Real Property or otherwise, except as disclosed in Schedule 4.25. None of the Corporation, any Subsidiary, any Predecessor or any Seller has received any notice of alleged, actual or potential responsibility for, or any inquiry or investigation regarding, the presence, Release or threatened Release of any Hazardous Substance at any location, whether at the Real Property or otherwise, which Hazardous Substances were allegedly manufactured, used, generated, processed, treated, stored, disposed or otherwise handled at or transported from the Real Property or otherwise, except as set forth in Schedule 4.25. None of the Corporation, any Subsidiary, any Predecessor or any Seller has received any notice of any other claim, demand or action by any Person alleging any actual or threatened injury or damage to any Person, property, natural resource or the environment arising from or relating to the presence, Release or threatened Release of any Hazardous Substances at, on, under, in, to or from the Real Property or in connection with any operations or activities thereat, except as set forth on Schedule 4.25. Neither the Current Real Property nor any operations or activities thereat is or has been subject to any judicial or administrative proceeding, order, consent, agreement or any lien relating to any Environmental Laws or Environmental Claims. Except as set forth on Schedule 4.25, (a) there are no underground storage tanks presently located at the Current Real Property and to the knowledge of the Corporation or

any Subsidiary, there have been no releases of any Hazardous Substances from any underground storage tanks or related piping at the Current Real Property, (b) there are no PCBs located at, on or in the Current Real Property and (c) there is no asbestos or friable asbestos-containing material located at, on or in the Current Real Property. To the knowledge of the Corporation and any Subsidiary, the Corporation and each Subsidiary have delivered to Buyer or its Representatives copies of all information requested by Buyer which has been supplied by or on behalf of the Corporation and any Subsidiary to any Governmental Authority having the duties of regulation, registration, authorization or enforcement of or under any Environmental Laws. Notwithstanding anything to the contrary contained in this Agreement, all representations and warranties in this Section given by the Corporation (or the officers signing the certificates referred to in Article 4(B) or Section 8.13) concerning or relating to the Real Property which does not constitute Current Real Property are given solely to the actual knowledge of management, without any independent duty of investigation or inquiry. In giving such representations and warranties, management has made no search of the files or records of the Corporation or any other Person.

- 4.26 Absence of Certain Changes. Except as set forth in Schedule 4.26, since December 31, 1993 there has not been:
- (a) any material adverse change in the business, financial condition or operations of the Corporation or any Subsidiary (including, without limitation, any material adverse change relating to the foundry sand issue disclosed at item B.1 on Schedule 4.25);
- (b) any damage, destruction, casualty or loss, whether covered by insurance or not, materially and adversely affecting the properties or Business of the Corporation or any Subsidiary;
- (c) any declaration, payment or setting aside for payment of any dividend or other distribution (whether in cash, stock or property) with respect to the capital stock, of the Corporation, or any direct or indirect redemption, purchase or other acquisition of any shares of capital stock of the Corporation;
- (d) any increase in the compensation of or granting of bonuses payable or to become payable by the Corporation or any Subsidiary to any officer or employee except increases occurring in the ordinary course of business and in amounts consistent with past practice and procedures;

business; or

- (e) any sale or transfer by the Corporation or any Subsidiary of any tangible or intangible asset, any mortgage or pledge or creation of any Encumbrance relating to any such asset, any lease of real property or equipment, or any cancellation of any debt or claim, except in the ordinary course of business;
 - (f) any other transaction not in the ordinary course of
 - (g) any change in accounting methods or principles.
- 4.27 No Brokers. Except for the obligation of Sellers to pay a certain transaction fee to McBain, Rose Partners, a California general partnership, which obligation is the sole responsibility of and shall be paid for by Sellers, none of the Corporation, any Subsidiary or any Seller has entered into any agreement, arrangement or understanding with any Person which will result in the obligation to pay any finder's fee, brokerage commission or similar payment in connection with the transactions contemplated hereby.
- 4.28 Absence of Certain Payments. None of the Corporation, any Subsidiary or any Affiliate or any of their respective officers, directors, employees or agents or other people acting on behalf of any of them, or any Affiliate of any of the foregoing, have (i) engaged in any activity prohibited by the United States Foreign Corrupt Practices Act of 1977 or any other similar law, regulation, decree, directive or order of any Governmental Authority and (ii) without limiting the generality of the preceding clause (i), used any corporate or other funds for unlawful contributions, payments, gifts or entertainment, or made any unlawful expenditures relating to political activity to officials of any Governmental Authority. None of the Corporation, any Subsidiary or any of their respective directors, officers, employees or agents or other Persons acting on behalf of any of them, or any Affiliate of any of the foregoing, has accepted or received any unlawful contributions, payments, gifts or expenditures.
 - 4.29 Products; Product Warranties.
- (a) The form of each product warranty relating to products currently manufactured or sold by the Corporation and each Subsidiary is attached to or set forth in Schedule 4.29. There has been no material change in any of the standard product warranties offered by the Corporation or any Subsidiary in the past five years.

- 42 -

- (b) Schedule 4.29 sets forth a true and complete list, of (i) all products manufactured, marketed or sold by the Corporation or any Subsidiary that have been recalled or withdrawn (whether voluntarily or otherwise) at any time during the past three years and (ii) all proceedings (whether completed or pending) at any time during the past three years seeking the recall, withdrawal, suspension or seizure of any product sold by the Corporation or any Subsidiary.
- (c) None of the Corporation or any Subsidiary is aware of any material defect in design, materials, manufacture or otherwise in any products heretofore or currently manufactured, distributed or sold by the Corporation or any Subsidiary or any defect in repair to or replacement of any such products which could give rise to any material claim.
- (d) Schedule 4.29 sets forth historical information regarding all warranty and other unreimbursed repair, maintenance and replacement expenses (including parts and labor, but not overhead) incurred by the Corporation and each Subsidiary with respect to products previously sold by it. The reserves for warranty expenses reflected on the Financial Statements and Interim Financial Statements are adequate and consistent with past practice and procedures.
- (e) Except as provided in any of the standard product warranties described in paragraph (a) of this Section 4.29 and as otherwise set forth on Schedule 4.29, neither the Corporation nor any Subsidiary has sold any products or services which are subject to an extended warranty beyond one year and which warranty has not yet expired.
- 4.30 Material Misstatements or Omissions. None of the representations or warranties by the Corporation in this Agreement contains any untrue statement of a material fact, or omits to state any material fact necessary to make the statements or facts contained therein not misleading.
 - B. Management Certificates.

Simultaneously with the execution and delivery of this Agreement, each of Peter J. Andrews, John J. O'Grady, Jr., Dale M. Clements, Michael J. Brayne, Garry S. Barber, Kurt H.C. Bottcher, Timothy Lamley, Louis J. Evangelista, Jr. and Michael Macomson, Jr. has executed and delivered to Buyer a certificate, in form and substance satisfactory to Buyer, certifying that, to his knowledge, each of the representations and warranties set forth in Sections 4.1 through 4.29, and all disclosures set forth on any of the Schedules attached to this Agreement, are true and

accurate in all material respects, and do not contain any untrue statements of a material fact.

C. Representations and Warranties of Sellers.

In addition, each Seller severally (but not jointly) represents and warrants to Buyer as follows:

- 4.31 Power and Authority; Authorized; Binding Effect. Such Seller has all necessary power and authority and has taken all action necessary to execute and deliver this Agreement, to consummate the transactions contemplated by this Agreement and to perform its obligations under this Agreement. This Agreement has been duly executed and delivered by such Seller and constitutes a legal, valid and binding obligation of such Seller enforceable against such Seller in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium or other similar laws relating to or affecting the rights of creditors generally, or by general principles of equity.
- 4.32 No Conflict or Violation. The execution and delivery of this Agreement, the consummation of the transactions contemplated by this Agreement and the fulfillment of the terms of this Agreement by such Seller (a) do not and will not constitute a breach of, or constitute an event, occurrence, condition or act which is or, with the giving of notice or the lapse of time, would become, a default under, or result in the acceleration of any obligations under, any term or provision of, any contract, agreement, indebtedness, encumbrance, commitment, license, franchise, permit, authorization or concession to which such Seller is a party, (b) do not and will not result in a violation by such Seller of any statute, rule, regulation, ordinance, code, order, judgment, writ, injunction, decree or award, and (c) do not and will not result in an imposition of any Encumbrance on the Shares held by such Seller.
- 4.33 Consents and Approvals. Except for any approvals required under the HSR Act, no consent, approval or authorization of, or declaration, filing or registration with, any Governmental Authority or other Person is required to be made or obtained by such Seller in connection with the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby.
- 4.34 No Proceedings. To the knowledge of such Seller, there is no action, order, writ, injunction, judgment or decree outstanding, or claim, suit, litigation, proceeding, arbitral action or investigation pending, or threatened or anticipated,

relating to or affecting in any adverse manner the transactions involving such Seller contemplated by this Agreement.

4.35 No Encumbrances. On the Closing Date, each Seller shall own all of the Shares issued in his, her or its name and set forth opposite his, her or its name on Exhibit A free of any Encumbrance and subject to no restriction with respect to transferability, other than restrictions generally applicable under federal or state securities laws.

ARTICLE 5 REPRESENTATIONS AND WARRANTIES OF BUYER

Buyer hereby represents and warrants to Sellers as follows:

- 5.1 Organization and Good Standing. Buyer is a Delaware corporation, duly organized, validly existing and in good standing under the laws of the state of Delaware.
- 5.2 Authorization; Binding Effect. Buyer has all necessary power and authority to execute and deliver this Agreement, to consummate the transactions contemplated by this Agreement and to perform its obligations under this Agreement. Copies of all resolutions of the board of directors of Buyer with respect to the transactions contemplated by this Agreement, certified by the Secretary or an Assistant Secretary of Buyer, in form satisfactory to counsel for Sellers, have been delivered to Sellers. This Agreement has been duly executed and delivered by Buyer and constitutes a legal, valid and binding obligation of Buyer, enforceable against Buyer in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium or other similar laws relating to or affecting the rights of creditors generally, or by general principles of equity.
- 5.3 No Conflict or Violation. The execution and delivery of this Agreement, the consummation of the transactions contemplated by this Agreement and the performance by Buyer of its obligations under this Agreement do not, and will not, result in (a) a violation of or a conflict with any provision of the Certificate of Incorporation, Bylaws or other organization certificates or documents of Buyer, (b) a breach of, or constitute an event, occurrence, condition or act which is, or with the giving of notice or the lapse of time would become, a default under, or result in the acceleration of any obligation under, any term or provision of any contract, agreement, indebtedness, lease, commitment, license, franchise, permit,

authorization or concession to which Buyer is a party or (c) a violation by Buyer of any statute, rule, regulation, ordinance, code, order, judgment, writ, injunction, decree or award.

- 5.4 Consents and Approvals. Except as required under the HSR Act and as otherwise set forth on Schedule 5.4, no consent, approval or authorization of, or declaration, filing or registration with, any Governmental Authority or other Person is required to be made or obtained by Buyer in connection with the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby.
- 5.5 No Proceedings. There is no action, order, writ, injunction, judgment or decree outstanding or claim, suit, litigation, proceeding, arbitral action or investigation pending, or to the knowledge of Buyer, threatened or anticipated against, relating to or affecting in any materially adverse manner the transactions contemplated by this Agreement.
- 5.6 No Brokers. Except for the obligation of IDEX to Richland-Gordon, which obligation is the sole responsibility of and shall be paid for by IDEX, neither Buyer nor IDEX has entered into or will enter into any agreement, arrangement or understanding with any Person which will result in the obligation to pay any finder's fee, brokerage commission or similar payment in connection with the transaction contemplated hereby.
- 5.7 Acquisition for Investment. The Shares to be purchased pursuant to the terms of this Agreement will be acquired by Buyer for investment for its own account and not with a view to a distribution or resale of any of such Shares. Each certificate representing the Shares shall bear a legend in substantially the following form (as well as any other legend required by any applicable blue sky or other state securities law):

"THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 OR ANY STATE SECURITIES LAW. THE SECURITIES MAY NOT BE SOLD OR TRANSFERRED IN THE ABSENCE OF EXEMPTION THEREFROM UNDER SAID ACT OR THE RULES AND REGULATIONS PROMULGATED THEREUNDER, OR UNDER ANY STATE SECURITIES LAW."

ARTICLE 6 COVENANTS AND CONDUCT OF THE PARTIES PRIOR TO CLOSING

The Corporation covenants with Buyer (with respect to each of Sections 6.1 through 6.8), each Seller severally (but not jointly) covenants with Buyer (with respect to Sections 6.4, 6.5(h), 6.6 and 6.9 only), and Buyer covenants with the Corporation and each Seller, as follows:

- on the date of this Agreement and ending on the Closing Date, Buyer and each Representative of Buyer may continue to conduct, at its own expense and during regular business hours, a due diligence review of the Corporation and each Subsidiary. Subject to the foregoing, Buyer and each Representative of Buyer shall be granted full access to all Current Real Property in connection with such due diligence review. In connection with such due diligence review (including, without limitation, the Pre-Acquisition Financial Review), the Corporation and each Subsidiary, and each Representative of the Corporation and each Subsidiary, shall, upon reasonable prior notice, (a) cooperate with Buyer and each Representative of Buyer, (b) provide all information, and all documents and other data relating to such information, reasonably requested by Buyer or any Representative of Buyer and (c) permit Buyer and each Representative of Buyer to inspect any assets of the Corporation and each Subsidiary.
- Environmental Audits. In addition to any 6.2 environmental investigations and audits conducted by Buyer or its Representatives with Sellers' approval prior to the date of this Agreement, Buyer shall, at Buyer's sole expense, be permitted to cause further environmental audits of the Real Property to be conducted assessing the presence and or disposition of Hazardous Substances and compliance with Environmental Laws; provided, however, that Buyer will not conduct or cause to be conducted any "Phase II" environmental audits or other invasive testing procedures without the express prior written consent of Seller's Agent. Subject to the foregoing, and to the provisions of Section 6.1, the Corporation and each Subsidiary hereby grants a license to Buyer's qualified environmental consultants to enter upon the Real Property, upon giving the Corporation, or any Subsidiary reasonable notice, with men and materials to conduct such environmental audits. Buyer covenants and agrees that it will not contact any Governmental Authority or any other Person regarding, or convey to prior to Closing, any information derived from environmental investigations and audits conducted by or on behalf of Buyer.

- 47 -

- 6.3 Matters Relating to Real Property. The Corporation and each Subsidiary has delivered, or will prior to Closing deliver, to Buyer (a) copies of all existing policies of title insurance which are in full force and effect and (b) to the extent within the reasonable control of the Corporation or any Subsidiary, evidence sufficient to cause the deletion of all Prohibited Exceptions from each title insurance commitment or title opinion obtained by Buyer with respect to the Current Real Property.
- 6.4 Consents and Best Efforts. Buyer, each Seller and the Corporation shall cooperate with each other to make all filings required under the HSR Act within two (2) days after the execution and delivery of this Agreement. In addition, as soon as practicable, Buyer, the Corporation, each Subsidiary and each Seller, as applicable, shall commence all reasonable actions to obtain all other consents, approvals, permits and agreements of, to give all notices to, and to make all filings with, any Person as may be necessary to be obtained, given or made by each of them, to authorize, approve or permit the full and complete sale, conveyance, assignment or transfer of the Shares, free of all Encumbrances, on or before the Closing Date.
- 6.5 Conduct of Business by the Corporation and Sellers Pending Closing. From the date of this Agreement through the Closing Date, and except as otherwise specifically provided in this Agreement or consented to or approved by Buyer in advance in writing (which approval shall not be unreasonably withheld), the Corporation, each Subsidiary and each Seller (with respect to Section 6.5(h) only) shall conform to the following:
- (a) The Corporation and each Subsidiary shall carry on its business substantially in the same manner as heretofore conducted and shall not engage in any transaction or activity, enter into any agreement or make any commitment except in the ordinary course of business.
- (b) No change or amendment shall be made in or to the certificate or articles of incorporation or other governing or organizational charter or instruments of the Corporation or any Subsidiary.
- (c) The Corporation shall not declare, pay or set aside for payment any dividend or other distribution (whether in cash, stock or property) with respect to its capital stock or directly or indirectly redeem, purchase or otherwise acquire any shares of its capital stock.
- (d) The Corporation and each Subsidiary shall use its best efforts to preserve its corporate existence and business $\,$

organization intact and to preserve its properties, assets and relationships with its employees, suppliers, customers and others with whom it has business relations.

- (e) Except for contracts or commitments made in the ordinary course of business of the Corporation and each Subsidiary, and except as contemplated by Section 10.8, no contracts or commitments shall be entered into by or on behalf of the Corporation or any Subsidiary. No amendments or changes shall be made to any of the Contracts.
- (f) Neither the Corporation nor any Subsidiary shall (i) grant any increase in compensation, other than normal merit and cost-of-living increases to employees who are not officers of the Corporation or any Subsidiary, or (ii) enter into, or amend in any respect, any Employee Plans or Employee Contracts.
- (g) Neither the Corporation nor any Subsidiary shall create, incur, assume or guarantee or otherwise become liable with respect to any indebtedness other than in the ordinary course of business of the Corporation or such Subsidiary. Neither the Corporation nor any Subsidiary shall alter, amend or otherwise modify the terms or conditions of the Closing Date Funded Debt.
- (h) None of the Corporation, any Subsidiary or any Seller shall take any action which would cause, or fail to take any action the failure of which would cause, any representation or warranty of the Corporation or any Seller in this Agreement to be breached or to be or become untrue in any material respect.
- (i) The Corporation shall, on the Closing Date, pay the Management Bonuses and amounts due under the Godiva Commitment.
- 6.6 Notification of Certain Matters. The Corporation and each Seller shall give prompt notice to Buyer, and Buyer shall give prompt written notice to the Corporation and Sellers of (a) the occurrence, or failure to occur, of any event which occurrence or failure would be likely to cause any of their respective representations or warranties contained in this Agreement to be breached or to be or become untrue or inaccurate in any material respect any time from the date of this Agreement to the Closing Date and (b) any failure of any of them to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it or him under this Agreement.
- 6.7 Delivery of Interim Financial Statements. Within fifteen (15) business days after the end of each month after the

execution of this Agreement and prior to the Closing Date, the Corporation shall deliver or cause to be delivered to Buyer monthly and year-to-date interim financial statements of the Corporation and its Subsidiaries; provided that the Corporation shall have forty-five (45) business days after the end of each month to provide such interim financial statements with respect to the U.K. Subsidiaries only. Upon delivery, such year-to-date interim financial statements shall automatically become and be deemed to be Interim Financial Statements for purposes of this Agreement.

- 6.8 Covenants of Certain Others. Each of HFP, HMTC and each officer identified in Section 8.10 shall execute and deliver to Buyer on the Closing Date the Restrictive Covenants Agreement in the form attached to this Agreement as Exhibit 8.10.
- 6.9 Distributions; Related Person Transactions; Brokers.
 (a) Each Seller severally (but not jointly) covenants and agrees that:
- (i) To the extent, if any, that such Seller received any dividend or other distribution on or with respect to any of the Shares during the period from January 1, 1994 through the Closing Date, such Seller shall promptly upon request of Buyer pay to Buyer an amount equal to the amount of such dividend or distribution;
- (ii) To the extent, if any, that such Seller or any Affiliate of such Seller is a party as of the Closing Date to any Related Person transaction (as defined in Section 4.16 and including, without limitation, the Stockholders Agreement dated April 6, 1987 disclosed on Schedule 4.6), all rights of such Seller or such Affiliate under or in connection with any such Related Person transaction shall automatically terminate as of the Closing Date, except that this sentence shall not apply to the Employee Contracts set forth in Schedule 4.21; it being understood that all accrued management fees (plus miscellaneous expenses) owing to McBain, Rose Partners and accrued through the Closing shall be paid by the Corporation at or prior to Closing against delivery of a receipt in form reasonably satisfactory to Buyer to the effect that all obligations of any nature whatsoever to McBain, Rose partners have been fully paid.
- (iii) To the extent, if any, that such Seller caused the Corporation or any Subsidiary or Buyer to become obligated in any way for a broker or finders or other similar fee arising out of or in connection with the transactions contemplated by this Agreement, such Seller shall, after the Closing, indemnify and

hold the Corporation, each Subsidiary and Buyer harmless from and against any and all such liability.

- (iv) To the extent, if any, that there are, after the Closing, any outstanding options, warrants, rights, contracts, commitments, understandings or arrangements by which the Corporation is bound to issue any capital stock of the Corporation or any unpaid Management Bonus or Godiva Commitment in excess of the \$452,580 and \$180,000 referred to in Sections 1.1(x) and 1.1(q), respectively, each Seller shall severally reimburse Buyer for the excess proceeds received by such Seller (calculated as the excess amount such Seller received over the amount that such Seller would have received if such options, warrants, rights, contracts, commitments, understandings or arrangements had been fully exercised prior to Closing, or to the extent of any such Management Bonus or Godiva Commitment so owed after the Closing) divided pro rata based on stock ownership prior to the Closing among the Sellers.
- (b) Buyer covenants and agrees that, to the extent, if any, that Buyer caused any Sellers to become obligated in any way for a broker or finder fee or other similar fee arising out of or in connection with the transactions contemplated by this Agreement, Buyer shall, after the Closing Date, indemnify and hold each Seller harmless from and against any and all such liability.

ARTICLE 7 CONDITIONS TO SELLERS' OBLIGATIONS

The obligation of Sellers to consummate the transaction contemplated by this Agreement is subject to the satisfaction or, in the discretion of Sellers, waiver, on or prior to the Closing Date, of each of the following conditions:

- 7.1 Representations, Warranties and Covenants. All representations and warranties of Buyer contained in this agreement shall be true and correct in all material respects at and as of the Closing Date, except as and to the extent that the facts and conditions upon which such representations and warranties are based are expressly required or permitted to be changed by the terms of this Agreement, and Buyer shall have performed in all material respects all agreements and covenants required by this Agreement to be performed by it prior to or at the Closing Date.
- 7.2 Consents, Approvals and Filings. All consents, authorizations and approvals from, and all declarations, filings

and registrations with, any Person required to consummate the transactions contemplated by this Agreement shall have been obtained or made.

- 7.3 No Proceedings. No action, order, writ, injunction, judgment or decree shall be outstanding, and no claim, suit, litigation, proceeding, arbitral action or investigation shall be pending, threatened or anticipated against Buyer, Sellers, the Corporation or any Subsidiary, seeking to enjoin, or materially adversely affecting, the transactions contemplated by this Agreement.
- 7.4 Closing Certificate. Buyer shall have furnished Sellers with a certificate of an officer of Buyer, in form and substance satisfactory to Sellers, to evidence compliance with the conditions set forth in this Article 7.
- 7.5 Legal Opinion. Sellers shall have received an opinion of Hodgson, Russ, Andrews, Woods & Goodyear, substantially in form and substance as set forth in Exhibit 7.4.
- 7.6 HSR Act. The applicable waiting period, including any extensions thereof, under the HSR Act shall have expired or been terminated and no action shall have been taken by the Federal Trade Commission or the Antitrust Division of the Justice Department to prevent the consummation of the transactions contemplated by this Agreement.
- 7.7 TCW Warrants. TCW shall have transferred to Buyer all Warrants, or shall have exercised the Warrants and transferred to Buyer all Shares issued in respect thereof in accordance with Section 3.3, or shall have otherwise surrendered the Warrants for cancellation, in exchange for the payment by Buyer described in Section 2.4.

ARTICLE 8 CONDITIONS TO BUYER'S OBLIGATIONS

The obligation of Buyer to consummate the transaction contemplated by this Agreement is subject to the satisfaction or, in the discretion of Buyer, waiver, on or prior to the Closing Date, of each of the following conditions:

8.1 Representations, Warranties and Covenants. All representations and warranties of Sellers, together with representations and warranties contained in any certificate or certificate(s) delivered by certain executive officers pursuant to Article 4B, and the Corporation contained in this Agreement

shall be true and correct in all material respects at and as of the Closing Date, except as and to the extent that the facts and conditions upon which such representations and warranties are based are expressly required or permitted to be changed by the terms of this Agreement, and the Corporation and Sellers shall have performed in all material respects all agreements and covenants required by this Agreement to be performed by them prior to or at the Closing Date.

- 8.2 Consents, Approvals and Filings. All consents, authorizations and approvals from, and all declarations, filings and registrations with, any Person required to consummate the transactions contemplated by this Agreement shall have been obtained or made.
- 8.3 No Proceedings. No action, order, writ, injunction, judgment or decree shall be outstanding, and no claim, suit, litigation, proceeding, arbitral action or investigation shall be pending, threatened or anticipated against Buyer, Sellers, the Corporation or any Subsidiary, seeking to enjoin, or materially adversely affecting, the transactions contemplated by this Agreement.
- 8.4 No Interruption or Adverse Change. (a) No interruption of suspension of the Business as now conducted shall have occurred or, to the knowledge of the Corporation, any Subsidiary or any Seller, been threatened, and (b) no material adverse change in the business, prospects, assets or financial condition of the Corporation and its Subsidiaries, taken as a whole, shall have occurred or, to the knowledge of the Corporation, any Subsidiary or any Seller, been threatened
- 8.5 Closing Certificate. The Corporation and each Seller shall have furnished Buyer with a certificate to evidence compliance with the conditions required to be fulfilled by it as set forth in this Article 8 (which, as to each Seller, shall include certification only as to the representations, warranties and covenants of such Seller).
- 8.6 Legal Opinion. Buyer shall receive (a) a favorable opinion from Latham & Watkins, counsel to the Corporation and Sellers, substantially in form and substance as set forth in Exhibit 8.6(a) and (b) favorable opinions from local counsel to the U.S. Subsidiaries, substantially in the form of Exhibit 8.6(b).
- 8.7 Resignations. At the Closing, all directors of the Corporation and each Subsidiary (other than the Virgin Island resident directors of Hale Products International Inc. and other

than, in the discretion of Buyer, the directors of the U.K. Subsidiaries) shall submit their resignations from their respective boards dated as of the Closing

8.8 HSR Act. The applicable waiting period, including any extensions thereof, under the HSR Act shall have expired or been terminated and no action shall have been taken by the Federal Trade Commission or the Antitrust Division of the Justice Department to prevent the consummation of the transactions contemplated by this Agreement.

8.9 [Reserved]

- 8.10 Restrictive Covenants Agreements. Each of HFP, HMTC, Peter J. Andrews, John J. O'Grady, Jr., Dale M. Clements, Michael R. Macomson, Jr., Alex G. Tyshovnytsky, Garry S. Barber, Michael J. Brayne and Kurt H. C. Bottcher shall have executed and delivered a Restrictive Covenants Agreement substantially in the form attached to this Agreement as Exhibit 8.10.
- 8.11 Terms of Closing Date Funded Debt. There shall have been no change in the terms and conditions of the Closing Date Funded Debt.
- 8.12 TCW Warrants. TCW shall have transferred to Buyer all warrants, or shall have exercised the Warrants and transferred to Buyer all Shares issued in respect thereof in accordance with Section 3.3, or shall have otherwise surrendered the Warrants for cancellation, in exchange for the payment by Buyer described in Section 2.4.
- 8.13 Certificates Pursuant to Article 4(B) and Additional Matters. Each of Peter J. Andrews, John J. O'Grady, Jr., Dale M. Clements, Alex G. Tyshovnytsky, Michael R. Macomson, Jr., Garry S. Barber, Louis J. Evangelista, Jr., Michael J. Brayne, Timothy Lamley and Kurt Bottcher shall have executed and delivered to Buyer a certificate, dated the Closing Date, in substantially the same form and content as delivered on the date of this Agreement pursuant to Article 4(B) and additionally certifying (a) that the Closing Date Cash is at least \$2,500,000 and the amount thereof, (b) the principal balance of the Closing Date Funded Debt and (c) that the Corporation and the Subsidiaries have paid accounts payable in the ordinary course of business.

- 54 -

ARTICLE 9 CONFIDENTIALITY; INDEMNIFICATION; FURTHER ASSURANCES

Confidentiality Agreement. Each Seller and each other Person who has executed this Agreement has had access to, and has in varying degrees gained knowledge with respect to, the Business of the Corporation and its Subsidiaries, its trade secrets, financial results and information, processes and techniques, technical production and cost data, methods of doing business and information concerning customs and suppliers, and other valuable and confidential information relating to the business of the Corporation and its Subsidiaries (the "Confidential Information"). Each Seller and each such other Person acknowledges that unauthorized disclosure or misuse of the Confidential Information could cause irreparable damage to the Corporation, the Subsidiaries and Buyer subsequent to the Closing, and agrees that covenants by Sellers not to make unauthorized disclosures of the Confidential Information are essential to the growth and stability of the business of the Corporation, the Subsidiaries and Buyer. Accordingly, each Seller and each such other Person agrees that none of them will use or disclose any Confidential Information obtained in the course of their past connection with the Business of the Corporation and its Subsidiaries, other than information generally available to the public through sources other than Sellers and each such other Person.

Nothing in this Section 9.1 shall prevent any Seller or such other Person from disclosing or using any such information as may be reasonably necessary (i) in connection with any lawsuit or arbitration arising out of this Agreement, (ii) in connection with any judicial or administrative filing, investigation or proceeding or otherwise with or by a governmental agency or (iii) as otherwise may be required by law. The covenants contained in this Section 9.1 shall expire on the third anniversary of the date of this Agreement.

Notwithstanding the foregoing, Angus McBain, Walter Rose and McBain Rose Partners shall be entitled to use or disclose information relating to the background of the Corporation, description of the Business, historical investment and sale terms (including capital structure and valuation information), and summary financial information of the Corporation (not substantially exceeding in detail or scope the information customarily presented on the "Selected Financial Data" page of a prospectus, as specified under Item 301 of Regulation S-K of the Securities and Exchange Commission) for periods prior to the Closing Date, whether or not such information constitutes Confidential Information hereunder,

provided that the purpose of such disclosure or use is not to compete with the Corporation or its Business and not to assist any competitor of the Corporation.

9.2 Survival and Indemnifications.

- (a) Survival of Representations, Warranties and Covenants. All statements contained in the disclosure schedules or in any certificate or instrument or conveyance delivered by or on behalf of the parties pursuant to this Agreement (including, without limitation, the certificates delivered under Article 4(B) and Section 8.13) or in connection with the transactions contemplated hereby shall be deemed to be representations and warranties by the party or parties hereunder. The representations and warranties of the Corporation, each Seller and Buyer (or other signatories) contained herein (or in such other documents), and the covenants of the Corporation and each Seller contained in Sections 2.3, 6.1, 6.2, 6.3, 6.4, 6.5(h), 6.6, 6.7 and 6.8 shall not survive the Closing, and from and after the Closing, no party hereto and no signatory to any such other document shall have any liability to any other party with respect to such representations and warranties and specified covenants. All covenants by the Corporation, each Seller and Buyer contained in Section 1.5, Article 2 (other than Section 2.3), Sections 6.5 (other than Section 6.5(h)) and 6.9, Article 9 and Article 10 of this Agreement shall survive the Closing Date (but with respect to Section 6.9(a)(iv), shall survive for a period of only twelve (12) months following Closing Date), and the covenants contained in the Confidentiality Letter referred to in Section 10.12 shall survive any termination of this Agreement.
- (b) Indemnification by Sellers. Each Seller (severally with respect to all obligations of such Seller hereunder) hereby agrees to defend, indemnify and hold harmless Buyer (before and after the Closing) and the Corporation and each Subsidiary (after the Closing only) from, against and in respect of any and all loss, liability, deficiency or damage suffered or incurred by Buyer (before or after the Closing), and by the Corporation and each Subsidiary (after the Closing only), by reason of the breach of nonfulfillment of any covenant or agreement by such Seller contained in Sections 1.5, 2.1, 2.5, 3.2, 6.4 and 6.9, Article 9 and Sections 10.5 and 10.8 of this Agreement, and any and all actions, suits, proceedings, claims, demands, assessments, judgments, costs and expenses (including, but not limited to, legal and experts' fees and expenses) incident to any of the foregoing. In no event, however, shall any Seller be obligated to indemnify Buyer or the Corporation for any amount in excess of the pro rata portion of the Purchase Price actually (or, if the Closing occurred, which would have

been) received (Transaction Costs being deemed to be received for purposes of this sentence) by Seller hereunder.

- (c) Indemnification by Corporation. The Corporation (with respect to all obligations of the Corporation prior to Closing) hereby agrees to defend, indemnify and hold harmless Buyer from, against and in respect of any and all loss, liability, deficiency or damage suffered or incurred by Buyer by reason of the breach or nonfulfillment of any covenant or agreement by the Corporation contained in Sections 2.3, 6.1, 6.2, 6.4, 6.5(g), 6.7 and 6.9, Article 9 and Sections 10.5 and 10.8 of this Agreement, and any and all actions, suits, proceedings, claims, demands, assessments, judgments, costs and expenses (including, but not limited to, legal and experts' fees and expenses) incident to any of the foregoing.
- (d) Indemnification by Buyer. Buyer hereby agrees to indemnify and hold harmless Sellers from, against, and in respect of any and all loss, liability, deficiency or damage suffered or incurred by Sellers resulting from the breach or nonfulfillment of any covenant or agreement by Buyer contained in Article 2, Article 6, Article 9 and Article 10 of this Agreement, and any and all actions, suits, proceedings, claims, demands, assessments, judgments, costs and expenses, (including, but not limited to, legal fees and expenses) incident to any of the foregoing.

(e) Third-Party Claims.

- (i) Except as otherwise specifically provided in this Agreement, in order for Buyer, the Corporation, any Subsidiary, or Sellers, as the case may be, to be entitled to any indemnification provided hereunder, in respect of, arising out of, or involving a claim made by any Person other than Buyer or Sellers against any indemnified party, the indemnified party must notify the indemnifying party in writing of such claim within twenty (20) days after receipt by the indemnified party of written notice of such claim. Thereafter, the indemnified party shall deliver to the indemnifying party, within twenty (20) days after receipt by the indemnified party, copies of all further notices relating to such claim.
- (ii) If a third-party claim is made for which Buyer or any Seller is entitled to indemnification pursuant to this Article 9, the indemnifying party will be entitled to assume primary responsibility for the defense of such claim with counsel selected by the indemnifying party, provided such counsel is not reasonably objected to by the indemnified party. Should the indemnifying party assume the defense of such claim, the

indemnifying party will not be liable to the indemnified party for any legal expenses subsequently incurred by the indemnified party in connection with the defense of such claim.

- (iii) If the indemnifying party assumes the defense of a third-party claim as set forth in this Section 9.2(d), then (A) in no event will an indemnified party admit any liability with respect to, or settle, compromise or discharge, any such claim without the indemnifying party's prior written consent, and (B) each indemnified party shall be entitled to participate in, but not control, the defense of such claim with its own counsel at its own expense. If the indemnifying party does not assume the defense of any such claim, an indemnified party may defend such claim in a manner as it may deem appropriate (including, but not limited to, settling such claim, after giving twenty (20) days prior written notice of such settlement to the indemnifying party, on such terms as the indemnified party may deem appropriate).
- (iv) In the event that any claim for indemnification is made with respect to any third-party claim pursuant to this Section 9.2(d), (A) the party assuming primary responsibility for the defense of such claim shall at all times keep the other party informed as to the status of such claim, and (B) the party not primarily responsible for the defense of such claim shall cooperate fully with the other party in connection with such defense.
- (f) Liability of each Seller Several and Not Joint; Sole Remedy. All obligations of Sellers under this Agreement shall be several and not joint, and each Seller shall be obligated to indemnify Buyer under this Agreement only for those losses, liabilities, deficiencies or damages arising by reason of such Seller's breach, if any. No Seller shall have any liability for the representations, warranties, covenants or other obligations of any other Seller or of the Corporation contained in this Agreement. The right of indemnification set forth in this Section 9.2 shall be the sole remedy of any party for any breach of a covenant contained herein. In the event that, for any reason, Closing does not occur hereunder, no party shall have any liability to any other for a breach of covenant other than for an intentional breach of those covenants contained in Section 2.1, 2.2, 2.3, 2.4, 3.2, 6.1, 6.2, 6.4, 6.5(g), 6.9, Article 9 or Sections 10.5 or 10.8, and no party (nor any signatory to any certificate or instrument or conveyance delivered hereunder) shall have any liability to any other for a breach of representation or warranty hereunder.

- 9.3 Further Assurances. Both before and for a reasonable period of time after the Closing Date, each party will cooperate in good faith with each other party and will take all appropriate action and execute any agreement, instrument or other writing of any kind which may be reasonably necessary or advisable to carry out and confirm the transactions contemplated by this Agreement (including, but not limited to, obtaining consents or approvals from any Person for the transfer of the Shares that are transferred subject to consents or approvals being obtained).
- 9.4 Liability of Sellers in Certain Circumstances. Notwithstanding anything herein contained in this Article or in this Agreement to the contrary, including without limitation this Article 9 and Section 6.9(iv), each Seller, for purposes of the obligation under Section 6.9(iv), shall be liable pro rata for TCW's pro rata share of the proceeds as if in fact such Seller had received TCW's share of the proceeds.

ARTICLE 10 MISCELLANEOUS

- $10.1\,$ Termination. This Agreement may be terminated at any time prior to the Closing Date as follows:
 - (a) By mutual written agreement of Buyer and

Sellers;

(b) By Buyer or Sellers holding a majority of the Shares by written notice to the other in the event that all approvals under the HSR Act have not been obtained, or the Closing Date has not occurred for any other reason, on or prior to June 15, 1994, but only if the terminating party (which, in the case of Sellers, shall mean each and every Seller) is not in material and intentional breach of, or default under, any covenant required to be performed by such party, under this Agreement.

In the event of the termination of this Agreement by any party as provided in the preceding sentence, no party shall have any liability hereunder of any nature whatsoever, other than for any breach of any covenant contained in this Agreement which occurs prior to such termination, or for any breach of the Confidentiality letter referred to in Section 10.12. In the event that a condition precedent to any party's obligations is not satisfied or waived, nothing contained in this Agreement shall be deemed to require any party to terminate this Agreement,

- 59 -

rather than to waive such condition precedent and proceed with the Closing; provided, however, that the Closing of this Agreement notwithstanding a party's actual knowledge of the failure of a condition constitutes a waiver by that party of such condition and of any claim or right relating to the subject matter of any such condition.

10.2 Notices. Unless otherwise provided in this Agreement, any notice, request, instruction or other communication to be given hereunder by any party to the other shall be in writing and (a) delivered personally, (b) mailed by first- class mail, postage prepaid, (such mailed notice to be effective four days after the date it is mailed) or (c) sent by facsimile transmission, with a confirmation sent by way of one of the above methods, as follows:

If to Management Sellers, addressed to:

Hale Products, Inc. 700 Spring Mill Avenue Conshohocken, Pennsylvania 19428 Attn: Peter J. Andrews Dale M. Clements John J. O'Grady, Jr.

If to any other Seller, addressed to:

HFP Partners, L.P. 355 South Grand Avenue, 42nd Floor Los Angeles, California 90071

With a copy to:

Latham & Watkins Attn: Robert A. Koenig, Esq. 633 West Fifth Street, Suite 4000 Los Angeles, California 90071 Telephone: (213) 485-1234 Telecopier: (213 891-8763

If to Buyer, addressed to:

IDEX Corporation 630 Dundee Road, Suite 400 Northbrook, Illinois 60065 Attn: Donald N. Boyce
Wayne P. Sayatovic
Telephone: (708) 498-7070
Telecopier: (708) 498-3940

With a copy to:

Hodgson, Russ, Andrews, Woods & Goodyear

Attn: Richard E. Heath, Esq.
David V.L. Bradley, Esq.
Frank J. Notaro, Esq.

1800 One M & T Plaza
Ruffalo New York 14203

Buffalo, New York 14203 Telephone: (716) 856-4000 Telecopier: (716) 849-0349

Any party may designate in a writing to any other party any other address or telecopier number to which, and any other Person to whom or which, a copy of any such notice, request, instruction or other communication should be sent.

10.3 Arbitration. Except as provided for in Section 10.7, any controversy arising after the Closing out of or relating to this Agreement (including, without limitation, pursuant to Section 9.2), or relating to the breach hereof, or any cause of action arising from the acquisition of the Shares, shall be settled by arbitration conducted in Chicago, Illinois by a panel of three (3) impartial arbitrators selected by the attorneys for the respective parties, and otherwise conducted in accordance with the Commercial Arbitration Rules of the American Arbitration Association then in effect. The award rendered by the arbitrators shall be final and judgment upon the award rendered by the arbitrators may be entered upon it in any court having jurisdiction thereof. The arbitrators shall possess the powers to issue mandatory orders and restraining orders in connection with such arbitration. The expenses of the arbitration shall be borne by the losing party unless otherwise allocated by the arbitrators. The agreement to arbitrate shall be specifically enforceable under the prevailing arbitration law. During the continuance of any arbitration proceedings, the parties shall continue to perform their respective obligations under this Agreement.

with respect to the Corporation, any Subsidiary or any Seller shall mean (except as otherwise provided in the immediately succeeding sentence) actual knowledge, information or belief, as appropriate to the context of the statement in which the term is used, of any individual Seller or any officer, director or Representative of the Corporation, any Subsidiary or any Seller which is a partnership or a corporation, after having made due inquiry of the appropriate employees or records of the Corporation, any Subsidiary or Sellers with respect to the

matters which are relevant to the representation, warranty, covenant or agreement being made or given.

uith each other and to cooperate, both prior to and after the Closing, in issuing any press releases or otherwise making public statements with respect to the transactions contemplated by this Agreement, and, except as Buyer's counsel may advise in writing is otherwise required by law or the rules of the New York Stock Exchange as applicable to Buyer, no press release or other public statements shall be issued without the joint consent of Buyer and Sellers. Notwithstanding the foregoing, Buyer shall not, without the express written consent of Angus McBain or Walter Rose, use either of their names or the names of McBain, Rose Partners for any press release or public statement.

10.6 Choice of Law. This Agreement shall be construed, interpreted and the rights of the parties determined in accordance with the laws of Illinois, without regard to principles of conflicts of law, except that, with respect to matters of law concerning the internal corporate affairs of any corporate entity which is a party to or the subject of this Agreement, the law of the jurisdiction under which the respective entity was organized shall govern.

10.7 Equitable Remedies. Because a remedy at law for any breach of the provisions of Section 9.1 will be inadequate, in addition to all other remedies available to Buyer, Buyer (from and after the Closing) shall have the remedies of a restraining order, injunction or other equitable relief to enforce the provisions hereof. The parties hereto hereby agree that the issues in any action brought under Section 9.1 will be limited to claims under such section. All expenses, including, without limitation, attorneys fees and expenses, arising out of claims under Section 9.1, or under any other provision of this Agreement, shall be borne by the losing party to the fullest extent permitted by law.

10.8 Expenses. Except as otherwise provided in this Agreement, each of the Corporation, Sellers and Buyer shall pay its own legal, accounting and other expenses incident to this Agreement; provided, however, that the Corporation shall pay all of Sellers' reasonable legal and accounting fees and other miscellaneous expenses (including, without limitation, pursuant to Section 2.3(b)) incident to this Agreement and incurred prior to Closing (based on customary hourly rates). All such costs and expenses of Sellers which exceed the stated maximum amount shall be paid by Sellers.

10.9 Titles. The headings of the articles and sections of this Agreement are inserted for convenience of reference only and shall not affect the meaning or interpretation of this Agreement.

10.10 Waiver. No failure of Sellers or Buyer to require, and no delay by Sellers or Buyer in requiring, the other to comply with any provision of this Agreement shall constitute a waiver of the right to require such compliance. No failure of the Sellers or Buyer to exercise, and no delay by the Sellers or Buyer in exercising, any right or remedy under this Agreement shall constitute a waiver of such right or remedy. No waiver by the Sellers or Buyer of any right or remedy under this Agreement shall be effective unless made in writing. Any waiver by the Sellers or Buyer of any right or remedy under this Agreement shall be limited to the specific instance and shall not constitute a waiver of such right or remedy in the future.

10.11 Binding. This Agreement shall be binding upon the Corporation, Sellers and Buyer and upon each successor and assignee of the Corporation, Sellers and Buyer and shall inure to the benefit of, and be enforceable by, the Corporation, Sellers and Buyer and each successor and assignee of the Corporation, Sellers and Buyer; provided, however, that, except as provided for in the following sentence, neither the Corporation, Sellers nor Buyer shall assign any right or obligation arising pursuant to this Agreement without first obtaining the written consent of the other party. Buyer may assign all or a portion of its rights and obligations under this Agreement to one or more affiliates of Buyer, provided that Buyer shall remain liable hereunder notwithstanding any such assignment.

10.12 Entire Agreement. This Agreement contains the entire agreement between the Corporation, Sellers and Buyer with respect to the subject of this Agreement, and supersedes each course of conduct previously pursued, accepted or acquiesced in, and each written and oral agreement and representation previously made, by the Corporation, Sellers, McBain, Rose Partners or Buyer or their respective Representatives, with respect thereto, whether or not relied or acted upon including, without limitation, the letter from Donald N. Boyce to McBain Rose Partners dated February 22, 1994, but not including the Confidentiality Letter, dated January 13, 1994, between IDEX and the Corporation, which shall terminate only upon any Closing under this Agreement.

10.13 Modification. No course of performance or other conduct hereafter pursued, accepted or acquiesced in, and no oral agreement or representation made in the future, by the

Corporation, Sellers or Buyer, whether or not relied or acted upon, and no usage of trade, whether or not relied or acted upon, shall modify or terminate this Agreement, impair or otherwise affect any obligation of the Corporation, Sellers or Buyer pursuant to this Agreement or otherwise operate as a waiver of any such right or remedy. No modification of this Agreement or waiver of any such right or remedy shall be effective unless made in writing duly executed by the Corporation, Sellers' Agent and Buyer.

10.14 Counterparts. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original and all of which taken together shall constitute one and the same instrument. Any party may execute this Agreement by facsimile signature and the other parties shall be entitled to rely on such facsimile signature as evidence that this Agreement has been duly executed by such party. Any party executing this Agreement by facsimile signature shall immediately forward to the other party an original signature page by overnight mail.

10.15 Consent to Jurisdiction. Buyer, each Seller and the Corporation each hereby (a) consents to the jurisdiction of the United States District Court for the Northern District of Illinois or, if such court does not have jurisdiction over such matter, the applicable state court in the State of Delaware, and (b) irrevocably agrees that all actions or proceedings arising out of or relating to this Agreement shall be litigated in such court. Each of Sellers, Buyer and the Corporation accepts for himself or itself and in connection with his or its properties, generally and unconditionally, the exclusive jurisdiction and venue of the aforesaid courts and waives any defense of forum nonconveniens or any similar defense, and irrevocably agrees to be bound by any non-appealable judgment rendered thereby in connection with this Agreement. Notwithstanding anything contained in this Section 10.15, to the extent that Section 10.3 applies, it shall supersede and control the provisions of this Section 10.15.

IN WITNESS WHEREOF, the Corporation has caused to be executed by an officer of the Corporation, each of Sellers has executed and Buyer has caused to be executed by an officer of Buyer, on the day and year indicated at the beginning of this Agreement.

BUYER: HPI ACQUISITION CORP. By /s/ DONALD N. BOYCE Donald N. Boyce, President CORPORATION: HALE PRODUCTS, INC. By /s/ Peter J. Andrews -----President SELLERS: HFP PARTNERS, L.P. By /s/ Angus W. McBain, general partner By /s/ W. B. Rose Title HMTC PARTNERS, L.P. By /s/W. B. Rose , general partner By /s/ Angus W. McBain, general partner Title McBAIN PROFIT SHARING PLAN By /s/ Angus W. McBain, Trustee

, Plan Trustee

KLEINWORT BENSON EUROPEAN MEZZANINE FUND, LIMITED PARTNERSHIP

Kleinwort Benson
By (Guernsey) Limited, general partner

By Erik W. Linnes

Director, Kleinwort Benson Limited, Attorney in Fact for Kleinwort Benson (Guernsey) Limited Title

/s/ WALTER B. ROSE	/s/ H. ALFRED EBERHARDT
	H. Alfred Eberhardt
/s/ JOHN J. O'GRADY, JR.	/s/ DALE M. CLEMENTS
	Dale M. Clements
/s/ KENNETH F. HOFFMAN	/s/ PETER HOYNASH
Kenneth F. Hoffman	Peter Hoynash
/s/ MICHAEL R. MACOMSON, JR.	/s/ KURT H. C. BOTTCHER
Michael R. Macomson, Jr.	Kurt H. C. Bottcher
/s/ FRED W. BUCHLER	/s/ RICHARD E. TESKE
Fred W. Buchler	Richard E. Teske
/s/ JAMES P. LONIE	/s/ JERRY L. HOFFMAN
James P. Lonie	Jerry L. Hoffman
/s/ ROBERT J. LINSTER	/s/ DELBERT C. LUKENS, JR.
Robert J. Linster	Delbert C. Lukens, Jr.
/s/ JAMES J. MAHER, JR.	/s/ JOSEPH DAWIDZIUK
James J. Maher, Jr.	Joseph Dawidziuk

/s/ HARVEY M. WALTER	/s/ JOSEPH F. BRUCKNER
Harvey M. Walter	Joseph F. Bruckner
/s/ KALMAN BROITMAN	/s/ ALBERT P. ROMANO
Kalman Broitman	Albert P. Romano
/s/ JAMES A. KAYE	/s/ GARRY S. BARBER
James A. Kaye	Garry S. Barber
/s/ GARY HANDWERK	/s/ LEO E. KELLER
Gary Handwerk	Leo E. Keller
/s/ PETER J. ANDREWS	/s/ RITA L. BYRON
Peter J. Andrews	Rita L. Byron
/s/ LOUIS J. EVANGELISTA, JR.	/s/ DEREK HALL
Louis J. Evangelista, Jr.	Derek Hall
/S/ TIMOTHY LAMLEY	/S/ A. D. MOORE
Timothy Lamley /s/ CAROLYN BELL	A.D. Moore /s/ ANDREW ROE
Timothy Lamley /s/ CAROLYN BELL	A.D. Moore /s/ ANDREW ROE
Timothy Lamley /s/ CAROLYN BELL	A.D. Moore /s/ ANDREW ROE Andrew Roe /s/ COLIN POMFRET
Timothy Lamley /s/ CAROLYN BELL Carolyn Bell /s/ ROY PREEDY	A.D. Moore /s/ ANDREW ROE Andrew Roe /s/ COLIN POMFRET
Timothy Lamley /s/ CAROLYN BELL Carolyn Bell /s/ ROY PREEDY Roy Preedy /s/ PETER LEDGAR	A.D. Moore /s/ ANDREW ROE Andrew Roe /s/ COLIN POMFRET Colin Pomfret /s/ PHILLIP ROBERTS
Timothy Lamley /s/ CAROLYN BELL Carolyn Bell /s/ ROY PREEDY Roy Preedy /s/ PETER LEDGAR	A.D. Moore /s/ ANDREW ROE Andrew Roe /s/ COLIN POMFRET Colin Pomfret /s/ PHILLIP ROBERTS
Timothy Lamley /s/ CAROLYN BELL	A.D. Moore /s/ ANDREW ROE Andrew Roe /s/ COLIN POMFRET Colin Pomfret /s/ PHILLIP ROBERTS Phillip Roberts
Timothy Lamley /s/ CAROLYN BELL	A.D. Moore /s/ ANDREW ROE Andrew Roe /s/ COLIN POMFRET Colin Pomfret /s/ PHILLIP ROBERTS Phillip Roberts /s/ MICHAEL J. BRAYNE

- 67 -

/s/	DANIEL	. EL	WI(CK									
					 	 	 -	 	-	 	-	-	
Dani	iel Elw	vick											

/s/ GERALD BOMSTAD, JR.

Gerald Bomstad, Jr.

/s/ ALEX G. TYSHOVNYTSKY

Alex G. Tyshovnytsky

The undersigned acting individually acknowledges and agrees to be bound by the covenants set forth in Section 9.1 of this Agreement and is executing this Agreement in an individual capacity solely for purposes of acknowledging his obligations under such Section

.

Angus McBain

EXHIBIT A

STOCKHOLDERS

NAME	SHARES	CERTIF. NO(S).
HFP Partners, L.P.1	189,400	113
HMTC Partners, L.P.1	50,600	11
Walter B. Rose1	40,000	111
McBain Profit Sharing Plan1	40,000	112
H. Alfred Eberhardt	30,000	2
John J. O'Grady, Jr.	10,000	3
Dale M. Clements	7,000	5
Kenneth F. Hoffman	7,000	7
Peter Hoynash	4,000	9
Michael R. Macomson, Jr.	10,000	10
Kurt H.C. Bottcher	8,527	12, 26, 27, 43, 60, 76
Fred W. Buchler	2,284	13, 28, 44, 61, 77
Richard E. Teske	1,632	14, 29, 45, 62, 78
James P. Lonie	1,632	15, 30, 46, 63, 79
Jerry L. Hoffman	1,632	16, 31, 47, 64, 80
Robert J. Linster	1,305	17, 32, 48, 65, 81
Delbert C. Lukens, Jr.	1,105	18, 33, 66, 82
James J. Maher, Jr.	979	19, 34, 49, 67, 83
Joseph Dawidziuk	979	20, 35, 50, 68, 84
Harvey M. Walters	979	21, 36, 51, 69, 85
Joseph F. Bruckner	979	22, 37, 52, 70, 86
Kalman Broitman	979	23, 38, 53, 71, 87
Albert P. Romano	879	24, 39, 54, 72
James A. Kaye	979	25, 40, 55, 74, 92
Garry S. Barber	5,000	41, 115
Gary Handwerk	814	42
Leo E. Keller	514	57, 90

NAME 		SHARES	(CERTIF. NO(S).
Peter J. Andrews		20,605		59, 95
Rita L. Byron		800		75, 91
Louis J. Evangelista, Jr.		2,500		93
Kleinwort Benson European Mezzanine Fund Limited Partnership2		17,555		94
Derek Hall3		4,682		96
Timothy Lamley3		3,121		97
A.D. Moore3		468		98
Carolyn Bell3		156		99
Andrew Roe3		156		100
Roy Preedy3		156		101
Colin Pomfret3		156		102
Peter Ledgar3		156		103
Phillip Roberts3		156		104
Peter Jackson3		156		105
Michael Brayne3		4,682		106
Frank Mason3		156		107
Christopher Hill3		156		108
Daniel Elwick3		156		109
Gerald Bomstad, Jr.		1,000		110
Alex Tyshovnytsky		3,000		114
	TOTAL	479,171		
	WARRANTS (TCW)	100,000		
		579,171 		

Address is c/o McBain, Rose Partners, 355 South Grand Avenue, 42nd Floor, Los Angeles, CA 90071

- 3 -

- 2 Address is 20 Fenchurch Street, London EC3 3DB England, Attention: Eric Linness
- 3 Address is c/o Godiva Products Limited, Charles Street, Warwick CV34 5LR England

Addresses of all other stockholders is c/o Hale Products, Inc., 700 Spring Mill Avenue, Conshohocken, Pennsylvania 19428, Attention: Peter Andrews, John J. O'Grady, Jr. and Dale M. Clements.

EXHIBIT 7.5

LEGAL OPINION OF HODGSON, RUSS, ANDREWS, WOODS & GOODYEAR

_____, 1994

THE PARTIES LISTED ON SCHEDULE A TO THIS LETTER c/o HFP Partners, L.P. 355 South Grand Avenue, 42nd Floor Los Angeles, California 90071

Ladies and Gentlemen:

Re: Stock Purchase Agreement and Guaranty

We have acted as counsel to (1) HPI Acquisition Corporation (the "Purchaser") in connection with the execution and delivery to you by the Purchaser of a Stock Purchase Agreement dated as of May 5, 1994 (the "Agreement") and (2) IDEX Corporation (the "Guarantor") in connection with the execution and delivery to you by the Guarantor of a Guaranty Agreement dated May 5, 1994 (the "Guaranty"). This letter is being delivered to you pursuant to Section 7.5 of the Agreement.

The opinions set forth in this letter, whether or not qualified by the phrase "to our knowledge," are subject to the following qualifications:

(a) The opinions set forth in this letter are based solely upon (a) our review of, as submitted to us, (i) the Agreement, (ii) the Guaranty, (iii) the Certificate of Incorporation, By-laws and records of the corporate proceedings of each of the Purchaser and the Guarantor, (iv) the governmental certificates listed on Schedule B attached to this letter (individually a "Governmental Certificate") and (v) an Officer's Certificate, dated the date of this letter, executed by Wayne P. Sayatovic, the Vice President of the Purchaser and the Vice President of the Guarantor, (the "Officer's Certificate"), a copy of which is attached to this letter, (items (i) through (v) being collectively the "Reviewed Documents"), (b) the current recollection of those of our present attorneys who, since January 1, 1992, have devoted substantive attention to specific legal matters as to which we have been consulted by the Purchaser or the Guarantor (the "Attorney Information") and (c) such review of published sources of law as we have deemed necessary based

solely upon our review of the Reviewed Documents and the Attorney Information. Other than our review of the Reviewed Documents, we have made no inquiry or other investigation as to any factual matter (including, but not limited to, (a) any review of any of the files and other records of the Purchaser, the Guarantor or any court or other governmental authority or (b) any review of any of our files).

- We have assumed without any inquiry or other investigation (a) the legal capacity of each natural person, (b) the satisfaction by you of each legal requirement applicable to you to the extent necessary to make the Agreement and the Guaranty enforceable against you and to enable you to enforce the Agreement against the Purchaser and to enforce the Guaranty against the Guarantor, (c) there not occurring with respect to any transaction contemplated by the Agreement or the Guaranty any mutual mistake of fact or misunderstanding, fraud, duress or undue influence or any conduct that does not comply with any requirement of good faith, fair dealing or conscionability, (d) your acting in good faith and without written notice of any valid defense available to the Purchaser or the Guarantor against the enforcement of any right granted you by the Agreement or the Guaranty, (e) there being no agreement or understanding, whether written or oral, between you and the Purchaser or the Guarantor, and there being no usage of trade or course of dealing between you and the Purchaser or the Guarantor, that would define, supplement or qualify any provision of the Agreement or the Guaranty, (f) the accuracy on the date of this letter as well as on the date stated in either Governmental Certificate of each statement as to any factual matter contained in such Governmental Certificate and the accuracy on the date of this letter as well as on the date made of each statement as to any factual matter made in any other of the Reviewed Documents and (g) the genuineness of each signature on any of the Reviewed Documents, the completeness of each of the Reviewed Documents, the authenticity of each of the Reviewed Documents submitted to us as an original, the conformity to the original of each of the Reviewed Documents submitted to us as a copy and the authenticity of the original of the Reviewed Documents submitted to us as a copy.
- (c) We do not express any opinion concerning any law other than the law of New York, the federal law of the United States of America and the General Corporation Law of Delaware (the "DGCL").
- (d) Any opinion set forth in this letter concerning the enforceability of any document against any party (a) means that (i) such document constitutes a binding contract of such

party under the law of the State of New York and the federal law of the United . States, (ii) as to such party such document is not invalid in its entirety under such law because of a specific statutory prohibition or public policy and is not subject in its entirety to a contractual defense under such law and (iii) subject to the following sentence, a remedy substantially enforcing the effect of such document is available under such law if such party is in material breach of a representation, warranty or covenant thereunder but (b) does not mean that (i) any particular remedy is available under such law or (ii) every provision of such document (including, but not limited to, any provision giving a consent or waiver, granting any power of attorney or providing for indemnification or exculpation) will be upheld or enforced in any or each circumstance by a court applying such law. Furthermore, the enforceability of such document against such party may be limited or otherwise affected by (a) any bankruptcy, insolvency, reorganization, receivership, moratorium, marshaling, arrangement, assignment for benefit of creditors fraudulent transfer, fraudulent conveyance or other statute, rule, regulation or other law affecting the rights and remedies of creditors generally, (b) any general principle of equity, whether applied by a court of law or equity, including, but not limited to, any principle (i) governing the availability of specific performance, injunctive relief or any other equitable remedy that is subject to the discretion of a court, (ii) affording any equitable defense, (iii) requiring good faith, fair dealing and reasonableness in the performance and enforcement of a contract by a party seeking the enforcement of such contract, (iv) requiring consideration of the materiality of a breach of a contract by a party against whom or which the enforcement of such contract is sought and consideration of the consequences of such breach to a party seeking such enforcement, (v) requiring consideration at the time the enforcement of $\bar{\mathbf{a}}$ contract is attempted of the impracticality or impossibility of the performance of such contract or (vi) affording any defense to the enforcement of a contract based upon the unconscionability of the conduct after such contract has been entered into of a party seeking such enforcement) and (c) any generally applicable rule of law that (i) limits or otherwise affects the enforcement of any provision of a contract that purports to require waiver of any obligation of good faith, fair dealing, diligence and reasonableness, (ii) provides that a forum selection clause in a contract is not necessarily binding on a court, (iii) limits the availability of a remedy under certain circumstances where another remedy has been elected, (iv) limits the right of a creditor to use force or cause a breach of the peace in enforcing any right, (v) limits the availability of a provision releasing, exculpating or exempting a party from, or requiring indemnification of a party for, liability for its own action or

inaction, (vi) where less than all of a contract may be unenforceable, limits the enforceability of the balance of such contract to circumstances in which the unenforceable portion is not an essential part of the agreed upon exchange under such contract, (vii) governs and affords judicial discretion regarding the determination of damages and entitlement to attorneys' fees and other costs and (viii) permits a party that has materially failed to render or offer any performance required by a contract to cure such failure unless permitting such cure would unreasonably hinder the aggrieved party from making substitute arrangements for such performance or in the circumstances it was important to the aggrieved party that such performance occur by the date stated in such contract.

- (e) Any opinion set forth in this letter (a) deals only with the specific legal issue or issues it explicitly addresses, (b) does not address any other matter and (c) does not address any legal issue arising under (i) any statute, rule, regulation or other law relating to any securities, commodity or other future, pension, employee benefit, antitrust, unfair competition, fraudulent transfer, fraudulent conveyance, environmental, land use, subdivision, tax, copyright, racketeering, health, safety, labor, forfeiture or criminal matter or fiduciary requirement or (ii) any statute, ordinance, rule, regulation or other law of any municipality.
- (f) We advise you that one of our general partners, Richard E. Heath, is a member of the board of directors of the Guarantor.

Subject to the qualifications set forth in this letter, it is our opinion that:

- 1. The Purchaser (a) is a corporation duly incorporated, validly existing and in good standing under the law of Delaware and (b) has the corporate power to execute, deliver to you and perform the Agreement.
- 2. The execution, delivery to you and performance by the Purchaser of the Agreement have been duly authorized by all necessary corporate action of the Purchaser.
- 3. The Agreement (a) has been duly executed and delivered to you by the Purchaser and (b) if it were governed by the law of New York, would be enforceable against the Purchaser.
- 4. The execution and delivery to you by the Purchaser of the Agreement do not (a) violate the Certificate of Incorporation or By-laws of the Purchaser or (b) to our

Page 5

knowledge, violate any judgment, decree, or order of any court or other governmental authority applicable to the Purchaser, (c) assuming the non-existence of any judgment, decree or order that would be violated by such execution and delivery, (i) violate the DGCL or any United States federal statute, rule or regulation applicable to the Purchaser or (ii) other than any consent or approval required pursuant to the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, require any consent, approval or authorization of, or filing with, any Delaware court or other Delaware governmental authority pursuant to the DGCL or any United States federal court or other United States federal governmental authority.

- 5. The Guarantor (a) is a corporation duly incorporated, validly existing and in good standing under the law of Delaware and (b) has the corporate power to execute, deliver to you and perform the Guaranty.
- 6. The execution, delivery to you and performance by the Guarantor of the Guaranty have been duly authorized by all necessary corporate action of the Guarantor.
- 7. The Guaranty (a) has been duly executed and delivered to you by the Guarantor and (b) if it were governed by the law of New York, would be enforceable against the Guarantor.
- 8. The execution and delivery to you by the Guarantor of the Guaranty do not (a) violate the Certificate of Incorporation or By-laws of the Guarantor, (b) to our knowledge, violate any judgment, decree or order of any court or other governmental authority applicable to the Guarantor, (c) assuming the non-existence of any judgment, decree or order that would be violated by such execution and delivery, (i) violate the DGCL or any United States federal statute, rule or regulation applicable to the Guarantor or (ii) other than any consent or approval required pursuant to the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, require any consent, approval or authorization of any Delaware court or other Delaware governmental authority pursuant to the DGCL or any United States federal court or other United States federal governmental authority.
- 9. To our knowledge there is no pending or overtly threatened action, suit, proceeding inquiry or investigation by or before any Governmental Authority (as defined in the Agreement) that is against or involves the Purchaser or the Guarantor and that questions or challenges the validity of the Agreement or any of the transactions contemplated by the Agreement or otherwise seeks to prevent or have the effect of

preventing the consummation of the transactions contemplated thereby.

10. To our knowledge, no unsatisfied judgment, order, writ, injunction, decree, directive or assessment or other command of any Governmental Authority (as defined in the Agreement) that questions or challenges the validity of the Agreement or any of the transactions contemplated by the Agreement or would otherwise have the effect of preventing the consummation of the transactions contemplated thereby has been entered against or served upon the Purchaser or the Guarantor.

This letter is intended solely for your benefit and, without our express written consent, may not be relied upon, referred to or otherwise used by any other person or other than in connection with the Agreement and the Guaranty.

Very truly yours,

HODGSON, RUSS, ANDREWS, WOODS & GOODYEAR

Ву

Richard E. Heath

SCHEDULE B

GOVERNMENT CERTIFICATES

	Certificate	State	Date
HPI Acquisition Corporation	Good Standing	Delaware	, 1994
IDEX Corporation	Good Standing	Delaware	, 1994

EXHIBIT 8.6(A)

LEGAL OPINION OF LATHAM & WATKINS

[OPINION OF LATHAM & WATKINS]

May ___, 1994

HPI Acquisition Corp. c/o IDEX Corporation 630 Dundee Road, Suite 400 Northbrook, Illinios 60065

Re: Sale of Stock of Hale Products, Inc. Pursuant to Stock

Purchase Agreement dated May ___, 1994

Ladies and Gentlemen:

We have acted as special counsel to Hale Products, Inc., a Delaware corporation (the "Company"), HFP Partners, L.P. and HMTC Partners, L.P., each of which is a Delaware limited partnership (each a "Partnership" and collectively, the "Partnerships"), and (except as noted below) each other Seller (as defined below), in connection with the sale of the outstanding capital stock of the Company to HPI Acquisition Corp., a Delaware corporation ("Buyer") pursuant to that certain Stock Purchase Agreement dated as of May ___, 1994 (the "Purchase Agreement") by and among the Company, the Partnerships, the persons and entities listed on Schedule A thereto (together with the Partnerships, the "Sellers") and Buyer. We have not, however, acted as counsel to Kleinwort Benson [need to see stock ledger for exact name] or TCW Special Placements Fund II in connection with the transactions referred to herein, and do not render any opinion on their behalf. Accordingly, as used herein, the term "Sellers" shall not include either of them. The opinions expressed below are rendered to you in accordance with Section 8.6 of the Purchase Agreement. Each of the capitalized terms used and not otherwise defined herein shall have the meanings specified in the Purchase Agreement.

- (a) The Purchase Agreement;
- (b) The Certificate of Incorporation, Bylaws and board resolutions relevant to the subject transactions of the Company;
- (c) The Certificate of Limited Partnership and the Limited Partnership Agreement of each of the Partnerships (the "Partnership Agreements");
 - (d) [The McBain Profit Sharing Plan] (the "Plan"); and
- (e) The written contracts or agreements to which the Partnerships and the Company are parties and which are identified to us by a general partner of each Partnership or an officer of the Company (as applicable) as material to such Partnership or the Company (the "Material Adreements").

As to certain factual matters, we have examined and with your permission have relied upon certificates and advice of public officials, certificates of the Sellers, a general partner of each Partnership and an officer of the Company, and such other documents as we have deemed necessary or appropriate in connection with this opinion. We have not reviewed the files or records of the Company or any Seller, any court or any state, county or other governmental body.

In our examination, we have assumed the legal capacity of each natural person, the genuineness of all signatures, the authenticity of all documents submitted to us as originals, and the conformity to authentic original documents of all documents submitted to us as copies.

We have investigated such questions of law for the purpose of rendering this opinion as we have deemed necessary. We are opining herein as to the effect on the subject transaction only of the internal laws of the State of Illinois and the federal laws of the United States of America and the General Corporation Law and the Revised Uniform Limited Partnership Act of the State of Delaware and assume no reponsibility as to the applicability to the transaction, or the effect thereon, of the laws of any other jurisdiction or any other laws of the State of Delaware.

Whenever a statement herein is qualified by "to the best of our knowledge" or a similar phrase, it is intended to indicate that those attorneys in this firm who have rendered legal services in connection with the Purchase Agreement do not have current actual knowledge of the inaccuracy of such statement. However, except as otherwise expressly indicated, we have not undertaken any independent investigation to determine that accuracy of any such statement, and no inference that we have any knowledge of any matters pertaining to such statement should be drawn from our representation of the Company and the Sellers.

On the basis of the foregoing, and in reliance thereon, and subject to the limitations, qualifications and exceptions hereinafter set forth, we are of the opinion that, as of the date hereof:

- 1. Each Partnership has been duly formed and is validly existing and in good standing under the laws of the State of Delaware, with partnership power and authority to conduct its business as currently conducted, to own or lease its assets and to enter into the Purchase Agreement and perform its obligations thereunder.
- 2. The Company has been duly incorporated and is validly existing and in good standing under the laws of the State of Delaware, with corporate power and authority to conduct its business as currently conducted, to own or lease its assets and to enter into the Purchase Agreement and perform its obligations thereunder. Based solely on certificates from public officials, we confirm that the Company is qualified to do business in the States of and
- 3. Each of Hale Products Technology, Inc. and Hale Products Finance, Inc. has been duly incorporated and is validly existing and in good standing under the laws of the State of Delaware, with corporate power and authority to conduct its business as currently conducted and to own or lease its assets. [Based solely on certificates from public officials, we confirm that each of them is qualified to do business in the States of and .] [Other subsidiary opinions must come from local counsel.]
- 4. The execution, delivery and performance of the Purchase Agreement has been duly authorized by all necessary corporate or partnership action of the Company and the Partnerships, as applicable, and the Purchase Agreement has been duly executed and delivered by the Company and each Seller.
- 5. The Purchase Agreement constitutes a legally valid and binding obligation of the Company and each Seller, enforceable against each of them in accordance with its terms.
- 6. Neither the execution and delivery of the Purchase Agreement by either Partnership nor the consummation of the transactions contemplated thereby will (i) violate its Partnership Agreement, (ii) breach or cause a default under any term or provision of any Material Agreement of such Partnership or (iii) to the best of our knowledge, violate any judgment, decree, injunction, writ or order applicable to the Partnership. Neither the execution and delivery of the Purchase Agreement by the Company nor the consummation of the transactions contemplated thereby will (i) violate the Company's Certificate of Incorporation, (ii) except as set forth in the Purchase Agreement, the exhibits thereto or the disclosure schedules attached thereto, breach or cause a default under any term or provision of any Material Agreement of the Company or (iii) to the best of our knowledge, violate any judgment, decree, injunction, writ or order applicable to the Company.

- 7. No authorization, consent, order, permit or approval of, or filing with, any governmental authority, or, to the best of our knowledge, any other person, is required for the execution and delivery of the Purchase Agreement by the Company or the Sellers or the consummation by the Company or the Sellers of the transactions contemplated on their part thereby, except as set forth in the Purchase Agreement, the exhibits thereto or the disclosure schedules attached thereto.
- 8. To the best of our knowledge, the execution and performance of the Purchase Agreement by the Company and the Sellers will not (i) violate or result in a failure to comply with any statute, law, ordinance, regulation or rule of any federal, state or local government or any other governmental department or agency, except where such violation would not have a material adverse effect on the Company's financial condition or Business or its or the Sellers' ability to perform their respective obligations under the Purchase Agreement, or (ii) result in the imposition of any Encumbrance on the Shares or any assets of the Company or any Subsidiary.
- 9. To the best of our knowledge, except as set for thin the Purchase Agreement, the exhibits thereto or the disclosure schedules attached thereto, no action, order, injunction, judgment, decree, claim, suit, litigation or proceeding is pending or threatened against the Company or any Seller by or before any governmental authority which would materially adversely affect the Company's or the Sellers' ability to perform their respective obligations under the Purchase Agreement or the transactions contemplated thereby.
- The authorized capital stock of the Company consists 10. of shares of Common Stock (the "Common Stock"), of which issued and are outstanding as of the date hereof. shares are All of such outstanding shares of Common Stock are duly and validly issued, fully paid and nonassessable. Except as set forth in the Purchase Agreement, the exhibits thereto or the disclosure schedules attached thereto, to our knowledge, there are no outstanding options, warrants, convertible securities or other rights issued by the Company exercisable or exchangeable for or convertible into capital stock of the Company. Assuming you have no notice of any adverse claim with respect to the shares of Common Stock sold to you by the Sellers, and that such shares have been purchased and paid for by you and endorsed and delivered to you in accordance with the Purchase Agreement, upon Closing, the shares so delivered to you will have been acquired by you free of adverse claims arising in repsect of the Sellers.

Our opinion[s] in paragraph 2 [and 3] above with respect to the Company's [and the Finance and Technology's] qualification to do business as [a] foreign corporation is [are] based solely on certificates of public officials in the States of and .

- (a) Such opinions are subject to the effect of bankruptcy, insolvency, reorganization, moratorium, and other similar laws relating to or affecting the rights of creditors generally;
- (b) Such opinions are subject to the effect of general principles of equity, including without limitation concepts or materiality, reasonableness, good faith and fair dealing and the possible unavailability of specific performance, injunctive relief or other equitable remedies, whether considered in a proceeding in equity or at law;
- (c) Certain rights, remedies and waivers contained in the Purchase Agreement may be limited or rendered ineffective by applicable laws or judicial decisions governing such provisions, but such laws or judicial decisions do not render the Purchase Agreement invalid or unenforceable as a whole;
- (d) Under certain circumstances, provisions providing for the indemnification of or contribution to a party with respect to a liability may be unenforceable where such indemnification or contribution is based on such party's own wrongful or negligent acts or is otherwise contrary to public policy; and
- (e) We express no opinion with respect to the enforceability of Section 9.1 of the Purchase Agreement.

In rendering the opinions expressed in paragraph 6 and 7 above insofar as they require interpretation of the Material Agreements of the Company and the Partnerships (i) we have assumed with your permission that all courts of competent jurisdiction would enforce such agreements as written but would apply the internal laws of the State of Illinois without giving effect to any choice of law provisions contained therein or any choice of law principles which would result in application of the internal laws of any other state and (ii) to the extent that any questions of legality or legal construction have arisen in connection with our review, we have applied the laws of the State of Illinois in resolving such questions. We advise you that the Material Agreements are governed by other laws, that such laws may vary substantially from the law assumed to govern for purposes of this opinion, and that this opinion may not be relied upon as to whether or not a breach or default would occur under the law actually governing such Material Agreements.

Our opinions in paragraph 7 above as to authorizations, consents, orders, permits or approvals of, or filings with, any governmental authority are based upon a review of those authorizations, consents, orders, permits, approvals or filings which, in our experience, are normally applicable to transactions of the type contemplated by the Purchase Agreement. However, we express no opinion regarding authorizations, consents, approvals or filings relating to or which may be required under state securities laws, antitrust laws or the law of unfair competition. In addition, insofar as our opinions in paragraph 7 relate to authorizations, consents or approvals of, or filings with, any person other than governmental authorities, we

have based such opinions solely on our review of those contracts or agreements identified to us as Material Agreements of the Company or the Partnerships in certificates of an officer of the Company and a general partner of the Partnerships, respectively, which certificates are attached as exhibits hereto. Similarly, our opinions in paragraph 8 above as to the imposition of Encumbrances (insofar as such Encumbrances may arise based on contracts, agreements or relationships with persons other than governmental authorities) are based solely on such a review. We explicitly note, in regard to all of the foregoing opinions, that we have not identified or undertaken to review any contracts or agreements (other than the Purchase Agreement) to which any Seller other than the Partnerships is a party or by which it is bound. Moreover, we have assumed for all purposes herein that each Seller who is a natural person has the legal right to alienate marital property without the consent of his or her spouse under the laws of the jurisdiction in which such Seller resides.

Our opinions in paragraph 8 above as to compliance with certain statute, laws, ordinances, regulations or rules are based upon a review of those statutes, laws, ordinances, regulations or rules which, in our experience, are normally applicable to transactions of the type contemplated by the Purchase Agreement, and statutes, laws, ordinances, regulations or rules applicable to corporations or partnerships conducting operating businesses. However, we express no opinion regarding the effect of state securities laws, antitrust laws or the law of unfair competition.

We have assumed, for purposes of our opinions, the truth of your representations and warranties contained in Section 5.7 of the Purchase Agreement.

To the extent that the obligations of the Company or the Sellers may be dependent upon such matters, we assume for purposes of this opinion that the Buyer is duly organized and incorporated, validly existing and in good standing under the laws of its jurisdiction of incorporation, that it is duly qualified to engage in the transactions covered by this opinion, that the Purchase Agreement has been duly authorized, executed and delivered by the Buyer and constitutes the Buyer's legal and valid and binding obligation, enforceable in accordance with its terms, and that the Buyer has the requisite corporate and legal power and authority to carry on its business as now being conducted and to perform its obligations under the Purchase Agreement. We express no opinion as to compliance by the Buyer with any state or federal laws or regulations applicable to the subject transactions because of the nature of the Buyer's business or the business of any of its affiliates.

This opinion is rendered solely for the benefit of the Buyer in connection with the transactions described above and may not be relied upon by the Buyer or any other person or entity for any other purpose. This opinion is not to be used, circulated, quoted or referred to without our prior written consent.

Very truly yours,

FORMS OF LOCAL COUNSEL OPINIONS

- 1. The Subsidiary has been duly incorporated and is validly existing and in good standing under the laws of the State of , with corporate power and authority to conduct its business as currently conducted and to own or lease its assets. [Based solely on certificates from public officials, we confirm that the Subsidiary is qualififed to do business in the State(s) of and .]
- 2. The authorized capital stock of the Subsidiary consists of shares of common stock, of which shares are issued and shares are outstanding as of the date hereof. All of such outstanding shares are duly and validly issued, fully paid and nonassessable. Based solely on our review of the stock ledger of the Subsidiary, we confirm that, to the best of our knowledge, all of such outstanding shares are owned of recored by [Hale Products, Inc.] Except as set forth in the Purchase Agreement, the exhibits thereto or the disclosure schedules attached thereto, to the best of our knowledge, there are no outstanding options, warrants, convertible securities or other rights issued by the Subsidiary exercisable or exchangeable for or convertible into capital stock of the Subsidiary.

EXHIBIT 8.10

RESTRICTIVE COVENANTS AGREEMENTS

- Form for Peter J. Andrews.
 Form for other U.S. management executives.
 Form for Michael J. Brayne.
 Form for Kurt H.C. Bottcher.
 Form for HFP and HMTC.
- 1. 2. 3. 4. 5.

RESTRICTIVE COVENANTS AGREEMENT

THIS RESTRICTIVE COVENANTS AGREEMENT ("Agreement"), dated as
of May, 1994, is by and between HPI ACQUISITION CORP., a Delaware
corporation with its principal place of business at 630 Dundee Road, Suite 400,
Northbrook Illinois 60054 (together with its affiliates, "Buyer"); HALE
PRODUCTS, INC., a Delaware corporation with its principal place of business at
700 Spring Mill Avenue, Conshohocken, Pennsylvania 19428; and PETER J. ANDREWS,
an individual residing at
("Covenantor").

RECTTALS:

WHEREAS, pursuant to a Stock Purchase Agreement, dated as of the date of this Agreement, by and among Buyer, Covenantor, HFP Partners, L.P., HMTC Partners, L.P. and others (the "Stock Purchase Agreement"), Buyer is purchasing all of the issued and outstanding shares of common stock of Hale Products, Inc., a Delaware corporation (together with its Subsidiaries (as defined in the Stock Purchase Agreement), the "Corporation"); and

 $\label{eq:WHEREAS} \mbox{WHEREAS, Covenantor is employed as an Executive Officer of the Corporation; and}$

WHEREAS, in addition to serving as an Executive Officer of the Corporation, Covenantor is a stockholder of the Corporation and, pursuant to the transactions contemplated by the Stock Purchase Agreement, is receiving valuable consideration for his stock in the Corporation; and

WHEREAS, on and after the date of this Agreement, the Corporation will continue to conduct its business consisting of the manufacture, distribution, sales and service of fire fighting pumps, rescue equipment and tools and other related products, and other related activities (the "Business"); and

WHEREAS, during the period of his employment with the Corporation, Covenantor has had access to, and has gained knowledge with respect to, the Business; and

WHEREAS, in order to protect Buyer and the Business, and Buyer's investment in the Business, it is necessary that Covenantor refrain from direct or indirect competition in the Business to be carried on by the Corporation.

NOW, THEREFORE, in consideration of the money to be paid to Covenantor pursuant to the Stock Purchase Agreement, the mutual covenants and agreements hereinafter set forth, and other good and valuable consideration, the receipt of which is hereby

acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

Covenant. For a period of two (2) years following Covenantor's ceasing to be employed by the Corporation regardless of reason, Covenantor agrees that he shall not, directly or indirectly, for his own account or as agent, employee, officer, director, trustee, consultant or partner, or as a stockholder or equity owner of any corporation or any other entity (except for ownership of securities constituting less than five percent (5%) of any class of securities of a public company), or member of any firm or otherwise, (a) engage or attempt to engage, in the Restricted Territory (as hereinafter defined), in the Business or any other business activity which is the same as, substantially similar to or directly or indirectly competitive with the business conducted by the Corporation at the date of Covenantor's ceasing to be employed by the Corporation, (b) employ or solicit the employment of any person who was employed by the Corporation at the date of Covenantor's ceasing to be employed by the Corporation or within six months prior to such date, (c) canvass or solicit business in competition with the business conducted by the Corporation at the date of Covenantor's ceasing to be employed by the Corporation from any person or entity who during the last six months of Covenantor's employment shall have been a customer of the Corporation, or from any person or entity which there is reason to believe might in the future become a customer of the Corporation as a result of marketing efforts, contacts or other facts and circumstances of which Covenantor is aware during the period of Covenantor's employment with the Corporation, (d) willfully dissuade or discourage any person or entity from using, employing or conducting business with the Corporation, or (e) disrupt or interfere with, or seek to disrupt or interfere with, the business or contractual relationship between the Corporation and any supplier who during the last six months of Covenantor's employment with the Corporation shall have supplied components, materials or services to the Corporation. For purposes of this Agreement, the term Restricted Territory shall mean the United States, the United Kingdom of Great Britain, Northern Ireland and Continental Europe.

Confidentiality Agreement.

(a) Covenantor has had, and will continue to have, access to, and has gained, and will continue to gain, knowledge with respect to, all of the Business, the Corporation's trade secrets, financial results and information, processes and techniques, methods of doing business and information concerning customers and suppliers, and other valuable and confidential information, which is not generally known to the public (the "Confidential Information"). The parties acknowledge that unauthorized disclosure or misuse of the Confidential Information

following Covenantor's ceasing to be employed by the Corporation may cause irreparable damage to Buyer, the Corporation and the Business. The parties also agree that covenants by Covenantor not to make unauthorized disclosures of the Confidential Information are essential to the growth and stability of the Business. Accordingly, Covenantor agrees that, following the Covenantor's ceasing to be employed by the Corporation regardless of reason, he shall not use or disclose any Confidential Information obtained by him in the course of his employment with the Corporation.

- (b) Nothing in Section 2(a) shall prevent Covenantor from disclosing or using any such information as may be reasonably necessary (i) in connection with any lawsuit or arbitration arising out of this Agreement, (ii) in connection with any judicial or administrative filing, investigation or proceeding or otherwise with or by a governmental agency or (iii) as otherwise may be required by law.
- 3. Reasonableness of Limitations. Covenantor agrees that the territorial, time and other limitations contained in this Agreement are reasonable and properly required for the adequate protection of the business and affairs of Buyer and the Corporation, and in the event that any one or more of such territorial, time or other limitations is found to be unreasonable by a court of competent jurisdiction, Covenantor agrees to submit to the reduction of said territorial, time or other limitations to such an area, period or otherwise as the court may determine to be reasonable. In the event that any limitation under this Agreement is found to be unreasonable or otherwise invalid in any jurisdiction, in whole or in part, Covenantor acknowledges and agrees that such limitation shall remain and be valid in all other jurisdictions.
- 4. Remedies. Covenantor acknowledges that Buyer and the Corporation may suffer damages incapable of ascertainment in the event the provisions of this Agreement are breached and that Buyer and the Corporation may be irreparably damaged in the event that the provisions of this Agreement are not enforced, and Covenantor represents and warrants that his attorneys have thoroughly and completely reviewed this Agreement with him and that Covenantor fully understands the contents hereof. Therefore, should any dispute arise with respect to the breach or threatened breach of this Agreement, Covenantor agrees and consents that, in addition to any and all other remedies available to Buyer and the Corporation, an injunction or restraining order or other equitable relief may be issued or ordered by a court of competent jurisdiction restraining any breach or threatened breach of this Agreement.

- 4 -

- 5. Jurisdiction. The parties hereto agree that the exclusive venue and place of trial for the resolution of any disputes arising in connection with the interpretation or enforcement of this Agreement shall be the United States District Court for the Northern District of Illinois.
- 6. Miscellaneous. This Agreement shall be governed by and shall be construed in accordance with the internal laws of the State of Pennsylvania without regard to principles of conflicts of laws. All headings and subtitles contained herein are for convenience of reference only and are not of substantive effect. This Agreement constitutes the entire agreement among the parties hereto with respect to the subject matter hereof and supersedes all prior negotiations and understandings, written or oral, with respect to the subject matter hereof. There are no oral agreements in connection with this Agreement. Neither this Agreement nor any provision hereof may be waived, terminated, modified or amended orally or by any course of conduct but only by an agreement in writing duly executed by all of the parties hereto. If any article, section, portion, subsection or subportion of this Agreement shall be determined to be unenforceable or invalid, then such article, section, portion, subsection or subportion shall be modified to the minimum extent required so as to render the article, section, portion, subsection or subportion valid to the maximum extent under applicable law, and any such determination shall not affect the remainder of this Agreement, which shall be and shall remain binding and effective as against all parties hereto.

 $\,$ IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement as of the date first above written.

HPI ACQUISITION CORP.	
Ву	Title
HALE PRODUCTS, INC.	11110
Ву	Title
Peter J. Andrews	

RESTRICTIVE COVENANTS AGREEMENT

THIS RESTRICTIVE COVENANTS AGREEMENT ("Agreement"), o	dated a	as
of May, 1994, is by and between HPI ACQUISITION CORP., a Delaware		
corporation with its principal place of business at 630 Dundee Road, S	Suite 4	400,
Northbrook Illinois 60054 (together with its affiliates, "Buyer"); HAL	_E	
PRODUCTS, INC., a Delaware corporation with its principal place of bus	siness	at
700 Spring Mill Avenue, Conshohocken, Pennsylvania 19428; and JOHN J.	O'GRAD	DY,
JR., an individual residing at		
("Covenantor").		

RECTTALS:

WHEREAS, pursuant to a Stock Purchase Agreement, dated as of the date of this Agreement, by and among Buyer, Covenantor, HFP Partners, L.P., HMTC Partners, L.P. and others (the "Stock Purchase Agreement"), Buyer is purchasing all of the issued and outstanding shares of common stock of Hale Products, Inc., a Delaware corporation (together with its Subsidiaries (as defined in the Stock Purchase Agreement), the "Corporation"); and

 $\,$ WHEREAS, Covenantor is employed as an Executive Officer of the Corporation; and

WHEREAS, in addition to serving as an Executive Officer of the Corporation, Covenantor is a stockholder of the Corporation and, pursuant to the transactions contemplated by the Stock Purchase Agreement, is receiving valuable consideration for his stock in the Corporation; and

WHEREAS, on and after the date of this Agreement, the Corporation will continue to conduct its business consisting of the manufacture, distribution, sales and service of fire fighting pumps, rescue equipment and tools and other related products, and other related activities (the "Business"); and

WHEREAS, during the period of his employment with the Corporation, Covenantor has had access to, and has gained knowledge with respect to, the Business; and

WHEREAS, in order to protect Buyer and the Business, and Buyer's investment in the Business, it is necessary that Covenantor refrain from direct or indirect competition in the Business to be carried on by the Corporation.

NOW, THEREFORE, in consideration of the money to be paid to Covenantor pursuant to the Stock Purchase Agreement, the mutual covenants and agreements hereinafter set forth, and other good and valuable consideration, the receipt of which is hereby

acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

Covenant. For a period of one (1) year following Covenantor's ceasing to be employed by the Corporation regardless of reason, Covenantor agrees that he shall not, directly or indirectly, for his own account or as agent, employee, officer, director, trustee, consultant or partner, or as a stockholder or equity owner of any corporation or any other entity (except for ownership of securities constituting less than five percent (5%) of any class of securities of a public company), or member of any firm or otherwise, (a) engage or attempt to engage, in the Restricted Territory (as hereinafter defined), in the Business or any other business activity which is the same as, substantially similar to or directly or indirectly competitive with the business conducted by the Corporation at the date of Covenantor's ceasing to be employed by the Corporation, (b) employ or solicit the employment of any person who was employed by the Corporation at the date of Covenantor's ceasing to be employed by the Corporation or within six months prior to such date, (c) canvass or solicit business in competition with the business conducted by the Corporation at the date of Covenantor's ceasing to be employed by the Corporation from any person or entity who during the last six months of Covenantor's employment shall have been a customer of the Corporation, or from any person or entity which there is reason to believe might in the future become a customer of the Corporation as a result of marketing efforts, contacts or other facts and circumstances of which Covenantor is aware during the period of Covenantor's employment with the Corporation, (d) willfully dissuade or discourage any person or entity from using, employing or conducting business with the Corporation, or (e) disrupt or interfere with, or seek to disrupt or interfere with, the business or contractual relationship between the Corporation and any supplier who during the last six months of Covenantor's employment with the Corporation shall have supplied components, materials or services to the Corporation. For purposes of this Agreement, the term Restricted Territory shall mean the United States, the United Kingdom of Great Britain, Northern Ireland and Continental Europe.

Confidentiality Agreement.

(a) Covenantor has had, and will continue to have, access to, and has gained, and will continue to gain, knowledge with respect to, all of the Business, the Corporation's trade secrets, financial results and information, processes and techniques, methods of doing business and information concerning customers and suppliers, and other valuable and confidential information, which is not generally known to the public (the "Confidential Information"). The parties acknowledge that unauthorized disclosure or misuse of the Confidential Information

following Covenantor's ceasing to be employed by the Corporation may cause irreparable damage to Buyer, the Corporation and the Business. The parties also agree that covenants by Covenantor not to make unauthorized disclosures of the Confidential Information are essential to the growth and stability of the Business. Accordingly, Covenantor agrees that, following the Covenantor's ceasing to be employed by the Corporation regardless of reason, he shall not use or disclose any Confidential Information obtained by him in the course of his employment with the Corporation.

- (b) Nothing in Section 2(a) shall prevent Covenantor from disclosing or using any such information as may be reasonably necessary (i) in connection with any lawsuit or arbitration arising out of this Agreement, (ii) in connection with any judicial or administrative filing, investigation or proceeding or otherwise with or by a governmental agency or (iii) as otherwise may be required by law.
- 3. Reasonableness of Limitations. Covenantor agrees that the territorial, time and other limitations contained in this Agreement are reasonable and properly required for the adequate protection of the business and affairs of Buyer and the Corporation, and in the event that any one or more of such territorial, time or other limitations is found to be unreasonable by a court of competent jurisdiction, Covenantor agrees to submit to the reduction of said territorial, time or other limitations to such an area, period or otherwise as the court may determine to be reasonable. In the event that any limitation under this Agreement is found to be unreasonable or otherwise invalid in any jurisdiction, in whole or in part, Covenantor acknowledges and agrees that such limitation shall remain and be valid in all other jurisdictions.
- 4. Remedies. Covenantor acknowledges that Buyer and the Corporation may suffer damages incapable of ascertainment in the event the provisions of this Agreement are breached and that Buyer and the Corporation may be irreparably damaged in the event that the provisions of this Agreement are not enforced, and Covenantor represents and warrants that his attorneys have thoroughly and completely reviewed this Agreement with him and that Covenantor fully understands the contents hereof. Therefore, should any dispute arise with respect to the breach or threatened breach of this Agreement, Covenantor agrees and consents that, in addition to any and all other remedies available to Buyer and the Corporation, an injunction or restraining order or other equitable relief may be issued or ordered by a court of competent jurisdiction restraining any breach or threatened breach of this Agreement.

- 5. Jurisdiction. The parties hereto agree that the exclusive venue and place of trial for the resolution of any disputes arising in connection with the interpretation or enforcement of this Agreement shall be the United States District Court for the Northern District of Illinois.
- and shall be construed in accordance with the internal laws of the State of Pennsylvania without regard to principles of conflicts of laws. All headings and subtitles contained herein are for convenience of reference only and are not of substantive effect. This Agreement constitutes the entire agreement among the parties hereto with respect to the subject matter hereof and supersedes all prior negotiations and understandings, written or oral, with respect to the subject matter hereof. There are no oral agreements in connection with this Agreement. Neither this Agreement nor any provision hereof may be waived, terminated, modified or amended orally or by any course of conduct but only by an agreement in writing duly executed by all of the parties hereto. If any article, section, portion, subsection or subportion of this Agreement shall be determined to be unenforceable or invalid, then such article, section, portion, subsection or subportion of subsection or subportion valid to the maximum extent required so as to render the article, section, portion, subsection or subportion valid to the maximum extent under applicable law, and any such determination shall not affect the remainder of this Agreement, which shall be and shall remain binding and effective as against all parties hereto.

 $\,$ IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement as of the date first above written.

HPI ACQUISITION CORP.	
Ву	Title
HALE PRODUCTS, INC.	
Ву	Title
John J. O'Grady, Jr.	

RESTRICTIVE COVENANTS AGREEMENT

THIS RESTRICTIVE COVENANTS AGREEMENT ("Agreement"), dated as of May __, 1994, is by and between HPI ACQUISITION CORP., a Delaware corporation with its principal place of business at 630 Dundee Road, Suite 400, Northbrook Illinois 60054 (together with its affiliates, "Buyer"); GODIVA PRODUCTS LIMITED, a company incorporated in England whose principal place of business is at Charles Street, Warwick, Warwickshire, England CV34 5LR; and MICHAEL J. BRAYNE, an individual residing at ______("Covenantor").

RECITALS:

WHEREAS, pursuant to a Stock Purchase Agreement, dated as of the date of this Agreement, by and among Buyer, Covenantor, HFP Partners, L.P., HMTC Partners, L.P. and others (the "Stock Purchase Agreement"), Buyer is purchasing all of the issued and outstanding shares of common stock of Hale Products, Inc., a Delaware corporation (together with the Subsidiaries (as defined in the Stock Purchase Agreement), the "Corporation"); and

 $\mbox{WHEREAS},$ Godiva Products Limited is a wholly-owned subsidiary of Hale Products, Inc.; and

 $$\operatorname{\textsc{WHEREAS}}$, Covenantor is employed as an Executive Officer of Godiva Products Limited; and$

WHEREAS, in addition to serving as an Executive Officer of Godiva Products Limited, Covenantor is a stockholder of the Corporation and, pursuant to the transactions contemplated by the Stock Purchase Agreement, is receiving valuable consideration for his stock in the Corporation; and

WHEREAS, on and after the date of this Agreement, the Corporation will continue to conduct its business consisting of the manufacture, distribution, sales and service of fire fighting pumps, rescue equipment and tools and other related products, and other related activities (the "Business"); and

WHEREAS, during the period of his employment with the Corporation, Covenantor has had access to, and has gained knowledge with respect to, the Business; and

WHEREAS, in order to protect Buyer and the Business, and Buyer's investment in the Business, it is necessary that Covenantor refrain from direct or indirect competition in the Business to be carried on by the Corporation.

 $\,$ NOW, THEREFORE, in consideration of the money to be paid to Covenantor pursuant to the Stock Purchase Agreement, the

mutual covenants and agreements hereinafter set forth, and other good and valuable consideration, the receipt of which is hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

Covenant. For a period of one (1) year following Covenantor's ceasing to be employed by the Corporation regardless of reason, Covenantor agrees that he shall not, directly or indirectly, for his own account or as agent, employee, officer, director, trustee, consultant or partner, or as a stockholder or equity owner of any corporation or any other entity (except for ownership of securities constituting less than five percent (5%) of any class of securities of a public company), or member of any firm or otherwise, (a) engage or attempt to engage, in the Restricted Territory (as hereinafter defined), in the Business or any other business activity which is the same as, substantially similar to or directly or indirectly competitive with the business conducted by the Corporation at the date of Covenantor's ceasing to be employed by the Corporation, (b) employ or solicit the employment of any person who was employed by the Corporation as an executive officer at the date of Covenantor's ceasing to be employed by the Corporation or within six months prior to such date, (c) canvass or solicit business in competition with the business conducted by the Corporation at the date of Covenantor's ceasing to be employed by the Corporation from any person or entity who during the last six months of Covenantor's employment shall have been a customer of the Corporation, or from any person or entity which there is reason to believe might in the future become a customer of the Corporation as a result of marketing efforts, contacts or other facts and circumstances of which Covenantor is aware during the period of Covenantor's employment with the Corporation, (d) willfully dissuade or discourage any person or entity from using, employing or conducting business with the Corporation, or (e) disrupt or interfere with, or seek to disrupt or interfere with, the business or contractual relationship between the Corporation and any supplier who during the last six months of Covenantor's employment with the Corporation shall have supplied components, materials or services to the Corporation. For purposes of this Agreement, the term Restricted Territory shall mean the United States, the United Kingdom of Great Britain, Northern Ireland and Continental Europe.

Confidentiality Agreement.

(a) Covenantor has had, and will continue to have, access to, and has gained, and will continue to gain, knowledge with respect to, all of the Business, the Corporation's trade secrets, financial results and information, processes and techniques, methods of doing business and information concerning customers and suppliers, and other valuable and confidential

information, which is not generally known to the public (the "Confidential Information"). The parties acknowledge that unauthorized disclosure or misuse of the Confidential Information following Covenantor's ceasing to be employed by the Corporation may cause irreparable damage to Buyer, the Corporation and the Business. The parties also agree that covenants by Covenantor not to make unauthorized disclosures of the Confidential Information are essential to the growth and stability of the Business. Accordingly, Covenantor agrees that, following the Covenantor's ceasing to be employed by the Corporation regardless of reason, he shall not use or disclose any Confidential Information obtained by him in the course of his employment with the Corporation.

- (b) Nothing in Section 2(a) shall prevent Covenantor from disclosing or using any such information as may be reasonably necessary (i) in connection with any lawsuit or arbitration arising out of this Agreement, (ii) in connection with any judicial or administrative filing, investigation or proceeding or otherwise with or by a governmental agency or (iii) as otherwise may be required by law.
- 3. Reasonableness of Limitations. Covenantor agrees that the territorial, time and other limitations contained in this Agreement are reasonable and properly required for the adequate protection of the business and affairs of Buyer and the Corporation, and in the event that any one or more of such territorial, time or other limitations is found to be unreasonable by a court of competent jurisdiction, Covenantor agrees to submit to the reduction of said territorial, time or other limitations to such an area, period or otherwise as the court may determine to be reasonable. In the event that any limitation under this Agreement is found to be unreasonable or otherwise invalid in any jurisdiction, in whole or in part, Covenantor acknowledges and agrees that such limitation shall remain and be valid in all other jurisdictions.
- 4. Remedies. Covenantor acknowledges that Buyer and the Corporation may suffer damages incapable of ascertainment in the event the provisions of this Agreement are breached and that Buyer and the Corporation may be irreparably damaged in the event that the provisions of this Agreement are not enforced, and Covenantor represents and warrants that his attorneys have thoroughly and completely reviewed this Agreement with him and that Covenantor fully understands the contents hereof. Therefore, should any dispute arise with respect to the breach or threatened breach of this Agreement, Covenantor agrees and consents that, in addition to any and all other remedies available to Buyer and the Corporation, an injunction or restraining order or other equitable relief may be issued or

ordered by a court of competent jurisdiction restraining any breach or threatened breach of this Agreement.

- 5. Jurisdiction. The parties hereto agree that the exclusive venue and place of trial for the resolution of any disputes arising in connection with the interpretation or enforcement of this Agreement shall be the English Courts.
- 6. Miscellaneous. This Agreement shall be governed by and shall be construed in accordance with the internal laws of England without regard to principles of conflicts of laws. All headings and subtitles contained herein are for convenience of reference only and are not of substantive effect. This Agreement constitutes the entire agreement among the parties hereto with respect to the subject matter hereof and supersedes all prior negotiations and understandings, written or oral, with respect to the subject matter hereof. There are no oral agreements in connection with this Agreement. Neither this Agreement nor any provision hereof may be waived, terminated, modified or amended orally or by any course of conduct but only by an agreement in writing duly executed by all of the parties hereto. If any article, section, portion, subsection or subportion of this Agreement shall be determined to be unenforceable or invalid, then such article, section, portion, subsection or subportion valid to the maximum extent under applicable law, and any such determination shall not affect the remainder of this Agreement, which shall be and shall remain binding and effective as against all parties hereto.

IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement as of the date first above written.

HPI ACQUISITION CORP.	
Ву	
	Title
GODIVA PRODUCTS LIMITED	
Ву	
	Title
Michael J. Bravne	

RESTRICTIVE COVENANTS AGREEMENT

THIS RESTRICTIVE COVENANTS AGREEMENT ("Agreement"), dated as of May __, 1994, is by and between HPI ACQUISITION CORP., a Delaware corporation with its principal place of business at 630 Dundee Road, Suite 400, Northbrook Illinois 60054 (together with its affiliates, "Buyer"); HALE PRODUCTS EUROPE GMBH, a company incorporated in Germany whose principal place of business is at Industriegebiet-Nord, Benzstrasse 4, 64807 Dieburg, Germany; and KURT H.C. BOTTCHER, an individual residing at

("Covenantor").

RECITALS:

WHEREAS, pursuant to a Stock Purchase Agreement, dated as of the date of this Agreement, by and among Buyer, Covenantor, HFP Partners, L.P., HMTC Partners, L.P. and others (the "Stock Purchase Agreement"), Buyer is purchasing all of the issued and outstanding shares of common stock of Hale Products, Inc., a Delaware corporation (together with the Subsidiaries (as defined in the Stock Purchase Agreement), the "Corporation"); and

WHEREAS, Hale Products Europe GMBH is an indirect wholly-owned subsidiary of Hale Products, Inc.; and

 $\mbox{WHEREAS},$ Covenantor is employed as an Executive Officer of Hale Products Europe GMBH; and

WHEREAS, in addition to serving as an Executive Officer of Hale Products Europe GMBH, Covenantor is a stockholder of the Corporation and, pursuant to the transactions contemplated by the Stock Purchase Agreement, is receiving valuable consideration for his stock in the Corporation; and

WHEREAS, on and after the date of this Agreement, the Corporation will continue to conduct its business consisting of the manufacture, distribution, sales and service of fire fighting pumps, rescue equipment and tools and other related products, and other related activities (the "Business"); and

WHEREAS, during the period of his employment with the Corporation, Covenantor has had access to, and has gained knowledge with respect to, the Business; and

WHEREAS, in order to protect Buyer and the Business, and Buyer's investment in the Business, it is necessary that Covenantor refrain from direct or indirect competition in the Business to be carried on by the Corporation.

 $\,$ NOW, THEREFORE, in consideration of the money to be paid to Covenantor pursuant to the Stock Purchase Agreement, the

mutual covenants and agreements hereinafter set forth, and other good and valuable consideration, the receipt of which is hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

Covenant. For a period of one (1) year following Covenantor's ceasing to be employed by the Corporation regardless of reason, Covenantor agrees that he shall not, directly or indirectly, for his own account or as agent, employee, officer, director, trustee, consultant or partner, or as a stockholder or equity owner of any corporation or any other entity (except for ownership of securities constituting less than five percent (5%) of any class of securities of a public company), or member of any firm or otherwise, (a) engage or attempt to engage, in the Restricted Territory (as hereinafter defined), in the Business or any other business activity which is the same as, substantially similar to or directly or indirectly competitive with the business conducted by the Corporation at the date of Covenantor's ceasing to be employed by the Corporation, (b) employ or solicit the employment of any person who was employed by the Corporation as an executive officer at the date of Covenantor's ceasing to be employed by the Corporation or within six months prior to such date, (c) canvass or solicit business in competition with the business conducted by the Corporation at the date of Covenantor's ceasing to be employed by the Corporation from any person or entity who during the last six months of Covenantor's employment shall have been a customer of the Corporation, or from any person or entity which there is reason to believe might in the future become a customer of the Corporation as a result of marketing efforts, contacts or other facts and circumstances of which Covenantor is aware during the period of Covenantor's employment with the Corporation, (d) willfully dissuade or discourage any person or entity from using, employing or conducting business with the Corporation, or (e) disrupt or interfere with, or seek to disrupt or interfere with, the business or contractual relationship between the Corporation and any supplier who during the last six months of Covenantor's employment with the Corporation shall have supplied components, materials or services to the Corporation. For purposes of this Agreement, the term Restricted Territory shall mean the United States, the United Kingdom of Great Britain, Northern Ireland and Continental Europe.

Confidentiality Agreement.

(a) Covenantor has had, and will continue to have, access to, and has gained, and will continue to gain, knowledge with respect to, all of the Business, the Corporation's trade secrets, financial results and information, processes and techniques, methods of doing business and information concerning customers and suppliers, and other valuable and confidential

information, which is not generally known to the public (the "Confidential Information"). The parties acknowledge that unauthorized disclosure or misuse of the Confidential Information following Covenantor's ceasing to be employed by the Corporation may cause irreparable damage to Buyer, the Corporation and the Business. The parties also agree that covenants by Covenantor not to make unauthorized disclosures of the Confidential Information are essential to the growth and stability of the Business. Accordingly, Covenantor agrees that, following the Covenantor's ceasing to be employed by the Corporation regardless of reason, he shall not use or disclose any Confidential Information obtained by him in the course of his employment with the Corporation.

- (b) Nothing in Section 2(a) shall prevent Covenantor from disclosing or using any such information as may be reasonably necessary (i) in connection with any lawsuit or arbitration arising out of this Agreement, (ii) in connection with any judicial or administrative filing, investigation or proceeding or otherwise with or by a governmental agency or (iii) as otherwise may be required by law.
- 3. Reasonableness of Limitations. Covenantor agrees that the territorial, time and other limitations contained in this Agreement are reasonable and properly required for the adequate protection of the business and affairs of Buyer and the Corporation, and in the event that any one or more of such territorial, time or other limitations is found to be unreasonable by a court of competent jurisdiction, Covenantor agrees to submit to the reduction of said territorial, time or other limitations to such an area, period or otherwise as the court may determine to be reasonable. In the event that any limitation under this Agreement is found to be unreasonable or otherwise invalid in any jurisdiction, in whole or in part, Covenantor acknowledges and agrees that such limitation shall remain and be valid in all other jurisdictions.
- 4. Remedies. Covenantor acknowledges that Buyer and the Corporation may suffer damages incapable of ascertainment in the event the provisions of this Agreement are breached and that Buyer and the Corporation may be irreparably damaged in the event that the provisions of this Agreement are not enforced, and Covenantor represents and warrants that his attorneys have thoroughly and completely reviewed this Agreement with him and that Covenantor fully understands the contents hereof. Therefore, should any dispute arise with respect to the breach or threatened breach of this Agreement, Covenantor agrees and consents that, in addition to any and all other remedies available to Buyer and the Corporation, an injunction or restraining order or other equitable relief may be issued or

ordered by a court of competent jurisdiction restraining any breach or threatened breach of this Agreement.

- 5. Jurisdiction. The parties hereto agree that the exclusive venue and place of trial for the resolution of any disputes arising in connection with the interpretation or enforcement of this Agreement shall be the German Courts.
- 6. Miscellaneous. This Agreement shall be governed by and shall be construed in accordance with the internal laws of Germany without regard to principles of conflicts of laws. All headings and subtitles contained herein are for convenience of reference only and are not of substantive effect. This Agreement constitutes the entire agreement among the parties hereto with respect to the subject matter hereof and supersedes all prior negotiations and understandings, written or oral, with respect to the subject matter hereof. There are no oral agreements in connection with this Agreement. Neither this Agreement nor any provision hereof may be waived, terminated, modified or amended orally or by any course of conduct but only by an agreement in writing duly executed by all of the parties hereto. If any article, section, portion, subsection or subportion of this Agreement shall be determined to be unenforceable or invalid, then such article, section, portion, subsection or subportion valid to the maximum extent required so as to render the article, section, portion, subsection or subportion valid to the maximum extent under applicable law, and any such determination shall not affect the remainder of this Agreement, which shall be and shall remain binding and effective as against all parties hereto.

 $\,$ IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement as of the date first above written.

HPI ACQUISITION CORP.	
Ву	Title
HALE PRODUCTS EUROPE GMBH	
Ву	Title
	11116
Kurt H.C. Bottcher	

RESTRICTIVE COVENANTS AGREEMENT

THIS RESTRICTIVE COVENANTS AGREEMENT ("Agreement"), dated as of May ___, 1994, is by and between HPI ACQUISITION CORP., a Delaware corporation with its principal place of business at 630 Dundee Road, Suite 400, Northbrook Illinois 60054 (together with its affiliates, "Buyer"); HALE PRODUCTS, INC., a Delaware corporation with its principal place of business at 700 Spring Mill Avenue, Conshohocken, Pennsylvania 19428; and HFP PARTNERS, L.P., a Delaware limited partnership with an address at 355 South Grand Avenue, 42nd Floor, Los Angeles, California 90071 ("Covenantor").

RECITALS:

WHEREAS, pursuant to a Stock Purchase Agreement, dated as of the date of this Agreement, by and among Buyer, Covenantor, HMTC Partners, L.P. and others (the "Stock Purchase Agreement"), Buyer is purchasing all of the issued and outstanding shares of common stock of Hale Products, Inc., a Delaware corporation (together with its Subsidiaries (as defined in the Stock Purchase Agreement), the "Corporation"); and

WHEREAS, Covenantor is a stockholder of the Corporation and, pursuant to the transactions contemplated by the Stock Purchase Agreement, is receiving valuable consideration for its stock in the Corporation; and

WHEREAS, on and after the date of this Agreement, the Corporation will continue to conduct its business consisting of the manufacture, distribution, sales and service of fire fighting pumps, rescue equipment and tools and other related products, and other related activities (the "Business"); and

WHEREAS, during the period that it has owned stock in the Corporation, Covenantor has had access to, and has gained knowledge with respect to, the Business; and

WHEREAS, in order to protect Buyer and the Business, and Buyer's investment in the Business, it is necessary that Covenantor refrain from direct or indirect competition in the Business to be carried on by the Corporation.

NOW, THEREFORE, in consideration of the money to be paid to Covenantor pursuant to the Stock Purchase Agreement, the mutual covenants and agreements hereinafter set forth, and other good and valuable consideration, the receipt of which is hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

Covenant. For a period of one (1) year from the date of this Agreement, Covenantor agrees that it shall not, directly or indirectly, for its own account or as agent, employee, officer, director, trustee, consultant or partner, or as a stockholder or equity owner of any corporation or any other entity (except for ownership of securities constituting less than five percent (5%) of any class of securities of a public company), or member of any firm or otherwise, (a) engage or attempt to engage, in the Restricted Territory (as hereinafter defined), in the Business or any other business activity which is the same as, substantially similar to or directly or indirectly competitive with the business conducted by the Corporation at the date of this Agreement, (b) employ or solicit the employment of any person who is employed by the Corporation at the date of this Agreement or was employed by the Corporation within the past six months, (c) canvass or solicit business in competition with the business conducted by the Corporation at the date of this Agreement from any person or entity who during the past six months shall have been a customer of the Corporation, or from any person or entity which there is reason to believe might in the future become a customer of the Corporation as a result of marketing efforts, contacts or other facts and circumstances of which Covenantor is aware, (d) willfully dissuade or discourage any person or entity from using, employing or conducting business with the Corporation, or (e) disrupt or interfere with, or seek to disrupt or interfere with, the business or contractual relationship between the Corporation and any supplier who during the past six months shall have supplied components, materials or services to the Corporation. For purposes of this Agreement, the term Restricted Territory shall mean the United States, the United Kingdom of Great Britain, Northern Ireland and Continental Europe.

Confidentiality Agreement.

(a) Covenantor has had access to, and has gained knowledge with respect to, all of the Business, the Corporation's trade secrets, financial results and information, processes and techniques, methods of doing business and information concerning customers and suppliers, and other valuable and confidential information, which is not generally known to the public (the "Confidential Information"). The parties acknowledge that unauthorized disclosure or misuse of the Confidential Information following the date of this Agreement may cause irreparable damage to Buyer, the Corporation and the Business. The parties also agree that covenants by Covenantor not to make unauthorized disclosures of the Confidential Information are essential to the growth and stability of the Business. Accordingly, Covenantor agrees that it shall not use or disclose any Confidential Information obtained by it while a stockholder the Corporation.

- (b) Nothing in Section 2(a) shall prevent Covenantor from disclosing or using any such information as may be reasonably necessary (i) in connection with any lawsuit or arbitration arising out of this Agreement, (ii) in connection with any judicial or administrative filing, investigation or proceeding or otherwise with or by a governmental agency or (iii) as otherwise may be required by law.
- 3. Reasonableness of Limitations. Covenantor agrees that the territorial, time and other limitations contained in this Agreement are reasonable and properly required for the adequate protection of the business and affairs of Buyer and the Corporation, and in the event that any one or more of such territorial, time or other limitations is found to be unreasonable by a court of competent jurisdiction, Covenantor agrees to submit to the reduction of said territorial, time or other limitations to such an area, period or otherwise as the court may determine to be reasonable. In the event that any limitation under this Agreement is found to be unreasonable or otherwise invalid in any jurisdiction, in whole or in part, Covenantor acknowledges and agrees that such limitation shall remain and be valid in all other jurisdictions.
- 4. Remedies. Covenantor acknowledges that Buyer and the Corporation may suffer damages incapable of ascertainment in the event the provisions of this Agreement are breached and that Buyer and the Corporation may be irreparably damaged in the event that the provisions of this Agreement are not enforced, and Covenantor represents and warrants that its attorneys have thoroughly and completely reviewed this Agreement with it and that Covenantor fully understands the contents hereof. Therefore, should any dispute arise with respect to the breach or threatened breach of this Agreement, Covenantor agrees and consents that, in addition to any and all other remedies available to Buyer and the Corporation, an injunction or restraining order or other equitable relief may be issued or ordered by a court of competent jurisdiction restraining any breach or threatened breach of this Agreement.
- 5. Jurisdiction. The parties hereto agree that the exclusive venue and place of trial for the resolution of any disputes arising in connection with the interpretation or enforcement of this Agreement shall be the United States District Court for the Northern District of Illinois.
- 6. Miscellaneous. This Agreement shall be governed by and shall be construed in accordance with the internal laws of the State of Pennsylvania without regard to principles of conflicts of laws. All headings and subtitles contained herein are for convenience of reference only and are not of substantive effect. This Agreement constitutes the entire agreement among

the parties hereto with respect to the subject matter hereof and supersedes all prior negotiations and understandings, written or oral, with respect to the subject matter hereof. There are no oral agreements in connection with this Agreement. Neither this Agreement nor any provision hereof may be waived, terminated, modified or amended orally or by any course of conduct but only by an agreement in writing duly executed by all of the parties hereto. If any article, section, portion, subsection or subportion of this Agreement shall be determined to be unenforceable or invalid, then such article, section, portion, subsection or subportion shall be modified to the minimum extent required so as to render the article, section, portion, subsection or subportion valid to the maximum extent under applicable law, and any such determination shall not affect the remainder of this Agreement, which shall be and shall remain binding and effective as against all parties hereto.

IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement as of the date first above written.

HPI	ACQUISITIO	ON CORI	Ρ.	
Ву				
				Title
HFP	PARTNERS,	L.P.		
Ву			general	partner
Bv				
,				Title

FIRST AMENDMENT DATED AS OF MAY 23, 1994 TO SECOND AMENDED AND RESTATED CREDIT AGREEMENT DATED AS OF JANUARY 29, 1993

THIS AMENDMENT, dated as of May 23, 1994, is entered into by and among IDEX CORPORATION, a Delaware corporation (the "Borrower"), the banking institutions (the "Banks") signatory to the hereinafter defined Agreement and CONTINENTAL BANK, N.A. ("Continental") as agent for the Banks (in such capacity, the "Agent").

RECITALS:

- A. The Borrower, the Banks and the Agent have entered into that certain Second Amended and Restated Credit Agreement dated as of January 29, 1993 (said Credit Agreement shall hereinafter be referred to as the "Agreement"; the terms defined in the Agreement and not otherwise defined herein shall be used herein as defined in the Agreement).
- B. The Borrower, the Banks and the Agent wish to amend certain provisions of the Agreement.
- C. Therefore, in consideration of the premises herein contained, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties hereto agree as follows:
 - 1. AMENDMENTS TO THE AGREEMENT.
- 1.1. Section 1.2. of the Agreement. Section 1.2. of the Agreement is hereby amended as of the date hereof by deleting the amount "\$100,000,000" from the third line thereof and inserting the amount "\$150,000,000" in lieu thereof.
- 1.2 Section 1.3.(b) of the Agreement. Section 1.3.(b) of the Agreement is hereby amended as of the date hereof by deleting the amount "1/2 of 1%" from the sixth line thereof and inserting the amount "1/4 of 1%" in lieu thereof.
- 1.3. Section 2.1. of the Agreement. Section 2.1. of the Agreement is hereby amended as of the date hereof by deleting the amount "\$5,000,000" from the ninth line thereof and inserting the amount "\$2,000,000" in lieu thereof.
- 1.4. Section 2.2. of the Agreement. Section 2.2. of the Agreement is hereby amended as of the date hereof by deleting the first sentence thereof and inserting the following sentence in lieu thereof:
- "All Loans made by each Bank shall be evidenced by a promissory note of Borrower which shall be in the form of Exhibit B attached hereto (herein, as endorsed, amended or otherwise modified or, in accordance with Section 11.12(a) or

2

otherwise, replaced from time to time, called the "Notes"), dated the Second Amended and Restated Closing Date, or such replacement date, if applicable, payable to the order of such Bank in the principal amount of its Percentage of the then Total Commitment Amount, and in each case subject, however, to the provisions of such Notes to the effect that the principal amount payable thereunder at any time shall not exceed the then unpaid principal amount of all Loans evidenced thereby."

- 1.5. Section 3.1. of the Agreement.
- (a) Section 3.1. of the Agreement is hereby amended as of the date hereof by deleting the first sentence thereof and inserting the following sentence in lieu thereof:
- "All Loans shall be made and maintained as, or be converted into, Loans having a floating rate of interest for an undetermined period determined on the basis of the Reference Rate (herein called "Domestic Loans") or, at Borrower's election made in accordance with this Section, Loans having a fixed rate of interest for a determined period calculated on the basis of the Eurodollar Interbank Rate (Reserve Adjusted) (herein called "Eurodollar Loans"); provided, however, that there shall be no more than eight (8) Eurodollar Interest Periods outstanding at any one time."
- (b) Section 3.1. of the Agreement is hereby amended as of the date hereof by deleting the amount "\$5,000,000" from the third line of subsection (a) thereof and inserting the amount "\$2,000,000" in lieu thereof.
- 1.6. Section 7.2.6. of the Agreement. Subsection 7.2.6.(a) of the Agreement is hereby amended as of the date hereof by deleting Subsection 7.2.6.(a) in its entirety and inserting the following in lieu thereof:
 - "(a) Borrower will not at any time declare, pay or make any dividend or distribution of any shares of capital stock of any class of Borrower (other than dividends or distributions payable in its stock, or warrants to purchase its stock, or splitups or reclassifications of its stock into additional or other shares of its stock, or conversions from one class of common stock into another), or apply, or permit any of its Subsidiaries to apply, any of its funds, property or assets to the purchase, redemption or other retirement of any shares, or of any options to purchase or acquire any shares, of capital stock of Borrower except, so long as both before and after giving effect to such payment, no Default shall occur and be continuing, Borrower may pay quarterly cash dividends on its Common Stock in an aggregate amount (for all such dividends in any Fiscal Year) not greater than 50% of the cumulative consolidated net earnings of Borrower and its Subsidiaries (commencing with Fiscal Year 1993), as reflected in the consolidated statements of income and cash flow delivered pursuant to Section 7.1.1 of this Agreement, beginning on January 1, 1993; provided, however, that in no event shall the aggregate amount of such dividends in any Fiscal Year exceed the Net Income for such Fiscal Year.'

 $\ensuremath{\mathtt{3}}$ 1.7. Section 10.1. of the Agreement.

(a) Definition of "Applicable Eurodollar Rate Margin". Section 10.1 of the Agreement is hereby amended as of the date hereof by deleting the table in the definition of "Applicable Eurodollar Rate Margin" in its entirety and replacing it with the following table:

Operating Cash Flow to Adjusted Interest Expense Ratio equal to or greater

		less than 3.5	than 3.5, but less than 4.0	equal to or greater than 4.0	
Funded Indebtedness to	greater than .74	1.50	1.50	1.50	
Funded Indebtedness Plus	equal to or less than .74, but greater than .60	1.50	1.25	1.25	
Equity Ratio	equal to or less than .60, but greater than .50	1.50	1.25	1.00	
	equal to or less than .50		1.00	. 75	

⁽b) Definition of "Applicable Standby Letter of Credit Fee". Section 10.1 of the Agreement is hereby amended as of the date hereof by deleting the table in the definition of "Applicable Standby Letter of Credit Fee" and inserting the following table in lieu thereof:

Operating Cash Flow to Adjusted Interest Expense Ratio

		Expense Natio			
		less than 3.5	equal to or greater than 3.5, but less than 4.0	equal to or greater than 4.0	
Funded Indebtedness to	greater than .74	1.00	1.00	1.00	
Funded Indebtedness Plus	equal to or less than .74, but greater than .	1.00	. 75	.75	
Equity Ratio	equal to or less than .60, but greater than .	1.00	. 75	. 50	
	equal to or less than .		. 75	. 50	

- 2. WARRANTIES. To induce the Agent and the Banks to enter into this Amendment, the Borrower warrants that:
- 2.1. Authorization. The Borrower is duly authorized to execute and deliver this Amendment and the Replacement Notes (as hereinafter defined) and is and will continue to be duly authorized to borrow monies under the Agreement, as amended hereby, and to perform its obligations under the Agreement, as amended hereby, and the Replacement Notes.
- 2.2. No Conflicts. The execution and delivery of this Amendment and the Replacement Notes, and the performance by the Borrower of its obligations under the Agreement, as amended hereby, and the Replacement Notes, do not and will not conflict with any provision of law or of the charter or by-laws of the Borrower or any Subsidiary or of any agreement binding upon the Borrower or any Subsidiary.
- 2.3. Validity and Binding Effect. The Agreement, as amended hereby, is, and each of the Replacement Notes when duly executed and delivered will be, a legal, valid and binding obligation of the Borrower, enforceable against the Borrower in accordance with its terms, except as enforceability may be limited by bankruptcy, insolvency or other similar laws of general application affecting the enforcement of creditors' rights or by general principles of equity limiting the availability of equitable remedies.
- 3. CONDITIONS PRECEDENT TO AMENDMENTS. The amendments contemplated by Section 1 hereof are subject to the satisfaction of each of the following conditions precedent:
- 3.1. Documentation. The Borrower shall have delivered to the Agent all of the following, each duly executed and dated the date hereof, in form and substance satisfactory to the Agent:
 - (a) Replacement Notes. Promissory notes of the Borrower (the "Replacement Notes"), each substantially in the form set forth as Exhibit A hereto delivered to the Agent for the account of each of the Banks in the full amount equal to such Lender's Commitment.

Upon receipt of the Replacement Notes, each Bank will: (i) record the aggregate unpaid principal amount of such Bank's Note dated January 29, 1993 (the "Original Notes") issued under the Agreement in its records or, at its option, on the schedule attached to the Replacement Note issued to it as the aggregate unpaid principal amount of the Replacement Note issued to it; (ii) mark the Original Note issued to it as replaced by the Replacement Note; and (iii) return the Original Note issued to it to the Agent for delivery to the Borrower upon the Borrower's request. Thereafter, all references in the Agreement and any and all instruments or documents provided for therein or delivered or to be delivered thereunder or in connection therewith to the Original Notes shall be deemed references to the Replacement Notes. The replacement of the Original Notes with the Replacement Notes shall not be construed (i) to deem paid or forgiven the unpaid principal amount of, or unpaid accrued interest on, the Original Notes outstanding at the time of replacement, or (ii) to release, cancel, terminate or otherwise adversely affect all or any part of any lien, mortgage, deed of trust, assignment, security interest or other encumbrance heretofore granted to or for the benefit of the payees of the Original Notes which has not otherwise been expressly released.

- (b) Borrower Resolutions. Copies for each Bank duly certified by the secretary or an assistant secretary of the Borrower, of (i) resolutions of the Borrower's Board of Directors authorizing or ratifying the execution and delivery of this Amendment and the Replacement Notes and authorizing the borrowings under the Agreement, as amended hereby, (ii) all documents evidencing other necessary corporate action, and (iii) all approvals or consents, if any, with respect to this Amendment and the Replacement Notes.
- (c) Incumbency Certificate. Certificates for each Bank of the secretary or an assistant secretary of the Borrower certifying the names of the Borrower's officers authorized to sign this Amendment, the Replacement Notes and all other documents or certificates to be delivered hereunder, together with the true signatures of such officers.
- (d) Opinion. An opinion of Latham & Watkins, special counsel to the Borrower, addressed to the Agent and the Banks, in substantially the form of Exhibit B hereto.
- (e) Certificate. A certificate of an Authorized Officer of the Borrower as to the matters set out in Sections 3.2 and 3.3 hereof.
 - (f) Other. Such other documents as the Agent may reasonably request.
- 3.2. No Default. As of the closing date hereof, no Default shall have occurred and be continuing.
- 3.3. Warranties. As of the date hereof, the warranties in Article VI of the Agreement and in Section 2 of this Amendment shall be true and correct as though made on such date, except for such changes as are specifically permitted under the Agreement.
 - 4. GENERAL.

- 4.1. Expenses. The Borrower agrees to pay the Agent, upon demand, for all reasonable expenses, including reasonable attorneys' and legal assistants' fees incurred by the Agent in connection with the preparation, negotiation and execution of this Amendment, the Replacement Notes and any document required to be furnished therewith.
- 4.2. Governing Law. This Amendment and the Replacement Notes shall each be deemed to be a contract made under and governed by the internal laws of the State of Illinois. For purposes of any action or proceeding involving this Amendment, Borrower hereby expressly submits to the jurisdiction of all federal and state courts located in the State of Illinois and consents that it may be served with any process or paper by registered mail or by personal service within or without the State of Illinois, provided a reasonable time for appearance is allowed.
- 4.3. Successors. This Amendment shall be binding upon the Borrower, the Agent and the Banks and their respective successors and assigns, and shall inure to the benefit of the Borrower, the Agent and the Banks and their successors and assigns.
- 4.4. Confirmation of the Agreement. Except as amended hereby, the Agreement shall remain in full force and effect and is hereby ratified and confirmed in all respects.
- 4.5. References to the Agreement. Upon the effectiveness of this Amendment, each reference in the Agreement to "this Agreement," "hereunder," "hereof," or words of like import, and each reference to the Agreement in any and all instruments or documents provided for in the Agreement or delivered or to be delivered thereunder or in connection therewith, shall, except where the context otherwise requires, be deemed a reference to the Agreement, as amended hereby.
- 4.6. Effective Date. This Amendment shall become effective as of the date first written above upon the execution and delivery of counterparts of this Amendment by each of the Banks, the Guarantors and the Borrower.
- 4.7. Counterparts. This Amendment may be executed in any number of counterparts, and by different parties hereto in separate counterparts, each of which when so executed and delivered shall be deemed to be an original and all of which taken together shall constitute one and the same agreement.

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed and delivered at Chicago, Illinois by their respective officers thereunto duly authorized as of the date first written above.

> IDEX CORPORATION, a Delaware corporation

By: /s/ Wayne P. Sayatovic Title: Vice President - Finance

PERCENTAGE OF TOTAL COMMITMENT

CONTINENTAL BANK N.A., as a Bank [22.5%]

and as Agent

By: /s/ Bruce A. Simon Title: Senior Vice President

BANK OF SCOTLAND [10.0%]

By: /s/ Catherine Oniffrey Title: Vice President

NATIONAL CITY BANK [20.0%]

By: /s/ Frank F. Pagura Title: Account Officer

PITTSBURGH NATIONAL BANK [20.0%]

> By: /s/ William S. Richards Title: Assistant Vice President

[12.5%] UNION BANK

By: /s/ Patrick M. Cassidy Title: Vice President

[15.0%] UNITED STATES NATIONAL BANK

OF OREGON

By: /s/ Timothy A. Miller Title: Corporate Banking Credit Officer

The undersigned hereby acknowledge and agree to this Amendment, and agree that

the Guaranty Agreement shall remain in full force and effect and is hereby ratified and confirmed this 23rd day of May, 1994.

BAND-IT-IDEX, INC. VIKING PUMP, INC.

By: /s/ Wayne P. Sayatovic
Title: Vice President
By: /s/ Wayne P. Sayatovic
Title: Vice President

VIBRATECH, INC. WARREN RUPP, INC.

By: /s/ Wayne P. Sayatovic
Title: Vice President

By: /s/ Wayne P. Sayatovic
Title: Vice President

LUBRIQUIP, INC. CORKEN, INC.

By: /s/ Wayne P. Sayatovic
Title: Vice President

By: /s/ Wayne P. Sayatovic
Title: Vice President

STRIPPIT, INC. PULSAFEEDER

By: /s/ Wayne P. Sayatovic
Title: Vice President

By: /s/ Wayne P. Sayatovic
Title: Vice President

FORM OF NOTE

\$		

January 29, 1993 as amended and restated as of May ____, 1994

FOR VALUE RECEIVED, the undersigned, IDEX CORPORATION, a Delaware corporation (herein called "Borrower"), promises to pay to the order of ______ (herein called "Bank") on June 30, 1999 the principal sum of _____ DOLLARS (\$______) or, if less, the principal amount of all Loans made by Bank to Borrower from time to time pursuant to that certain Second Amended and Restated Credit Agreement, dated as of January 29, 1993 (herein, together with all amendments and other modifications, if any, thereafter from time to time made thereto, called the "Credit Agreement"), among (i) Borrower, (ii) Continental Bank N.A., as agent (herein in such capacity called "Agent") and individually, and (iii) the various banking institutions (including the Bank) as are, or may from time to time become, parties thereto.

The unpaid principal amount of this Note from time to time outstanding shall bear interest as provided in Section 2.4 of the Credit Agreement.

The making of all Loans, the length of any Eurodollar Interest Periods applicable to any Eurodollar Loans and all payments on account of the principal hereof may be endorsed by the holder hereof on the grid attached to this Note.

All payments of principal and interest on this Note shall be payable in lawful currency of the United States of America at the office of Agent at 231 South LaSalle Street, Chicago, Illinois in same day or immediately available funds.

This Note is a Note referred to in, and evidences indebtedness incurred under, the Credit Agreement, to which reference is made for a description of the security for this Note and for a statement of the terms and conditions on which the Borrower is permitted and required to make prepayments of principal of this Note and on which the indebtedness evidenced hereby may be declared to be or shall become immediately due and payable.

Capitalized terms used herein without definition shall have the respective meanings provided in the Credit Agreement.

This Note is a restatement of the indebtedness evidenced by, and is a replacement of, that certain Note of the undersigned dated January 29, 1993 in the face principal amount of \$_______ payable to the order of Bank, and nothing contained herein or in the First Amendment to the Credit Agreement dated the date hereof shall be construed (i) to deem paid or forgiven the unpaid principal amount of, or unpaid accrued interest on, said Note outstanding at the time of its replacement by this Note, or (ii) to release, cancel, terminate or otherwise adversely affect all or any part of any lien, mortgage, deed of trust, assignment, security interest or other encumbrance heretofore granted to or for the benefit of the payee of said Note which has not otherwise been expressly released.

THIS NOTE HAS BEEN DELIVERED IN CHICAGO, ILLINOIS AND SHALL BE DEEMED TO BE A CONTRACT MADE UNDER AND GOVERNED BY THE INTERNAL LAWS OF THE STATE OF ILLINOIS.

IDEX CORPORATION, a Delaware corporation

3y:			
itle:			

Date	Amount of Loan Made	Amount of Principal Repaid	Unpaid Principal Balance	Total	Notation Made By	

FORM OF OPINION OF SPECIAL COUNSEL TO THE BORROWER

To: Continental Bank N.A., as Agent for the Banks 231 South LaSalle Street Chicago, Illinois 60697

Each Bank party to the Agreement referred to below

Ladies and Gentlemen:

We have acted as special counsel for IDEX Corporation, a Delaware corporation (the "Borrower") in connection with a First Amendment dated as of May ____, 1994 (the "Amendment") to the Credit Agreement dated as of January 29, 1993, entered into between the Borrower, you, as Agent and as a Bank, and the banking institutions signatory thereto (the "Agreement"), and the transactions and other documents and instruments described therein. Unless otherwise defined herein, capitalized terms used herein shall have the respective meanings assigned to such terms in the Amendment.

In so acting, we, as counsel for the Borrower, have made such factual inquiries, and we have examined or caused to be examined such questions of law, as we have considered necessary or appropriate for the purposes of this opinion and, upon the basis of such inquiries and examinations, advise you that, in our opinion:

- 1. The Borrower is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware.
- 2. The Borrower has full corporate power and authority to enter into the Amendment. The Borrower has full corporate power and authority to perform its obligations under the Agreement, as amended by the Amendment, and under the Replacement Notes.
- 3. The execution and delivery of the Amendment and the Replacement Notes, the performance by the Borrower of its obligations under the Agreement, as amended by the Amendment, and under the Replacement Notes, and the borrowings by the Borrower under the Agreement, as amended by the Amendment, have been duly authorized by all necessary corporate action, and the Amendment and the Replacement Notes have been duly executed and delivered on behalf of the Borrower and constitute valid and binding obligations of the Borrower, enforceable in accordance with their respective terms, except as enforceability may be limited by bankruptcy, insolvency or other similar laws of general application affecting the enforcement of creditors' rights or by general principles of equity limiting the availability of equitable remedies.
- 4. None of the execution and delivery of the Amendment by Borrower, and the compliance with and performance of the terms and conditions thereof by Borrower on or prior to the date hereof (A) conflicts with, results in a breach or violation of, or constitutes a default under, any of the terms, conditions or provisions of (x) the Certificate of Incorporation or Bylaws of Borrower, (y) any term of any material indenture, loan agreement or other Instrument evidencing borrowed money, or any material order, writ, judgment or decree, in each case known to us, to which Borrower is a party or by which any of Borrower's properties or assets are bound,

or (z) any Illinois or United States federal statute, rule or regulation or the General Corporation law of Delaware, or (B) results in the creation of any Security Interest upon any of the properties or assets of Borrower under any agreement referred to in clause (y) above.

5. Neither the making of the Amendment or the Replacement Notes nor performance of the Agreement, as amended by the Amendment, or the Replacement Notes nor the borrowing under the Agreement, as amended by the Amendment, requires the consent or approval of any governmental instrumentality.

Our opinions in paragraph 3 above as to the enforceability of the agreements referred to therein are subject to (i) bankruptcy, insolvency, reorganization, moratorium or similar laws now or hereafter in effect relating to our limiting creditor's rights generally, (ii) general principles of equity, whether such enforceability is considered in a proceeding in equity or at law, (iii) public policy considerations or court decisions which may limit the rights of Banks or Agent to obtain certain indemnifications, and (iv) court decisions limiting enforceability to circumstances in which there is a material breach of a material provision of an agreement.

Very truly yours,

1 EXHIBIT 24.1

CONSENT OF INDEPENDENT AUDITORS

We consent to the incorporation by reference in the Registration Statement (Form S-8 No. 33-47678, 33-56586 and 33-67688) of IDEX Corporation of our report dated March 24, 1994. With respect to the consolidated financial statements of Hale Products, Inc. included in the Quarterly Report of IDEX Corporation (Form 10-Q) for the period ended June 30, 1994.

/s/ ERNST & YOUNG ERNST & YOUNG LLP

Philadelphia, Pennsylvania August 8, 1994